Regulation of Proteinase-Activated Receptor-2 (PAR-2) in Airway Epithelium

by

Vivek Dipak Gandhi

A thesis submitted in partial fulfillment of the requirements for the degree of

Doctor of Philosophy in Experimental Medicine

Department of Medicine University of Alberta

© Vivek Dipak Gandhi, 2018

#### Abstract

Proteinase-activated receptor-2 (PAR-2), a pro-inflammatory G-coupled receptor, is activated by exogenous or endogenous serine proteinases. PAR-2 is involved in the pathogenesis of inflammatory diseases including asthma. PAR-2 activation on airway epithelial cells (AECs) results in the release of inflammatory mediators that play an important role in the development of asthma. PAR-2 expression is upregulated on AECs in asthma and upregulated PAR-2 expression augments airway inflammation. Hence it is important to inhibit PAR-2 upregulation, but mechanisms involved in PAR-2 regulation are poorly studied.

Using *in vivo* and *in vitro* models, we showed that insulin regulates PAR-2 expression and upregulated PAR-2 increased the pro-inflammatory potential of AECs. We further showed that insulin regulates PAR-2 expression through PI3K-Akt-FOXO-1 pathway. This observation led to the hypothesis for future studies that inflammation-induced localized insulin resistance in the airway epithelium in asthma may result in PAR-2 upregulation on AECs.

PAR-2 expression is increased in inflammatory diseases, thus we studied the effect of stress stimuli, which are associated with inflammation, on PAR-2 expression in AECs. Our preliminary results suggest that oxidative stress and hypoxia do not modulate PAR-2 expression. We found that withdrawal of media supplements containing growth factors and hormones resulted in PAR-2 upregulation, and again PAR-2 upregulation increased the pro-inflammatory potential of AECs. We also showed that, in addition to insulin, hydrocortisone, bovine pituitary extract and retinoic acid regulate PAR-2 expression in AECs.

We cultured AECs obtained from asthmatic patients. We found that PAR-2 expression is increased in AECs from asthmatic compared to healthy individuals and this upregulated PAR-2 expression is maintained when cells were cultured *ex vivo* for multiple passages. Asthmatic AECs also showed similar media supplement deprivation-mediated PAR-2 regulation.

Our study indicates the possibility that inflammatory microenvironments such as insulin resistance and decreased cell growth nutrients may play a role in regulation of expression of proinflammatory PAR-2 receptors. Since our study highlights the role of inflammatory microenvironment in PAR-2 expression, our observations can be applied to inflammatory diseases of other organs such as colon and skin, where localized inflammation is present and PAR-2 expression is increased. Similar studies can be performed which may identify common dysregulatory mechanisms involved in increased PAR-2 expression in different inflammatory conditions and may lead to the development of novel therapeutic strategies to inhibit PAR-2 upregulation and control PAR-2-mediated inflammation.

#### Preface

Proteinase-activated receptor-2 (PAR-2) is a pro-inflammatory receptor, which is activated by serine proteinases. These serine proteinases can be exogenous such as allergens from house dust mite, cockroach or endogenous such as mast cell tryptase. PAR-2 activation on airway epithelial cells (AECs) results in the release of mediators that play an important role in the pathogenesis of asthma. In asthma, PAR-2 expression is increased on AECs but the underlying mechanisms and consequences of PAR-2 upregulation are not known. This thesis represents the work that was performed to understand the regulation of PAR-2 expression on AECs.

This is an original work by Vivek Gandhi. I was responsible for performing all the experiments, data collection and analysis as well as manuscript preparation. I have mentioned below the experiments which were done by our collaborators.

Chapter 1 is the introduction. Here I describe the role of airway epithelium and PAR-2 in asthma pathogenesis, which leads to the rationale for studying PAR-2 expression regulation on AECs.

Chapter 2 of this thesis is published as **Gandhi VD**, Shrestha Palikhe N, Hamza SM, Dyck JRB, Buteau J, Vliagoftis H., "Insulin decreases expression of the proinflammatory receptor proteinase-activated receptor-2 on human airway epithelial cells", *J Allergy Clin Immunol*. 2018 Jun 8. doi: 10.1016/j.jaci.2018.04.040. Shrestha Palikhe N performed transfection experiments. Hamza SM provided lung tissues and blood plasma from mice.

Chapter 3 contains preliminary work aiming at understanding role of insulin on PAR-2 expression in mice lungs *in vivo*. Ethics approval to perform these animal experiments was received from the University of Alberta Research Ethics Board, Project Name "Serine protease receptors (PARs) in airway inflammation and allergic sensitization", No. AUP00000353, 19<sup>th</sup> December 2017.

Chapter 4 contains preliminary work aiming at understanding role of insulin on PAR-2 expression in airway smooth muscle cells (SMCs). The experiments with SMCs were performed in collaboration with Dr. Andrew Halayko, University of Manitoba. Dr. Halayko's lab treated the SMCs as per our discussion and sent us cell lysate to evaluate PAR-2 expression.

Chapter 5 is a manuscript in preparation, where we studied the effect of media supplement deprivation on PAR-2 expression in AECs obtained from healthy and asthmatic individuals. Ethics approval to obtain AECs from asthma patients was received from the University of Alberta Research Ethics Board, Project Name "Airway epithelial cells", No. Pro00001190, 12<sup>th</sup> March 2018.

Chapter 6 shows data from experiments that were performed to understand the effect of different stress stimuli on PAR-2 expression in AECs. One of the stress stimuli was mechanical stretch and human rhino virus infection; these experiments were done in collaboration with the laboratories of Dr. Richard Leigh and Dr. David Proud, University of Calgary. The lab in Calgary treated the AECs as per our discussion and sent us cell lysate to evaluate PAR-2 expression.

Chapter 7 presents the results of experiments performed to determine if media supplement deprivation regulates expression of G protein-coupled receptors, other than PAR-2, in AECs.

Chapter 8 is a general discussion about the findings presented in this thesis, which leads to the conceptual model for regulation of PAR-2 expression in AECs.

## Dedication

I dedicate my Ph.D. to my grandparents Mr. Manilal Gordhandas Gandhi and Mrs. Kusum Manilal Gandhi, and my parents Mr. Dipak Manilal Gandhi and Mrs. Rekha Dipak Gandhi. My words fall short to describe the effort and energy that you have invested in nurturing me and my character, which have made me the person that I am today. Your blessings, foresight and trust in me made this feat possible.

### Acknowledgements

I thank the Almighty for their blessings and for providing the environment for my foundational development that ensured my success.

I am grateful to my teachers from kindergarten, schools and colleges to equip me with discipline, which is essential to become a successful student.

I am sincerely thankful to my Ph.D. supervisor Dr. Harissios Vliagoftis for his immeasurable contribution in shaping formative years of my career as a researcher. Your criticism albeit with guidance and constructive feedback helped me improve myself. Your continuous guidance, trust, encouragement, support and patience throughout my Ph.D. allowed me to grow as a researcher and enjoy my studies but also provided me the freedom of participating in extracurricular activities that played a vital role in the development of personal and professional traits which will be invaluable assets throughout my career.

I would also like to thank my committee members Dr. Paige Lacy and Dr. Allan Murray for their support, suggestions and valuable feedback that helped further the research projects.

I would also like to extend my sincere gratitude towards Dr. Dean Befus. I am fortunate to have you as my mentor. Thank you very much for always listening to my queries and investing time for discussion, feedback and guidance, which have contributed positively to my training as a researcher.

A special thanks to all the collaborators Dr. Jean Buteau, Dr. Jason Dyck, Dr. Shereen Hamza, Dr. James Martin and Dr. Alice Panariti for their help, support and guidance in their area of expertise.

I also thank funding agencies: The Lung Association of Alberta and NWT and University of Alberta for their studentship support.

I pay a special thanks to AllerGen for providing support to go to different national and international conferences. A heartily thanks to AllerGen and the late Michelle for their role in my professional development, by allowing me to be part of the trainee symposiums and AllerGen Students and New Professional Network executive committee. The trainee symposiums and conferences have helped me build up a network with my fellow trainees.

I also acknowledge help of the staff of core facility Dr. Aja Reiger, Dorothy, Greg and Dr. Stephen Ogg for their training and guidance on flow cytometry and microscopy. I would also like to thank Iris de Guzman for all her help from enrollment into the program to setting up the Ph.D. defense.

I am highly thankful to the past and present members of the Pulmonary Research Group for their training and discussions, which enabled me to carry out research. I am truly thankful to all the friends I made throughout these unforgettable years; Drew, Nami, Meghan, Sarah, Khadija, Kurtis, Ed, Yahya, Chad, Tae Chul, Dhaif who never let me feel home-sick and always kept me refreshed.

I am fortunate to have friends like Pandu (Vivek), Shrenik and Sarju, who took care of my parents in my absence. I was blessed to find home away from home; Ketan uncle, Parul aunty, Priyank bhai, Urvi didi, Elida, Samir uncle, Veeru aunty, Aarti, Pramesh bhai, Kajal bhabhi, Kunal bhai, Darshana bhabhi, Jignesh bhai, Kiran, Manasa, Manisha, Guddi, Ravi, Anindya and Supriya, all played various roles by being my guardians, family members and friends. Their kids Deep, Meen (Bhuri), Meet (Kali), Tallu, Chahat, Charvi and Bholku gave me moments where I forgot all my worries and went back to my childhood days.

I am highly thankful to my family especially my uncle Kirankaka, sister Radha and brother-inlaw Shakti, who took care of all my family duties and also of my needs, which helped keep all my tensions at bay and enabled me to focus on my studies. Without their support this endeavour would not have been this smooth. I am equally thankful to my in-laws Pragna mummyji, Hinal Di and Parth bhai for their blessings and prayers for my success.

This acknowledgement would be incomplete without conveying a HUGE thank you to Binal, my friend-my wife. Your confidence in me is one reason that made me believe in myself. Simply this would not have been possible without your support.

Again thank you all!

# **Table of Contents**

Chapter 1	Introduction	1	
1.1	Inflammation	1	
1.2	Asthma		
1.3	Airway Epithelium		
1.3.1	Bronchial Epithelium	3	
1.3.2	Alveolar Epithelium	4	
1.3.3	Functions of Airway Epithelium	4	
1.3.4	Airway Epithelium in Asthma: Genetics	6	
1.3.5	Airway Epithelium in Asthma: Dysregulated Functions	9	
1.3.6	Airway Epithelium in Asthma: Interactions with Allergens	10	
1.4	Proteinase-Activated Receptor-2 (PAR-2)	13	
1.4.1	PAR-2 in Atopy and Asthma	15	
1.4.2	PAR-2 and Airway Epithelium in Asthma	17	
1.4.3	Regulation of PAR-2 Expression in AECs	20	
1.5	Insulin – An Anti-inflammatory Hormone	23	
1.5.1	Biological Effects of Insulin	24	
1.5.1.1	Metabolic Regulation: PI3K-Akt Pathway	24	
1.5.1.2	Mitogenic Regulation: Ras-ERK1/2 Pathway	25	
1.5.1.3	Regulation of Inflammation	25	
1.5.2	FOXO Transcription Factors	27	
1.5.3	Insulin Resistance	28	
1.6	Rationale, Hypotheses and Aims	28	
1.7	References	31	
Chapter 2	Insulin decreases expression of the pro-inflammatory receptor	58	
	proteinase-activated receptor-2 on human airway epithelial cells		
2.1	Abstract	58	
2.2	Introduction	60	
2.3	Materials and Methods	62	

2.4	Results	69
2.5	Discussion	80
2.6	References	84
2.7	Supplementary Figures	92
Chapter 3	Effect of insulin on proteinase-activated receptor-2 expression in	94
	the lungs <i>in vivo</i>	
3.1	Aim: To study PAR-2 expression in the lungs of mice with lower levels	94
	of circulating insulin	
3.2	Introduction	94
3.3	Methods	95
3.4	Results and Discussion	96
3.5	References	99
Chapter 4	Effect of insulin on proteinase-activated receptor-2 expression in	101
	primary human airway smooth muscle cells	
4.1	Aim: To study the effect of insulin on PAR-2 expression in human	101
	airway smooth muscle cells	
4.2	Introduction	101
4.3	Methods	102
4.4	Results and Discussion	103
4.5	References	105
Chapter 5	Growth media supplements regulate proteinase-activated receptor-	107
	2 expression in human bronchial airway epithelial cells	
5.1	Abstract	107
5.2	Introduction	109
5.3	Materials and Methods	111
5.4	Results	116
5.5	Discussion	130
5.6	References	134

Chapter 6	Effect of cellular stress on proteinase-activated receptor-2	141
	expression in primary human airway epithelial cells	
6.1	Aim: To study the effect of oxidative stress, hypoxia, cyclic stretch and	141
	human rhinovirus on PAR-2 expression in primary human AECs	
6.2	Introduction	141
6.3	Method	142
6.4	Results and Discussion	144
6.5	References	148
Chapter 7	Effect of media supplement deprivation on the expression of G	151
	protein-coupled receptors in primary human AECs	
7.1	Aim: To study effect of media supplement deprivation on the expression	151
	of G-coupled receptors in primary human airway epithelial cells (AECs)	
7.2	Introduction and rationale for selecting PAR-1, ADRB2 and P2RY2 for	151
	study	
7.3	Methods	154
7.4	Results and Discussion	155
7.5	References	159
Chapter 8	General Discussion	164
8.1	Discussion and Future Directions	164
8.2	Conceptual Model	174
8.3	References	175
	Appendix 1	184
	Appendix 2	185
	Appendix 3	188

# List of Tables

Table 1.1	Asthma susceptibility genes identified in AECs, the method of identification	7
	and their functions	
Table 1.2	Regulation of PAR-2 expression by different stimuli in cells other than AECs	21
Table 1.3	Regulation of PAR-2 expression by different stimuli in AECs	23

# List of Figures

Fig. 1.1	Airway epithelium interaction with allergen: innate and adaptive immune		
	responses leading to allergic sensitization and allergic airway inflammation		
Fig. 1.2	Model for PAR-2 activation	14	
Fig. 1.3	Biological functions of PAR-2-mediated activation of AECs	18	
Fig. 1.4	Canonical insulin signaling pathway	26	
Fig. 2.1	Plasma insulin levels, Akt phosphorylation in lungs and PAR-2 expression in	70	
	lungs of HFD- and control diet-fed mice		
Fig. 2.2	Insulin transcriptionally regulates PAR-2 mRNA expression in AECs	72	
Fig. 2.3	Insulin regulates PAR-2 protein expression in AECs	74	
Fig. 2.4	Insulin regulates PAR-2-mediated activation of AECs	75	
Fig. 2.5	Insulin regulates PAR-2 expression through PI3K activation	76	
Fig. 2.6	Insulin regulates PAR-2 expression by controlling FOXO-1 transcription	78	
	factor in AECs		
Fig. 2.7	Differentiated cultures of AECs show similar insulin-mediated PAR-2	79	
	regulation		
Supple-	Insulin deficiency upregulates PAR-2 expression in AECs	92	
mentary			
Fig. 2.E1			
Supple-	PAR-2 activation induces intracellular Ca <sup>++</sup> release from endoplasmic	93	
mentary	reticulum (ER) in AECs		
Fig. 2.E2			
Fig. 3.1	Effect of overnight fasting on insulin levels in serum and PAR-2 expression	96	
	in lungs		
Fig. 4.1	Effect of insulin on PAR-2 expression in airway SMCs	104	
Fig. 5.1	Media supplements regulate PAR-2 expression in submerged and	117	
	differentiated cultures of NBECs		
Fig. 5.2	Effect of different media supplements on PAR-2 expression in NBECs	119	
Fig. 5.3	Role of PI3K and FOXO-1 pathway in media supplement deprivation-	120	
		xiii	

induced PAR-2 upregulation

Fig. 5.4	Role of ERK1/2 in media supplement deprivation- induced PAR-2	121
	upregulation	
Fig. 5.5	Media supplement deprivation transcriptionally regulates PAR-2 mRNA	123
	expression in NBECs	
Fig. 5.6	Media supplements deprivation upregulates functional PAR-2 expression	125
Fig. 5.7	Effect of media supplement deprivation on cell cycle and apoptosis in	126
	NBECs	
Fig. 5.8	BECs from asthmatic individuals express increased PAR-2 mRNA and show	128
	similar media supplement-mediated PAR-2 regulation	
Fig. 6.1	Effect of oxidative stress on PAR-2 expression in AECs	144
Fig. 6.2	Effect of hypoxia on PAR-2 expression in AECs	145
Fig. 6.3	Cyclic stretch and HRV infection do not modulate PAR-2 expression in	147
	AECs	
Fig. 7.1	Effect of media supplement deprivation on the expression of PAR-1,	155
	ADRB2, P2RY2 and TLR-3 receptors	
Fig. 8.1	Conceptual Model	174

## List of Abbreviations

ABECs -	asthmatic bronchial epithelial cells			
ADRB2 -	Beta-2 adrenergic receptor			
AECs -	airway epithelial cells			
AHR -	airway hyperresponsiveness			
ALI -	air-liquid interface			
BEBM -	bronchial epithelium basal media			
BECs -	bronchial epithelial cells			
BEGM -	bronchial epithelial growth media			
BPE -	bovine pituitary extract			
DCs -	dendritic cells			
ECM -	extra-cellular matrix			
EMT -	epithelial-mesenchymal transition			
ER -	endoplasmic reticulum			
ERK1/2 -	extracellular signal-regulated kinase 1/2			
FOXO-1 CA -	FOXO-1 CA - FOXO-1 constitutively active			
FOXO-1 -	forkhead box protein O1			
GAPDH -	glyceraldehydes 3-phosphate dehydrogenase			
GPCRs -	G- protein coupled receptors			
HDM -	house dust mite			
HFD -	high fat diet			

- HPRT hypoxanthine-guanine phosphoribosyltransferase
- IRS-1 insulin receptor substrate-1
- JNK c-Jun N-terminal kinase
- LPS lipopolysaccharide
- NBECs normal bronchial epithelial cells
- P2RY2 P2Y purinoceptor 2
- PAR-1 proteinase-activated receptor-1
- PAR-2 AP PAR-2 activating peptide
- PAR-2 CP PAR-2 control peptide
- PAR-2 proteinase-activated receptor-2
- PARs proteinase-activated receptors
- PI3K phosphoinositide 3-kinase
- RA retinoic acid
- SAPK stress-activated protein kinase
- SMCs smooth muscle cells
- TEER Trans Epithelial Electrical Resistance
- Th2 type 2 helper T
- TI Alveolar type I cells
- TII Alveolar type II cells
- TLR-3 toll-like receptor 3
- TNF tumor necrosis factor

# **Chapter 1: Introduction**

### **1.1 Inflammation**

Inflammation is an initial protective immune response of tissue against injury and infection. Aulus Celsus first defined inflammation, by a combination of clinical sign and symptoms, as rubor et tumor cum calore et delore (redness and swelling with heat and pain) (1). Rudolf Virchow later added the fifth sign of inflammation: function laesa (loss of function) (1). The basic roles of inflammation are (i) to eliminate the injurious agent (ii) if elimination is not possible, then wall off the injurious agent and (iii) repair or replace the damaged tissue. With advances in cell biology and microscopy, we now have a better understanding of different types of inflammation, namely acute inflammation and chronic inflammation, which do not only differ by the time of onset or by the duration they last but also by the different types of cells involved. The defence mechanisms that the involved cells employ for the elimination of injurious agents also damage the body tissues. Thus, the longer the inflammation persists, the more damage is caused to the tissue which is observed in chronic inflammatory diseases of different tissues. So it is equally important to resolve the inflammation in a timely manner. During the resolution process, synthesis of pro-inflammatory mediators stops, which ceases the recruitment of inflammatory cells at the site of inflammation. Also the already recruited cells either go back to systemic circulation or undergo apoptosis; apoptotic cells are later cleared by phagocytes. If this homeostatic process is dysregulated, it results in increased levels of inflammatory mediators at the site of inflammation which recruits high number of inflammatory cells ultimately resulting in "tissue-damaging" chronic inflammation. We will now focus on asthma, a chronic inflammatory disease of the airways of the lung tissue.

### 1.2 Asthma

Asthma was first defined by Henry Salter in 1860 as "Paroxysmal dyspnoea of a peculiar character with intervals of healthy respiration between attacks" (2). Today asthma is defined as an airway disease characterized by episodes of shortness of breath and wheezing due to reversible airway obstruction that develops as a result of airway inflammation and airway hyperresponsiveness (AHR). According to the World Health Organization over 300 million

people suffer from asthma, with 250,000 deaths every year (<u>http://www.who.int/gard/publicat</u> <u>ions/GARD%20Book%202007.pdf</u>). According to a 2014 report from Statistics Canada, close to 2.5 million Canadians suffer from asthma (<u>http://www.statcan.gc.ca/tables-tableaux/sum-</u> <u>som/l01/cst01/health50a-eng.htm</u>) with a total burden of over \$4 billion in direct and indirect costs annually (<u>http://www.health.gov.on.ca/en/common/ministry/publications/reports/asthma</u> /asthma2.aspx).

Asthma is a complex disease involving interactions between genetics and environment (3). We now understand that the majority of asthma is atopic, meaning that an asthma attack occurs in allergic individuals upon contact with an allergen. There is genetic predisposition to atopy but the development of allergic sensitization also requires environmental factors such as exposure to a particular allergen. The inhaled air is one of the main sources for exposure to pathogens, allergens (aeroallergens) and pollutants (4). Airway epithelium acts as a physical barrier to prevent the invasion of the inhaled air constituents but also acts as an innate immune cell. The innate immune responses of airway epithelium play an important role in the development of allergic sensitization (5). Moreover, genome wide association studies, aiming at identifying genes that increase susceptibility to asthma development, have revealed genes that are primarily expressed in airway epithelium (6). These results have now set the airway epithelium as a prime interest in the pathogenesis of asthma.

### 1.3 Airway Epithelium

Throughout the body, epithelium lines the surface of organs and cavities; hence acts as a mechanical barrier against the external environment, for example airway epithelium against inhaled air, gut epithelium against ingested food and skin epithelium against atmospheric air. The type of cells, namely squamous, cuboidal and columnar defines the epithelium. Further epithelium can be comprised of a single layer (simple epithelium) or multiple layers (stratified epithelium) of cells. An additional type of epithelium is called pseudostratified epithelium which in fact is a single layer of cells but appears as multiple layers due to presence of cell nuclei at different heights.

The airway epithelium belongs to the pseudostratified columnar epithelium, where each cell of single layer of columnar epithelial cells is in contact with basal lamina. There are a variety of specialized cells within the airway epithelium and their distribution varies throughout the airways. For example, the most abundant cells in the bronchial epithelium of the large airways are basal cells, ciliated cells and secretory goblet cells. These cells gradually decrease as we approach the smaller airways. The secretory club cells, also known as Clara cells, are more abundant in the smaller airways. The alveolar epithelium is made up of type I and type II alveolar epithelial cells.

### **1.3.1 Bronchial Epithelium**

### Basal cells

Basal cells are present in large and smaller airways (7), however their numbers decrease in the smaller airways (8). They are relatively undifferentiated and can be identified by expression of the transcription factor p63 and cytokeratin 5 and 13 (9). These cells are considered as the main progenitor cells in the airways that possess stem cell-like self-renewal properties and also differentiate into goblet and ciliated epithelial cells (10). Further, these are the only cells in the airways that express hemidesmosomes and are thus tightly attached to the basement membrane via integrins.

### Ciliated cells

Ciliated cells are present in high numbers accounting for over 50% of airway epithelial cells. These cells are characterized by the presence of cilia on apical side (up to 300 cilia/ cell) and a high number of mitochondria below the apical membrane (11) reflective of the high need of energy for the constant ciliary beating. Basal cells, goblet cells and club cells differentiate into ciliated cells (12).

### Goblet cells

As secretory cells of the large airways goblet cells secrete mucus, a mixture of highly glycosylated mucin proteins. The mucus layer is mainly present in the upper airways and it is replaced by the surfactant in the lower airways as the number of goblet cells decreases and club

cells increases. Goblet cells also have self-renewal capacity and can also differentiate into ciliated cells (8).

## Club cells

Club cells are surfactant-secreting cells that are present in the smaller airways and can be characterized by electron-dense granules (11). These cells also possess stem cell-like self-renewal property and can differentiate into ciliated and goblet cells (13). Recent evidence suggest that Club cells can dedifferentiate into basal-like cells (14) or can also give rise to type I and type II alveolar epithelial cells (15).

## 1.3.2 Alveolar Epithelium

## Type I cells (T1)

These very thin cells cover over 95% of the internal surface of alveoli. Their morphology helps create a thin barrier between air and blood and thus enables gas exchange (16).

## Type II cells (TII)

The presence of secretory granules known as lamellar bodies, that store surfactant, is a characteristic of these cuboidal shaped cells (16). TII are secretory cells of alveoli region of the lung that synthesize and secrete surfactant and also maintain the levels of surfactant by reuptake (17). Upon injury, these cells act as progenitor cells to reconstitute alveolar epithelium (18).

### 1.3.3 Functions of Airway Epithelium

Since the airway epithelium is at an interface between external environment and lung tissue, it is the first site of interaction with the constituents of inhaled air including pathogens, aeroallergens, and pollutants. This interaction with external environment demands several important functions for epithelium. To execute these functions, epithelium is equipped with different types of specialized cells, as described above, which differ based on the type of tissue, for example types of specialized cells present in airway epithelium vs. gut epithelium are different.

#### Physical barrier property

The barrier property of epithelium is important for preventing entry of pathogens or particles into the subepithelial mucosa and thus limiting unnecessary activation of the immune system. The barrier property is attributed to the apical-junctional complex (AJC), which includes tight junctions, adherens junctions and desmosomes. The tight junctions are the AJC made up of tight junction proteins such as occludin, claudin and zonula occludens, which create a semipermeable barrier that regulates paracellular transport of solute across the epithelium (11). The tight junctions and adherens junctions also help polarize the epithelium (19). Adherens junctions and desmosomes, through cell-cell adhesion, maintain structural integrity of epithelium (20, 21). Club cells also help maintain integrity of epithelium by secreting antiproteinases such as secretory leukocyte protease inhibitor that neutralizes harmful proteinases (22). Finally interaction of epithelial cells, through integrins, with underlying extracellular matrix not only helps in maintaining mucosal barrier but also provides signals for epithelial cell survival and proliferation (23).

### Mucocilliary escalator

Goblet cells and ciliated cells work in collaboration to provide this defense mechanism. The mucus released by goblet cells traps the inhaled pathogen or particles, which is then cleared by the highly coordinated ciliary beating.

### Gas exchange

The thin layer of surfactant maintained by TII cells at the alveolar epithelium regulates surface tension and facilitates gas exchange by preventing alveoli from collapse (11). Surfactant secreted by club cells reduces surface tension of the bronchioles and helps them prevent their collapse during expiration. Interestingly, Club cells are shown to produce p450 mono-oxygenases which can metabolize xenobiotic compounds such as aromatic hydrocarbons (24) dissolved in surfactant fluid, thus maintains the purity of surfactant fluid.

### Epithelial restitution

Airway epithelial cells (AECs) exhibit a rapid rate of turnover. Moreover, since they are exposed to external environment they are more prone to encounter environmental insults and experience epithelial tissue damage. It is very important to re-establish the homeostasis as soon as possible by reconstruction of normal airway. The current literature does not indicate the presence of a single stem cell type that can differentiate into all the other types of lung epithelial cells. However, as described above, multiple cell types in the airways have stem cell-like properties that undergo differentiation, transdifferentiation or dedifferentiation (epithelial-mesenchymal transition) to give rise to the different cell types in the airway to repair the wound and establish the homeostasis.

### Airway epithelium: an immunologically active barrier

Being the front line soldiers, it is very important that AECs sense invading pathogens and mount an immune response against them. Moreover, it is equally important that AECs alert the immune system about the danger that they have encountered. The activation of the receptors on AECs not only results in the release of anti-microbial or anti-viral mediators but immune mediators such as cytokines and chemokines that recruit and activate other immune cells (25).

There is increasing evidence implicating the airway epithelium as a critical regulator of allergic sensitization and allergic airway inflammation. Allergic sensitization and allergic airway inflammation are involved in the pathogenesis of asthma. The above mentioned functions of airway epithelium are dysregulated in asthma, which puts airway epithelium at the focal point for the pathogenesis of asthma. It is still not clear that the observed dysregulation in structure and functions of airway epithelium is the result of genetics or airway-environment interactions.

### 1.3.4 Airway Epithelium in Asthma: Genetics

Studies aiming at understanding the genetics of asthma have shown a variety of genes associated with asthma susceptibility; many of these asthma susceptibility genes are expressed in airway epithelium (Table. 1.1).

<u>Table 1.1: Asthma susceptibility genes identified in AECs, the method of identification and their</u> <u>functions</u>

Gene	Method	Relevant function in asthma	Reference
ADRB2	Candidate gene approach	Bronchial smooth muscle relaxation (26)	(27)
TNF		Allergic sensitization, airway inflammation and hyperresponsiveness (28-30)	(27)
ADAM33		Associated with bronchial hyperresponsiveness (31)	(31)
PCDH1	Positional cloning	Epithelial differentiation and barrier function (32)	(33)
DPP10		Not known	(34)
HLA-G		Associated with bronchial hyperresponsiveness (35)	(35)
GPRA		Cell homeostasis (36)	(37)
SPINK5		Serine proteinase inhibitor (38)	(39)
IRAK-M		Inhibition of TLR signaling and inflammation (40)	(41)

IL1RL1/IL18R1		Receptor for IL-33 and IL-	(44-48)
		18 involved in the	
		development of Th2	
		inflammation (42, 43)	
н 22		Th2 inflammation (43) and	(44-48)
IL-33		impaired tolerance (49)	
HLA-DQ		Not known	(45, 47, 48)
		Signaling molecule for	
SAMD3		TGF $\beta$ and plays role in	(45, 47, 48)
	Genome-wide analysis	TGFβ-induced EMT (50)	
	study		
		Release of asthma related	
		mediators and activation	
OPMDI 3/GSDMB		of unfolded protein	(45-47)
OKWIDL5/OSDIVID		response pathway (51),	
		Sphingolipid metabolism	
		(52)	
TSLP		Th2 cell polarization and	(46, 48)
		Th2 inflammation (53)	(,)

As summarized in the Table 1.1, many of the identified genes are involved in the regulation of airway epithelial structure and functions, indicating that dysregulation of these epithelial properties could play a central role in asthma pathogenesis.

These studies focusing on understanding genetic component of asthma are association studies. Gene polymorphisms are shown to affect gene expression in tissue-specific manner (54) and expression of many of these asthma susceptibility genes, for example ADAM33 (55), IRAK-M (40), TSLP (56), is not limited to airway epithelium. Thus it is important to understand the effect

of the identified polymorphisms on gene expression and function specifically in airway epithelium. Efforts are already initiated in these directions by correlating the polymorphism and gene expression in AECs with disease phenotype (57, 58), which give more strength to the idea that airway epithelium is an important player in asthma pathogenesis.

#### 1.3.5 Airway Epithelium in Asthma: Dysregulated Functions

There is evidence that secretory functions of airway epithelium are altered in asthma. Asthmatic AECs are shown to express higher levels of inflammatory mediators constitutively (59-63), and also upon injury and infection (64, 65) or allergenic stimulation (66, 67), which reflects the increased pro-inflammatory potential of airway epithelium in asthma. The other properties of airway epithelium which are altered in asthma are cellular differentiation and regeneration. Studies have shown that ciliated cells are decreased in asthmatic airway epithelium (68), which may result in inefficient removal of inhaled pathogen or particles from the airways. On the other hand, the number of undifferentiated basal cells (63, 64) and mucus producing goblet cells (68, 69) are increased. The latter is associated with altered composition of mucus (70) and also increased levels of mucus (69) in asthma, which could contribute to airway obstruction and airway hyperresponsiveness (71), the characteristic features of asthma. The regeneration capacity of airway epithelium is important for wound healing after an injury. In vivo wound healing requires cell proliferation and epithelial-mesenchymal transition (EMT). During EMT, differentiated AECs undergo dedifferentiation and secrete extra-cellular matrix (ECM) to cover the wounded area, which facilitates the tissue repair process. However, if not regulated, this process could result in increased deposition of ECM and airway remodeling, another characteristic feature of asthma. Asthmatic AECs are less efficient at proliferation (65) and wound healing (65, 72) upon mechanical wounding. Moreover, in vitro cultures have shown that more number of asthmatic AECs, compared to non-asthmatic AECs, undergo EMT upon stimulation (50) indicating dysregulated epithelial repair process in asthma.

Many of these studies including (73) demonstrate that asthmatic AECs retain these altered characteristics when cultured *in vitro* suggesting a possibility of asthma-induced epigenetic changes in airway epithelium. This idea is supported by the studies showing differences in the

expression of epigenetic modifying enzymes (74) and epigenetic changes, such as DNA methylation, in AECs obtained from asthmatic individuals compared to healthy individuals (75).

Dysregulated functions of airway epithelium also increase epithelial interactions with allergens. There is evidence that mucocilliary clearance is not effective in the airways of asthma patients (76, 77), which may result in prolonged presence of the allergens in the airways allowing them to have protracted effects. In addition, decreased expression of anti-proteinases, for example, secretory leukocyte protease inhibitor (SLPI) by airway epithelium in asthma (78) renders barrier property of the epithelium more susceptible to the allergens containing proteinase activity (79-82). Finally, barrier property of airway epithelium is compromised in asthma as observed by decreased expression of proteins involved in the formation of tight junctions (83, 84) and adherens junctions (64, 84, 85) resulting in increased epithelial permeability. However, it is still not clear that the observed barrier dysfunction in asthmatic airway epithelium is the cause for asthma or an outcome of the inflammatory mediators present in asthma since the mediators released by AECs, such as tumor necrosis factor (TNF) (86), vascular endothelial growth factor (VEGF) (87) and thymic stromal lymphopoietin (TSLP) (88), or by other cells such as IL-13 (85) and IL-4 (89), are shown to affect the barrier property of airway epithelium by decreasing the expression of junction proteins. Ultimately, inefficient mucocilliary clearance and increased epithelial permeability facilitate allergen access to subepithelial mucosa. Allergens interact with a variety of immune cells, present in subepithelial mucosa, resulting in a strong immune response, which otherwise would not have occurred if the epithelial functions were optimal.

### 1.3.6 Airway Epithelium in Asthma: Interactions with Allergens

Being a first line of defense, immune responses generated by epithelium upon interactions with allergens play a crucial role in deciding the fate of the interactions. The first step of the interaction is the recognition of allergen by AECs. AECs express an array of pattern recognition receptors (PRRs) that sense the invading allergens. On top of the dysregulated epithelial functions, the other factor that affects the outcomes of epithelial-allergen interactions is the properties of allergens. For example, many of the asthma-relevant allergens from house dust mite (HDM), cockroach and fungi contain proteinase activity that activates pro-inflammatory proteinase-activated receptor-2 (PAR-2), which will be discussed in detail in the next section.

These airway epithelium-allergen interactions result in increased epithelial permeability and release of inflammatory mediators from epithelium, which activate downstream innate immune system. This immune response leads to the development of tolerance or allergic sensitization, however the factors that make the decision of sensitization development over tolerance is under investigation. Also interactions of airway epithelium with allergens after the development of sensitization result in the release of mediators that recruit immune cells resulting in allergic inflammation.



Fig. 1.1: Airway epithelium interaction with allergen: innate and adaptive immune responses leading to allergic sensitization and allergic airway inflammation. Modified from *Gandhi VD* and Vliagoftis H (2015) Front. Immunol. 6:147

AECs express PRRs (90) and PAR-2 (91); activation of these receptors by allergens results in the release of inflammatory mediators from AECs that shape the innate and adaptive immune responses towards the development of allergic sensitization and allergic airway inflammation (Fig. 1.1).

As depicted in Fig. 1.1, interactions of AECs with allergens decrease epithelial integrity resulting in allergens' entry across the epithelium. Simultaneously AECs also recruit DCs to the airways by releasing CCL2 and CCL20, where dendritic cells (DCs) take up the antigen and migrate to lymph nodes where they interact with and influence differentiation of T cells. Differentiation of T cells into type 2 helper T (Th2) cells is a vital step in the development of allergic sensitization. The mediators that affect this DC-T cell interaction and play important roles in the development Th2 cells are IL-33, TSLP, granulocyte-macrophage colony-stimulating factor (GM-CSF) and IL-25; AECs release these mediators upon interactions with allergens. Recent evidence suggests that IL-33 dysregulates lung Treg cells and impairs the development of tolerance to inhaled allergens (49). IL-33 (92) and TSLP (53) influence the DC-T cell interaction by inducing expression of OX40 ligand on DCs, which results in the development of T cell into Th2 cells (93). Airway epithelium-induced GM-CSF is also shown to play a vital role in the development of Th2 cells (94, 95). Th2 cells and type 2 innate lymphoid cells (ILC-2) cells, which are activated by IL-33 and IL-25, release Th2 mediators such as IL-4, IL-5 and IL-13 (96, 97). The released IL-4 and IL-13 act on B cells to produce allergen specific antibodies, which bind to mast cells resulting in the allergic sensitization (98). IL-4 (99-101) and IL-13 (102) also act on AECs to induce the release of mucus and other inflammatory mediators resulting in the development of the vicious cycle of inflammation.

Eosinophilic infiltration of the airways is one of the cardinal features of allergic airway inflammation (103). The mediators released by airway epithelium IL-33, IL-25 and by Th2 cells IL-5 also induce eosinophilia (104-106). Finally, AEC-allergen interactions release chemoattractants for the innate immune cells eosinophils and neutrophils, such as CXCL8, CCL5 and CCL11, and also release GM-CSF, which increases production and survival of granulocytes resulting in airway inflammation.

Apart from the mediators involved in allergic sensitization and allergic inflammation, airway epithelium releases mediators involved in airway remodeling, a prominent feature of asthma. Transforming growth factor- $\beta$  (TGF- $\beta$ ) (107, 108) and leukotriene C4 (LTC4) (108, 109) play a vital role in airway remodeling by inducing proliferation of airway smooth muscle cells (SMCs) and fibroblasts. Moreover TGF- $\beta$  induces differentiation of AECs (110) and fibroblasts (111) to myofibroblasts, a cell type involved in the production of ECM; uncontrolled deposition of ECM is a feature of airway remodeling. Matrix metalloproteinase-9 (MMP-9) also plays a role in airway remodeling not only by degrading ECM (112) but also by activating TGF- $\beta$  (113), which is released as a latent form.

#### 1.4 Proteinase-Activated Receptor-2 (PAR-2)

Proteinase-activated receptors (PARs) are a family of 7-transmembrane G protein-coupled receptors (GPCRs) that belong to the A rhodopsin-like GPCR subfamily (114, 115). Four members of this family have been identified so far, PAR-1 to PAR-4. All members of the family are activated by serine proteinases and have a unique activation mechanism; serine proteinases cleave the extracellular N-terminal of the receptor and reveal the tethered ligand (TL), which lies within the extracellular N-terminus of the receptors. The revealed TL (new N-terminal of the receptor) then folds and binds to the second extracellular loop of the cleaved receptors initiating conformation changes that activate downstream signaling pathways (116-119) (Fig. 1.2).

PAR-2 was first cloned and identified as a receptor for the serine proteinase trypsin in 1994 (120). Trypsin causes PAR-2 hydrolysis at position R<sup>36/S37</sup> to reveal the TL SLIGKV in humans and at position R<sup>38/S39</sup> to reveal the TL SLIGRL in mice and rats. Examples of serine proteinases that activate PAR-2 signaling are trypsins I, II, and IV, mast cell tryptase, coagulation factors VIIa and Xa, acrosin, granzyme A and kallikreins 2, 4, 6 and 14 (120-128).

Synthetic ligands, called activating peptides (AP), mimic the amino acid sequence of the TL of PAR-2 and activate PAR-2 without the requirement for proteolysis (Fig. 1.2); control peptides (CP), containing same amino acids as AP but in reverse order, are used as control during the experiments. PAR-2 is activated by APs mimicking TLs revealed by trypsin such as SLIGKV-NH<sub>2</sub> for human PAR-2 and SLIGRL-NH<sub>2</sub> for mouse or rat PAR-2 (129). Various other APs have

also been used to activate PAR-2. These correlate to TLs generated by proteinase-dependent PAR-2 activation in different species or are substituted peptides with better affinity and/or potency than the native TLs (129). AP-mediated PAR-2 activation mimics the effects of proteinases and has been used widely as an alternative to proteinase-induced activation.



Fig. 1.2: Model for PAR-2 activation

PAR-2 activation results in the activation of multiple G protein-dependent and independent signaling pathways. PAR-2 activation leads to  $G\alpha_s$ -dependent formation of cAMP (130),  $G\alpha_{12/13}$ mediated increase in Rho-kinase activity (130) and  $G\alpha_q$  mediated Ca<sup>++</sup> mobilization (121).
Further, PAR-2 activation results in the recruitment of  $\beta$ -arrestin-1 and -2 (131) to the receptor followed by ERK1/2 phosphorylation (132, 133) and receptor internalization and degradation (121).
Thus activation of PAR-2 is an irreversible proteolytic phenomenon mediated by serine proteinases.

PAR-2 has been studied extensively over the last 15 years because it was identified as a major player in inflammatory reactions in many tissues and organs. PAR-2 expression is widespread on cells in different tissues where PAR-2 activation is shown to induce inflammation, such as, colon (134, 135), skin (136, 137), liver (138) and joints (139) and also shown to play a role in

inflammatory pain sensing (140). PAR-2 is implicated in the pathogenesis of inflammatory respiratory diseases such as allergic asthma (141, 142), idiopathic pulmonary fibrosis (IPF) (143) and pulmonary arterial hypertension (PAH) (144).

### 1.4.1 PAR-2 in Atopy and Asthma

In the airways PAR-2 recognizes both endogenous and exogenous "danger" signals; these signals are serine proteinases released from inflammatory and other cells or inhaled through the air, respectively. In this respect PAR-2 has functional similarities to pattern recognition receptors. These proteinase danger signals, through PAR-2, activate the innate immune system and lead to pro-inflammatory effects. In many cases the same danger signals act through PAR-2 to activate the adaptive immune system. Animal models have shown that PAR-2 activation by inhaled exogenous allergens with proteinase activity from HDM (145), cockroach (146) and fungi (147, 148) play a vital role in the development of allergic asthma.

The first evidence for an association between PAR-2 activation and asthma came in 1999 when Cocks et al., showed that PAR-2 activation causes relaxation of trachea preparations from many species, including mice and humans, and also protects anesthetised mice from bronchoconstriction in vivo; these effects were mediated by PAR-2-induced release of prostaglandin E2 (PGE2) from AECs (149). However, soon after that observation a series of studies suggested that PAR-2 activation in the airways may have significant pro-inflammatory effects. Using PAR-2 knock out (KO) mice it was shown that PAR-2 activation results in expression of P-selectin and leukocyte rolling on endothelial cells, which is the initial step in leukocyte recruitment to the site of inflammation (150). A detrimental role of PAR-2 in the development of allergic airway inflammation in vivo was first reported in 2002 when Schmidlin et al., showed that PAR-2 KO mice were protected from the development of airway inflammation (141). The role of PAR-2 in allergic airway inflammation was confirmed in a study showing decreased accumulation of eosinophils and eosinophil chemotactic factors in the airways and lungs of PAR-2 KO mice (151). Since then the role of PAR-2 is reproduced using PAR-2 KO (152) or PAR-2 blockade (153, 154) models and using intranasal administration of biologically relevant allergens.

PAR-2 KO mice generated less IgE (141, 152) compared to wild type mice after the exposure to allergen, indicating that PAR-2 may be involved in the development of allergic sensitization. Using ovalbumin (OVA) as an innocuous antigen, our group showed that immune tolerance develops when an innocuous antigen comes in contact with the nasal mucosa but simultaneous PAR-2-mediated activation of the nasal and/or airway mucosa in the presence of an innocuous antigen acts as an adjuvant to develop allergic sensitization against the innocuous antigen (28). Inhaled allergens are taken up by DC in the airways to be presented to the immune system and in the absence of other danger signals this process leads primarily to immune tolerance and not allergic sensitization. In the same study, we also showed that PAR-2 activation increases capacity of lung DC to carry antigen and increases their ability to migrate to lymph nodes (28). Thus PAR-2 activation can be a danger signal that biases the immune response to the inhaled innocuous antigen / allergen towards allergic sensitization.

OVA does not contain proteinase activity hence cannot activate PAR-2 but these observations raised an interesting hypothesis that when innocuous antigen / allergens have intrinsic proteinase activity to activate PAR-2 they induce local inflammation and release inflammatory mediators or danger signals to bias the immune response to the inhaled allergen towards allergic sensitization. Our lab has shown, using allergens containing proteinase activity that PAR-2 activation leads to the development of allergic sensitization (153, 154). Our group has also shown that PAR-2 blockade, during allergen challenge of sensitized animals, prevents the development of allergic airway inflammation and AHR (142) and airway remodeling (155). These results indicate that PAR-2 is involved in the development of allergic sensitization and allergic airway inflammation.

In summary, PAR-2 KO and PAR-2 blockade studies have confirmed the involvement of airway PAR-2 in the development of allergic sensitization and allergic inflammation. In the airways, PAR-2 is expressed on bronchial SMCs (156, 157), fibroblasts (158) and lung microvascular endothelial cells (157, 159, 160) and AECs (157). Also, PAR-2 is expressed on tissue resident cells such as mast cells (161), DC (162), and macrophages (163) and on the cells which are recruited to the airways during inflammatory reactions such as eosinophils (164) and monocytes (165). The *in vivo* studies discussed have not shown the specific PAR-2-expressing cell types involved in these PAR-2 effects. It is also possible that PAR-2 activation on different cell types

is primarily responsible for different stages of the development of PAR-2-mediated allergic sensitization and allergic inflammation. To identify this, cell specific PAR-2-knock out models, where PAR-2 expression is eliminated in the airways from a particular type of structural cell or eliminated from a particular type of immune cell, can be used. Until then the only studies that give us clues on the role of PAR-2 on specific cells are *in vitro* studies with isolated or cultured cells, which suggest an important role of PAR-2 activation on airway epithelium for the asthma pathogenesis.

#### 1.4.2 PAR-2 and Airway Epithelium in Asthma

The outcomes of PAR-2 activation on AECs could play an important role in the development of allergic sensitization (Fig. 1.3). PAR-2 activation in AECs decreases the expression of gap junction protein connexin 26 (166) and also destabilizes a junction protein E-cadherin by transactivation of epidermal growth factor receptor (EGFR) (167). This result in decreased epithelial integrity and increased allergen access to underlying immune cells including DCs. DCs are the main antigen presenting cells in the body and are required for development of allergic sensitization (168). PAR-2 activation on AECs releases chemotactic factors for DCs airways (169) resulting in DCs recruitment to the airways. Once allergen gets captured and processed by the DCs, DCs then migrate to lymph nodes where they interact with and influence T cells differentiation. PAR-2 activation on AECs releases GM-CSF (95, 170, 171), TSLP (172) and IL-25 (173) that provide a favorable microenvironment for the development of allergic sensitization by supporting DC maturation and activation which in turn influence DC-T cell interaction and result in T cell differentiation into Th2 cells, the T cell phenotype that plays a role in the development of allergic sensitization are depicted in Fig. 1.1.

Increased in the numbers of eosinophils in the airways is one of the cardinal features of allergic airway inflammation (103). PAR-2 could contribute to this by inducing the release of CCL11, which acts as a eosinophil chemotactic factor, and GM-CSF, which increases the survival of eosinophils (170). Further, PAR-2 activation on AECs also releases mediators that attract cells involved in innate and adaptive immune responses in the airways (91, 148, 169-171, 174-176) resulting in airway inflammation (Fig. 1.3).

Finally, airway remodeling is a characteristic feature of asthma, where airways undergo structural changes including increase in airway smooth muscle, subepithelial fibrosis and changes in the cartilage (177). PAR-2 activation on AECs releases leukotriene 4 (LTC4) (108, 109) and matrix metalloproteinase-9 (113, 178), which could play direct or indirect roles in airway remodeling by inducing proliferation of bronchial SMCs and fibroblasts and also by degrading ECM.



Fig. 1.3: Biological functions of PAR-2-mediated activation of AECs

The other mechanism by which PAR-2 activation can perpetuate inflammation is by exhibiting synergy with other inflammatory mediators for AEC activation. Asthmatic airways are characterized by the presence of chronic inflammation; interestingly, inflammatory mediators

present in the airways, such as IL-4 and TNF, could synergize with allergens to amplify the PAR-2-mediated release of inflammatory mediators (172, 179, 180).

The *in vivo* and *in vitro* approaches give complementary and very important results allowing us to reach a unifying consensus about the role of PAR-2 activation on AECs in the pathogenesis of asthma. These *in vitro* studies to understand the effect of PAR-2 activation on AECs have been carried out using both alveolar and bronchial AECs. A549 (human lung carcinoma cells with properties of TII alveolar epithelial cells, (181)) is the most utilized cell line to study the effect of PAR-2 activation on alveolar epithelium, while various cell lines, such as BEAS-2B (virus transformed AECs (182)), 16HBE14o-, Calu-3, NCI-H292, and primary airway epithelial cultures have been used to study the effect of PAR-2 activation on the bronchial epithelium. In addition, both immersed cultures and air-liquid interface (ALI) cultures have been used in studies performed with bronchial epithelial cells (BECs). ALI cultures mimic a physiological airway epithelium and facilitate in vitro study of epithelial functions such as barrier property, mucus secretion and mucociliary escalator, which is not possible with immersed cultures. ALI cultures also provide a unique opportunity to study cellular interactions between epithelial cells and other immune cells by using co-culture systems. There is no clear published evidence to indicate that AECs from different locations in the airways respond differently to PAR-2 activation, although this would not be unexpected. Studies to compare cells from different location in the same experimental model will be very useful to better define the role of PAR-2 in different compartments of the airways.

In conclusion, PAR-2-mediated activation of the epithelium leads to the release of proinflammatory mediators including chemotactic mediators for immune cells (neutrophils, eosinophils, DC, T cells), mediators that activate immune cells (survival factors and factors leading to production of pro-inflammatory mediators) and mediators involved in tissue remodeling. These mediators then, directly or indirectly, lead to allergic sensitization, allergic airway inflammation and tissue remodeling.

#### 1.4.3 Regulation of PAR-2 Expression in AECs

PAR-2 expression is increased in inflammatory diseases such as asthma (183-185), atopic dermatitis (137), arthritis (165, 186), idiopathic pulmonary fibrosis (143) and pulmonary hypertension (187), diseases that evidence suggest also depend on PAR-2 activation. In all these cases the reasons for increased PAR-2 expression are not clear, but the consequences may be enhanced inflammation as has been shown in the case of asthma using PAR-2 overexpressing animals (141). Increased PAR-2 expression may also predispose the individuals to development of allergic sensitization, since we have shown that PAR-2 activation can deviate the immune response from tolerance to allergic sensitization (28). Therefore, understanding the mechanism of regulation of PAR-2 expression may allow us to device strategies to decrease PAR-2 expression and PAR-2-mediated inflammation in these conditions.

We have put a lot of effort in our lab to understand the role of PAR-2 in asthma, and especially on AECs, the first line of defense against inhaled inflammatory triggers and an important participant in the pathogenesis of inflammatory lung diseases. In this project my efforts have focused in understanding the regulation of PAR-2 expression on AECs, since there is very little information in this area.

There is evidence that PAR-2 expression is increased on airway SMCs (183, 184) and AECs (185) in patients with asthma. It is however, not known whether PAR-2 upregulation is a cause of the disease or an effect of the inflammatory environment in the airways. The literature indicates that a variety of inflammatory mediator upregulate PAR-2 in other cell types and other systems (summarized on Table 1.2 and Table 1.3). Many of these mediators are increased in the airways of patients with asthma and they may be the cause for PAR-2 upregulation in these patients. Inflamed and edematous airways in asthmatic individuals may also promote hypoxic environment around the epithelium, and hypoxia can also regulate PAR-2 expression in other systems. Changes in asthmatic airways may also promote epigenetic changes that may be responsible for altered PAR-2 expression. Finally genetic variations may also be the cause of altered PAR-2 expression in asthma, which would mean that the increased expression may precede the development of the disease. A PAR-2 coding region SNP has been associated with
atopy in Korean children, a SNP that increases mRNA stability leading to increased PAR-2 expression in peripheral blood mononuclear cells (PBMCs) (188).

Although there is no specific information about PAR-2 regulation on AECs by inflammatory mediators, there is evidence that HDM (166), cockroach (175) and mold (148) allergen may upregulate PAR-2 expression on airway epithelial cells, but the studies lack mechanistic components that will allow us to better understand this regulation. PAR-2 activation, by proteinases, results in rapid internalization of the receptor and decreased PAR-2 expression on the cell surface (189), but the studies mentioned above show increased expression using activators with similar proteolytic activity. More detailed studies will be needed to address these disparities. In addition, LPS can also upregulate PAR-2 expression on AECs (190), and the effect of allergens may be mediated, at least in part, through LPS contaminating the allergen preparation.

Stimuli	Cell culture model	Upregulation	Downregulation	Reference
IL-1β, Trypsin, LPS, PAR-2 AP, IL-6, GM-CSF	Human umbilical vain endothelial cells, P815 mast cells	mRNA		(191-193)
PDGF-BB, TGFβ	Skin fibroblast cells, Lung fibroblast cells	Protein		(194)
IL-1α, IL-1β, TNF, LPS, PDGF-BB, Recombinant cockroach allergen Per	Airway smooth muscle cells, Chondrocytes, Synovial cells, Human umbilical vain endothelial cells, Pulmonary arterial smooth muscle cells,	mRNA & protein		(144, 195- 201)

Table 1.2: Regulation of P	AR-2 expression by	v different stimuli in	cells other than AECs
	1		

a 7,	P815 mast cells			
Recombinant				
cockroach				
allergen Per				
a 1.0101				
Нурохіа	Pulmonary arterial smooth muscle cells, Human umbilical vain endothelial cells	mRNA & protein		(144), (202)
IL-1β, TNF, LPS	Human coronary arteries, Primary rat astrocytes	mRNA & functional response		(203, 204)
TNF	Human umbilical vain endothelial cells	mRNA, protein & functional response		(191)
Extracellular matrix	Skin fibroblast cells, Lung fibroblast cells		Protein	(194)
TGF β, IL-12	Chondrocytes, P815 mast cells		mRNA & protein	(196, 205)
IL-1β	Primary rat astrocytes		mRNA & functional response	(204)

Table 1 3. Regulation	of $PAR_2$ evore	ection by dif	fferent stimul	i in AECs
radic 1.5. Regulation	$011 \Lambda 1^{-2} CAPIC$	ssion by un	increme summu	I III ALCS

Stimuli	Cell culture model	Upregulation	Downregulation	Reference
PAR-2 AP, LPS, Cockroach allergens, Neutrophil elastase, House dust mite allergen Der p 1	A549 cells, Calu- 3 cells, Bronchial epithelial cell line, Human nasal epithelial cells	mRNA		(166, 175, 190, 206)
Purified mold allergen Pen c 13	A549 cells	mRNA & protein		(148)

Thus the available literature supports the notion that inflammatory microenvironment is crucial for PAR-2 regulation, but the mechanisms involved are poorly studied.

# 1.5 Insulin – An Anti-inflammatory Hormone

The  $\beta$  cells of Langerhans islets in the pancreas secrete insulin. Insulin was identified, for its role in glucose homeostasis, and purified by the researchers Banting, Best and Collip, under the guidance of McLeod at the University of Toronto in 1921-1922 (207). Further research has now established the role of insulin in energy metabolism by regulating metabolism of carbohydrate, lipid and protein (208, 209).

Early animal studies also supported an anti-inflammatory role for insulin. Researchers in the 1970s showed that carrageenan-induced rat paw edema was blunted by insulin administration and was augmented in insulin deficient animals (210). However, it was thought that the anti-inflammatory effect of insulin was mediated by lowering blood glucose since hyperglycemia is pro-inflammatory (211). Studies during the last 15 years have shown that anti-inflammatory

properties of insulin can be attributed to its ability to modulate key inflammatory molecules (212).

## 1.5.1 Biological effects of insulin

Insulin exerts its effect by activating multiple signaling pathways upon binding to the insulin receptor. Insulin is also shown to activate another structurally similar receptor insulin-like growth factor-1 receptor (213) albeit with low affinity than the insulin receptor (214). Insulin receptor consists of two extracellular  $\alpha$ -subunits and two intracellular  $\beta$ -subunits and belongs to a class of receptors called receptor tyrosine kinases (RTK). Binding of insulin to the  $\alpha$ -subunits results in receptor dimerization and activation of intrinsic tyrosine-kinase activity of the  $\beta$ -subunits, which then trans-phosphorylate tyrosine residues in  $\beta$ -subunits. These phosphorylated tyrosine residues act as a docking site for adaptor proteins that possess phosphotyrosine-binding (PTB) domains or Src homology 2 (SH2) domains such as insulin receptor substrate (IRS) and Src homology 2 domain containing transforming protein (Shc). It has been shown that these two adaptor proteins compete for the same binding site of the insulin receptor (215, 216). Based on the adapter protein that binds the insulin receptor, different signaling pathways are activated and mediate the regulatory functions of insulin (Fig. 1.4).

### 1.5.1.1 Metabolic Regulation: PI3K-Akt Pathway

Recruitment of IRS to the insulin receptor through the PTB domain (215) activates the PI3K-Akt signaling pathway. Once recruited, tyrosine residues of IRS are phosphorylated by intrinsic tyrosine-kinase activity of the insulin receptor. Tyrosine-phosphorylated IRS binds to phosphatidylinositol 3-kinase (PI3K) through the SH2 domain. There are three classes of PI3K according to their structural characteristics and substrate specificity; class I PI3K are the most commonly studied for insulin signaling as there are no good data for insulin-mediated activation of other classes of PI3K. Class I PI3K are further divided in to Class IA, activated by RTKs, GPCRs and G-protein Ras and Class IB, activated by only GPCRs. The literature suggests that Class IA (217), but not class IB (218, 219) are the main enzymes responding to insulin. PI3K IA contain two subunits: regulatory p85, which has SH2 domain, thus binds and recruits PI3K to IRS (220), and catalytic p110. Activated PI3K acts on the plasma membrane

phosphatidylinositol 4,5-bisphosphate (PIP2) and generates phosphatidylinositol 3,4,5trisphosphate (PIP3) (221), which acts as a docking site for proteins containing PH domain. This results in the recruitment of phosphoinositide-dependent protein kinase 1 (PDK-1) and Akt. PDK-1 phosphorylates Thr308 on Akt, while PDK-2 (mTORC2) phosphorylates Ser473 on Akt. These two phosphorylations activate Akt (serine/threonine kinase) (222-225), which will further regulate the downstream signaling involved in inhibition of apoptosis, uptake of glucose, synthesis of glycogen, fat and protein as energy reserves and prevent glucose production by inhibiting gluconeogenesis (glucose synthesis from non-carbohydrate precursors) and glycogenolysis (breakdown of glycogen to glucose) (226). Thus insulin, through PI3K-Akt pathway regulates cell survival, metabolism and protein synthesis.

### 1.5.1.2 Mitogenic Regulation: Ras-ERK1/2 Pathway

The binding of IRS or Shc to the insulin receptor activate the Ras-ERK1/2 pathway. Phospho-IRS or Phospho-Shc recruit adaptor/guanine nucleotide exchange factor complex Grb2/Sos (227). Sos mediates GDP/GTP exchange on Ras and activates Ras (227, 228), which then activates Raf (229). Raf activates mitogen-activated protein kinase kinase 1/2 (MEK1/2) by phosphorylating it on two serine residues (230). MEK1/2 is a dual specificity kinase that activates extracellular signal-regulated kinase 1/2 (ERK1/2) by tyrosine and threonine phosphorylation (231). Active ERK1/2 (serine/threonine kinase) further phosphorylates many substrates, which are localized in the cytoplasm or nucleus (232). This pathway results in the expression of genes and synthesis of proteins involved in cell growth, differentiation, survival and cell migration (233, 234).

## 1.5.1.3 Regulation of Inflammation

In addition of decreasing pro-inflammatory effects of hyperglycemia, by decreasing glucose levels, insulin has also anti-inflammatory properties by directly modulating expression or activity of pro-inflammatory molecules. Insulin decreases the recruitment of leukocytes by downregulating expression of chemokines (235), chemokine receptors (235) and cell adhesion molecules (236). TLRs are a group of innate immune receptors that identify a range of pathogen-associated molecular patterns to initiate the inflammatory cascade (237). Insulin

downregulates expression of TLRs (238), including TLR-4 (239), which in turn decreases TLR-4-mediated inflammation (240, 241). Insulin can also modulate the activity of transcription factors rather than affecting their expression, for example, insulin suppresses activity of nuclear factor kappa B (NF- $\kappa$ B) transcription factor by preventing its nuclear localization (242, 243); NF- $\kappa$ B upon activation induces inflammation by increasing expression of pro-inflammatory cytokine and cell adhesion molecules (244). In addition to NF- $\kappa$ B, a class of transcription factors, whose functions are directly under the negative regulation of insulin signaling, are the Forkhead box O (FOXO) transcription factors.



Fig. 1.4: Canonical insulin signaling pathway. GSK3, glycogen synthase kinase 3; TSC2, tuberous sclerosis complex 2; BAD, Bcl-2-associated death promoter; FOXO-1, forkhead box O-1; mTORC1, mammalian target of rapamycin complex 1.

#### **1.5.2 FOXO Transcription Factors**

The Forkhead family of transcription factors is comprised of more than 100 members and is characterized by a conserved DNA-binding domain called "Forkhead box" (245). They are named as "Forkhead" because the mutation in the first identified member FOXA resulted in a head structure that resembled a fork in *Drosophila* (245). FOXO transcription factors are part of this family. There are 4 FOXOs namely FOXO-1, FOXO-3, FOXO-4 and FOXO-6, which are regulated similarly by insulin/PI3K/Akt pathway (246).

Active FOXO-1 is localized to the nucleus. Insulin negatively regulates FOXO-1 activity through PI3K-Akt pathway. Insulin-activated Akt phosphorylates FOXO-1 on the conserved residues Threonine 24, Serine 256 and Serine 319 resulting in nuclear exclusion of FOXO-1 and inhibition of FOXO-1-mediated transcription (246). In the cytoplasm, the phosphorylated FOXO-1 is polyubiquitinated and degraded (247).

During low insulin condition, such as fasting, FOXO-1 localizes to the nucleus and induces the expression of FOXO-1-targeted genes involved in activities ranging from gluconeogenesis (248) to inflammation (249, 250) to protection against oxidative stress (251), the stress that could be the result of gluconeogenesis-induced hyperglycemia or inflammation. The majority of the studies highlighting the role of FOXO-1 are from organs which are involved in glucose homeostasis such as liver, adipose tissue and skeletal muscles because of involvement of FOXO-1 in glucose metabolism. The role of FOXO-1 is less studied in epithelial cells and the available literature suggests that FOXO-1 could regulate innate immune functions in epithelial cells from different systems. In keratinocytes, FOXO-1 promoted wound healing in cells cultured in normal glucose media (252, 253), while impaired wound healing in cells cultured in high glucose media (253). This observation may indicate contradicting roles of FOXO-1 in healing of normal wound vs. diabetic wound. In colonic epithelial cells, FOXO-1 is shown to play a role in TNF-mediated impaired barrier function (254). Finally, in AECs, FOXO-1 is activated upon bacterial infections to decrease internalization of bacteria and induce the release of inflammatory mediators and defensins, anti-microbial peptides (255).

### 1.5.3 Insulin resistance

Inflammation can lead to insulin resistance, a condition where insulin is not able to exert its functions because cells do not respond to insulin. Insulin resistance could be a result of decrease in expression of insulin signaling molecules or blockade in insulin signaling. Insulin resistance is best studied in adipose tissue and skeletal muscles, which are the primary sites of insulin action for glucose uptake. Inflammation is a major contributor to insulin resistance by multiple mechanisms. First, inflammatory mediators activate suppressor of cytokine signaling (SOCS) family of proteins that decrease the intrinsic tyrosine phosphorylation activity of insulin receptor (256) and increase the degradation of IRS-1 (257), which is downstream of insulin receptor in the in insulin signaling cascade. Second, inflammatory mediators such as IL-6 (258), IL-1 $\beta$  (259) and TNF (260) decrease the expression of insulin receptor and IRS-1. Finally, inflammation results in activation of serine / threonine kinases such as c-Jun N-terminal kinase (JNK) (261) and protein kinase C (PKC) (262) that phosphorylates IRS-1 on serine residues, which prevents the recruitment of PI3K and results in blockade of insulin signaling. This results in insulin resistance in adipose tissue and skeletal muscle and hence these tissues do not perform insulin-directed functions including glucose uptake.

#### 1.6 Rationale, Hypotheses and Aims

In asthma, PAR-2 expression is increased on airway epithelial cells but the mechanism underlying PAR-2 upregulation as well as the consequences of this upregulation are poorly studied. As we discussed, PAR-2 expression is increased in a variety of inflammatory diseases. It is believed that the inflammatory milieu upregulates PAR-2 expression in these diseases.

(i) Along with metabolic functions, insulin resistance results in blockade of antiinflammatory functions of insulin. Insulin acts anti-inflammatory by decreasing expression of pro-inflammatory receptors such as TLRs, which are activated by a particular molecular pattern. Similar to TLRs, PAR-2 is a pro-inflammatory receptor, which is activated by serine proteinases. PAR-2 expression is increased in adipose tissue during insulin resistance (263-265) and improving insulin sensitivity decreases PAR-2 expression (263) indicating insulin may regulate PAR-2 expression. It is also possible that insulin regulates PAR-2 expression in AECs and upregulated PAR-2 expression in airway epithelium in asthma (185) is due to altered insulin signaling in AECs. This possibility is plausible since asthmatic airways are chronically inflamed and the mediators that induce insulin resistance in adipose tissue, such as IL-6, IL-1 $\beta$  and TNF (266), are also increased in the airways in asthma (267-270). Interestingly, recent evidence has shown decreased expression of insulin receptor in AECs obtained from asthma patients compared to healthy individuals (271) raising the possibility that insulin regulates PAR-2 expression in AECs and the increased PAR-2 expression in AECs in asthma (185) could be the result of decreased insulin signaling in AECs. Finally, an indirect evidence demonstrating potential role of insulin in regulating PAR-2 expression showed that mice lacking insulin had increased PAR-2-mediated inflammation (272). However, PAR-2 expression was not studied in this mouse model it is possible that the increased in PAR-2-mediated inflammation was because of increased PAR-2 expression in the absence of insulin. Based on this evidence, we hypothesized that insulin regulates PAR-2 expression in AECs.

(ii) In asthma, airways are infiltrated by inflammatory cells such as eosinophils (273) and neutrophils (274). In addition of these inflammatory cells, other cells such as monocytes, macrophages, epithelial cells and smooth muscle cells, release chemokines, cytokines and reactive oxygen species (ROS) resulting in the presence of high levels of inflammatory mediators (275) and oxidative stress (276) in the airways of asthma patients. It is also shown that airway remodeling of the subepithelium, which is also observed in asthma, may limit the delivery of cell growth nutrients and oxygen (hypoxia) to the airway epithelium (277-279). There is evidence that inflammatory mediators (196, 198, 203), hypoxia (144, 202) and growth factor deprivation (280) regulate PAR-2 expression on cells other than AECs. We hypothesized that these cellular stressors of the inflammatory microenvironment regulate PAR-2 expression in AECs.

**Aim 1:** To study the effect of insulin on PAR-2 expression in mouse lungs *in vivo* and in primary cultures of human AECs *in vitro* (Chapter 2, Chapter 3 and Chapter 4 of the thesis).

**Aim 2:** To study the effect of oxidative stress, hypoxia and decreased levels of cell growth nutrients on PAR-2 expression in primary cultures of human AECs (Chapter 5, Chapter 6 and Chapter 7 of the thesis).

## **1.7 References**

1. Scott A, Khan KM, Cook JL, Duronio V. What is "inflammation"? Are we ready to move beyond Celsus? Br J Sports Med. 2004;38(3):248-9.

Sakula A. Henry Hyde Salter (1823-71): a biographical sketch. Thorax. 1985;40(12):887 8.

3. von Mutius E. Gene-environment interactions in asthma. J Allergy Clin Immunol. 2009;123(1):3-11; quiz 2-3.

4. Craig TJ. Aeroallergen sensitization in asthma: prevalence and correlation with severity. Allergy Asthma Proc. 2010;31(2):96-102.

5. Lambrecht BN, Hammad H. Allergens and the airway epithelium response: gateway to allergic sensitization. J Allergy Clin Immunol. 2014;134(3):499-507.

6. Cookson W. The immunogenetics of asthma and eczema: a new focus on the epithelium. Nat Rev Immunol. 2004;4(12):978-88.

7. Rock JR, Randell SH, Hogan BL. Airway basal stem cells: a perspective on their roles in epithelial homeostasis and remodeling. Dis Model Mech. 2010;3(9-10):545-56.

8. Evans MJ, Plopper CG. The role of basal cells in adhesion of columnar epithelium to airway basement membrane. Am Rev Respir Dis. 1988;138(2):481-3.

9. Evans MJ, Van Winkle LS, Fanucchi MV, Plopper CG. Cellular and molecular characteristics of basal cells in airway epithelium. Exp Lung Res. 2001;27(5):401-15.

10. Hong KU, Reynolds SD, Watkins S, Fuchs E, Stripp BR. In vivo differentiation potential of tracheal basal cells: evidence for multipotent and unipotent subpopulations. Am J Physiol Lung Cell Mol Physiol. 2004;286(4):L643-9.

11. Tam A, Wadsworth S, Dorscheid D, Man SF, Sin DD. The airway epithelium: more than just a structural barrier. Ther Adv Respir Dis. 2011;5(4):255-73.

12. Ayers MM, Jeffery PK. Proliferation and differentiation in mammalian airway epithelium. Eur Respir J. 1988;1(1):58-80.

13. Hong KU, Reynolds SD, Giangreco A, Hurley CM, Stripp BR. Clara cell secretory protein-expressing cells of the airway neuroepithelial body microenvironment include a label-retaining subset and are critical for epithelial renewal after progenitor cell depletion. Am J Respir Cell Mol Biol. 2001;24(6):671-81.

14. Tata PR, Mou H, Pardo-Saganta A, Zhao R, Prabhu M, Law BM, et al. Dedifferentiation of committed epithelial cells into stem cells in vivo. Nature. 2013;503(7475):218-23.

15. Zheng D, Soh BS, Yin L, Hu G, Chen Q, Choi H, et al. Differentiation of Club Cells to Alveolar Epithelial Cells In Vitro. Sci Rep. 2017;7:41661.

16. Dobbs LG, Johnson MD. Alveolar epithelial transport in the adult lung. Respir Physiol Neurobiol. 2007;159(3):283-300.

17. Wright JR, Hawgood S. Pulmonary surfactant metabolism. Clin Chest Med. 1989;10(1):83-93.

18. Barkauskas CE, Cronce MJ, Rackley CR, Bowie EJ, Keene DR, Stripp BR, et al. Type 2 alveolar cells are stem cells in adult lung. J Clin Invest. 2013;123(7):3025-36.

19. Baum B, Georgiou M. Dynamics of adherens junctions in epithelial establishment, maintenance, and remodeling. J Cell Biol. 2011;192(6):907-17.

20. Schneeberger EE, Lynch RD. Structure, function, and regulation of cellular tight junctions. Am J Physiol. 1992;262(6 Pt 1):L647-61.

21. Green KJ, Jones JC. Desmosomes and hemidesmosomes: structure and function of molecular components. FASEB J. 1996;10(8):871-81.

22. De Water R, Willems LN, Van Muijen GN, Franken C, Fransen JA, Dijkman JH, et al. Ultrastructural localization of bronchial antileukoprotease in central and peripheral human airways by a gold-labeling technique using monoclonal antibodies. Am Rev Respir Dis. 1986;133(5):882-90.

23. Sheppard D. Epithelial integrins. Bioessays. 1996;18(8):655-60.

24. Hukkanen J, Pelkonen O, Hakkola J, Raunio H. Expression and regulation of xenobioticmetabolizing cytochrome P450 (CYP) enzymes in human lung. Crit Rev Toxicol. 2002;32(5):391-411.

25. Gandhi VD, Vliagoftis H. Airway epithelium interactions with aeroallergens: role of secreted cytokines and chemokines in innate immunity. Front Immunol. 2015;6:147.

26. Kotlikoff MI, Kamm KE. Molecular mechanisms of beta-adrenergic relaxation of airway smooth muscle. Annu Rev Physiol. 1996;58:115-41.

27. Ober C, Hoffjan S. Asthma genetics 2006: the long and winding road to gene discovery. Genes Immun. 2006;7(2):95-100.

28. Ebeling C, Lam T, Gordon JR, Hollenberg MD, Vliagoftis H. Proteinase-activated receptor-2 promotes allergic sensitization to an inhaled antigen through a TNF-mediated pathway. J Immunol. 2007;179(5):2910-7.

29. Thomas PS, Heywood G. Effects of inhaled tumour necrosis factor alpha in subjects with mild asthma. Thorax. 2002;57(9):774-8.

30. Whitehead GS, Thomas SY, Shalaby KH, Nakano K, Moran TP, Ward JM, et al. TNF is required for TLR ligand-mediated but not protease-mediated allergic airway inflammation. J Clin Invest. 2017;127(9):3313-26.

31. Van Eerdewegh P, Little RD, Dupuis J, Del Mastro RG, Falls K, Simon J, et al. Association of the ADAM33 gene with asthma and bronchial hyperresponsiveness. Nature. 2002;418(6896):426-30.

32. Koning H, Sayers I, Stewart CE, de Jong D, Ten Hacken NH, Postma DS, et al. Characterization of protocadherin-1 expression in primary bronchial epithelial cells: association with epithelial cell differentiation. FASEB J. 2012;26(1):439-48.

33. Koppelman GH, Meyers DA, Howard TD, Zheng SL, Hawkins GA, Ampleford EJ, et al. Identification of PCDH1 as a novel susceptibility gene for bronchial hyperresponsiveness. Am J Respir Crit Care Med. 2009;180(10):929-35.

34. Allen M, Heinzmann A, Noguchi E, Abecasis G, Broxholme J, Ponting CP, et al. Positional cloning of a novel gene influencing asthma from chromosome 2q14. Nat Genet. 2003;35(3):258-63.

35. Nicolae D, Cox NJ, Lester LA, Schneider D, Tan Z, Billstrand C, et al. Fine mapping and positional candidate studies identify HLA-G as an asthma susceptibility gene on chromosome 6p21. Am J Hum Genet. 2005;76(2):349-57.

36. Vendelin J, Pulkkinen V, Rehn M, Pirskanen A, Raisanen-Sokolowski A, Laitinen A, et al. Characterization of GPRA, a novel G protein-coupled receptor related to asthma. Am J Respir Cell Mol Biol. 2005;33(3):262-70.

37. Laitinen T, Polvi A, Rydman P, Vendelin J, Pulkkinen V, Salmikangas P, et al. Characterization of a common susceptibility locus for asthma-related traits. Science. 2004;304(5668):300-4. 38. Mitsudo K, Jayakumar A, Henderson Y, Frederick MJ, Kang Y, Wang M, et al. Inhibition of serine proteinases plasmin, trypsin, subtilisin A, cathepsin G, and elastase by LEKTI: a kinetic analysis. Biochemistry. 2003;42(13):3874-81.

39. Walley AJ, Chavanas S, Moffatt MF, Esnouf RM, Ubhi B, Lawrence R, et al. Gene polymorphism in Netherton and common atopic disease. Nat Genet. 2001;29(2):175-8.

40. Kobayashi K, Hernandez LD, Galan JE, Janeway CA, Jr., Medzhitov R, Flavell RA. IRAK-M is a negative regulator of Toll-like receptor signaling. Cell. 2002;110(2):191-202.

41. Balaci L, Spada MC, Olla N, Sole G, Loddo L, Anedda F, et al. IRAK-M is involved in the pathogenesis of early-onset persistent asthma. Am J Hum Genet. 2007;80(6):1103-14.

42. Nakanishi K, Yoshimoto T, Tsutsui H, Okamura H. Interleukin-18 is a unique cytokine that stimulates both Th1 and Th2 responses depending on its cytokine milieu. Cytokine Growth Factor Rev. 2001;12(1):53-72.

43. Schmitz J, Owyang A, Oldham E, Song Y, Murphy E, McClanahan TK, et al. IL-33, an interleukin-1-like cytokine that signals via the IL-1 receptor-related protein ST2 and induces T helper type 2-associated cytokines. Immunity. 2005;23(5):479-90.

44. Gudbjartsson DF, Bjornsdottir US, Halapi E, Helgadottir A, Sulem P, Jonsdottir GM, et al. Sequence variants affecting eosinophil numbers associate with asthma and myocardial infarction. Nat Genet. 2009;41(3):342-7.

45. Moffatt MF, Gut IG, Demenais F, Strachan DP, Bouzigon E, Heath S, et al. A largescale, consortium-based genomewide association study of asthma. N Engl J Med. 2010;363(13):1211-21.

46. Torgerson DG, Ampleford EJ, Chiu GY, Gauderman WJ, Gignoux CR, Graves PE, et al. Meta-analysis of genome-wide association studies of asthma in ethnically diverse North American populations. Nat Genet. 2011;43(9):887-92.

47. Ramasamy A, Kuokkanen M, Vedantam S, Gajdos ZK, Couto Alves A, Lyon HN, et al. Genome-wide association studies of asthma in population-based cohorts confirm known and suggested loci and identify an additional association near HLA. PLoS One. 2012;7(9):e44008.

48. Demenais F, Margaritte-Jeannin P, Barnes KC, Cookson WOC, Altmuller J, Ang W, et al. Multiancestry association study identifies new asthma risk loci that colocalize with immune-cell enhancer marks. Nat Genet. 2018;50(1):42-53.

49. Chen CC, Kobayashi T, Iijima K, Hsu FC, Kita H. IL-33 dysregulates regulatory T cells and impairs established immunologic tolerance in the lungs. J Allergy Clin Immunol. 2017;140(5):1351-63 e7.

50. Hackett TL, Warner SM, Stefanowicz D, Shaheen F, Pechkovsky DV, Murray LA, et al. Induction of epithelial-mesenchymal transition in primary airway epithelial cells from patients with asthma by transforming growth factor-beta1. Am J Respir Crit Care Med. 2009;180(2):122-33.

51. Miller M, Tam AB, Cho JY, Doherty TA, Pham A, Khorram N, et al. ORMDL3 is an inducible lung epithelial gene regulating metalloproteases, chemokines, OAS, and ATF6. Proc Natl Acad Sci U S A. 2012;109(41):16648-53.

52. Breslow DK, Collins SR, Bodenmiller B, Aebersold R, Simons K, Shevchenko A, et al. Orm family proteins mediate sphingolipid homeostasis. Nature. 2010;463(7284):1048-53.

53. Ito T, Wang YH, Duramad O, Hori T, Delespesse GJ, Watanabe N, et al. TSLP-activated dendritic cells induce an inflammatory T helper type 2 cell response through OX40 ligand. J Exp Med. 2005;202(9):1213-23.

54. Fu J, Wolfs MG, Deelen P, Westra HJ, Fehrmann RS, Te Meerman GJ, et al. Unraveling the regulatory mechanisms underlying tissue-dependent genetic variation of gene expression. PLoS Genet. 2012;8(1):e1002431.

55. Tripathi P, Awasthi S, Husain N, Prasad R, Mishra V. Increased expression of ADAM33 protein in asthmatic patients as compared to non-asthmatic controls. Indian J Med Res. 2013;137(3):507-14.

56. Datta A, Alexander R, Sulikowski MG, Nicholson AG, Maher TM, Scotton CJ, et al. Evidence for a functional thymic stromal lymphopoietin signaling axis in fibrotic lung disease. J Immunol. 2013;191(9):4867-79.

57. Li X, Hawkins GA, Moore WC, Hastie AT, Ampleford EJ, Milosevic J, et al. Expression of asthma susceptibility genes in bronchial epithelial cells and bronchial alveolar lavage in the Severe Asthma Research Program (SARP) cohort. J Asthma. 2016;53(8):775-82.

58. Li X, Hastie AT, Hawkins GA, Moore WC, Ampleford EJ, Milosevic J, et al. eQTL of bronchial epithelial cells and bronchial alveolar lavage deciphers GWAS-identified asthma genes. Allergy. 2015;70(10):1309-18.

59. Semlali A, Jacques E, Koussih L, Gounni AS, Chakir J. Thymic stromal lymphopoietininduced human asthmatic airway epithelial cell proliferation through an IL-13-dependent pathway. J Allergy Clin Immunol. 2010;125(4):844-50.

60. Ying S, O'Connor B, Ratoff J, Meng Q, Mallett K, Cousins D, et al. Thymic stromal lymphopoietin expression is increased in asthmatic airways and correlates with expression of Th2-attracting chemokines and disease severity. J Immunol. 2005;174(12):8183-90.

61. Prefontaine D, Nadigel J, Chouiali F, Audusseau S, Semlali A, Chakir J, et al. Increased IL-33 expression by epithelial cells in bronchial asthma. J Allergy Clin Immunol. 2010;125(3):752-4.

62. Marini M, Vittori E, Hollemborg J, Mattoli S. Expression of the potent inflammatory cytokines, granulocyte-macrophage-colony-stimulating factor and interleukin-6 and interleukin-8, in bronchial epithelial cells of patients with asthma. J Allergy Clin Immunol. 1992;89(5):1001-9.

63. Kicic A, Sutanto EN, Stevens PT, Knight DA, Stick SM. Intrinsic biochemical and functional differences in bronchial epithelial cells of children with asthma. Am J Respir Crit Care Med. 2006;174(10):1110-8.

64. Hackett TL, Singhera GK, Shaheen F, Hayden P, Jackson GR, Hegele RG, et al. Intrinsic phenotypic differences of asthmatic epithelium and its inflammatory responses to respiratory syncytial virus and air pollution. Am J Respir Cell Mol Biol. 2011;45(5):1090-100.

65. Freishtat RJ, Watson AM, Benton AS, Iqbal SF, Pillai DK, Rose MC, et al. Asthmatic airway epithelium is intrinsically inflammatory and mitotically dyssynchronous. Am J Respir Cell Mol Biol. 2011;44(6):863-9.

66. Pichavant M, Charbonnier AS, Taront S, Brichet A, Wallaert B, Pestel J, et al. Asthmatic bronchial epithelium activated by the proteolytic allergen Der p 1 increases selective dendritic cell recruitment. J Allergy Clin Immunol. 2005;115(4):771-8.

67. Lordan JL, Bucchieri F, Richter A, Konstantinidis A, Holloway JW, Thornber M, et al. Cooperative effects of Th2 cytokines and allergen on normal and asthmatic bronchial epithelial cells. J Immunol. 2002;169(1):407-14.

Parker J, Sarlang S, Thavagnanam S, Williamson G, O'Donoghue D, Villenave R, et al. A
 3-D well-differentiated model of pediatric bronchial epithelium demonstrates unstimulated

morphological differences between asthmatic and nonasthmatic cells. Pediatr Res. 2010;67(1):17-22.

69. Ordonez CL, Khashayar R, Wong HH, Ferrando R, Wu R, Hyde DM, et al. Mild and moderate asthma is associated with airway goblet cell hyperplasia and abnormalities in mucin gene expression. Am J Respir Crit Care Med. 2001;163(2):517-23.

70. Lachowicz-Scroggins ME, Yuan S, Kerr SC, Dunican EM, Yu M, Carrington SD, et al. Abnormalities in MUC5AC and MUC5B Protein in Airway Mucus in Asthma. Am J Respir Crit Care Med. 2016;194(10):1296-9.

71. Evans CM, Raclawska DS, Ttofali F, Liptzin DR, Fletcher AA, Harper DN, et al. The polymeric mucin Muc5ac is required for allergic airway hyperreactivity. Nat Commun. 2015;6:6281.

72. Stevens PT, Kicic A, Sutanto EN, Knight DA, Stick SM. Dysregulated repair in asthmatic paediatric airway epithelial cells: the role of plasminogen activator inhibitor-1. Clin Exp Allergy. 2008;38(12):1901-10.

73. Gras D, Bourdin A, Vachier I, de Senneville L, Bonnans C, Chanez P. An ex vivo model of severe asthma using reconstituted human bronchial epithelium. J Allergy Clin Immunol. 2012;129(5):1259-66 e1.

74. Stefanowicz D, Ullah J, Lee K, Shaheen F, Olumese E, Fishbane N, et al. Epigenetic modifying enzyme expression in asthmatic airway epithelial cells and fibroblasts. BMC Pulm Med. 2017;17(1):24.

75. Stefanowicz D, Hackett TL, Garmaroudi FS, Gunther OP, Neumann S, Sutanto EN, et al. DNA methylation profiles of airway epithelial cells and PBMCs from healthy, atopic and asthmatic children. PLoS One. 2012;7(9):e44213.

 Mezey RJ, Cohn MA, Fernandez RJ, Januszkiewicz AJ, Wanner A. Mucociliary transport in allergic patients with antigen-induced bronchospasm. Am Rev Respir Dis. 1978;118(4):677-84.

77. Bateman JR, Pavia D, Sheahan NF, Agnew JE, Clarke SW. Impaired tracheobronchial clearance in patients with mild stable asthma. Thorax. 1983;38(6):463-7.

78. Raundhal M, Morse C, Khare A, Oriss TB, Milosevic J, Trudeau J, et al. High IFNgamma and low SLPI mark severe asthma in mice and humans. J Clin Invest. 2015;125(8):3037-50.

79. Kale SL, Agrawal K, Gaur SN, Arora N. Cockroach protease allergen induces allergic airway inflammation via epithelial cell activation. Sci Rep. 2017;7:42341.

80. Vinhas R, Cortes L, Cardoso I, Mendes VM, Manadas B, Todo-Bom A, et al. Pollen proteases compromise the airway epithelial barrier through degradation of transmembrane adhesion proteins and lung bioactive peptides. Allergy. 2011;66(8):1088-98.

81. Wan H, Winton HL, Soeller C, Tovey ER, Gruenert DC, Thompson PJ, et al. Der p 1 facilitates transepithelial allergen delivery by disruption of tight junctions. J Clin Invest. 1999;104(1):123-33.

82. Tai HY, Tam MF, Chou H, Peng HJ, Su SN, Perng DW, et al. Pen ch 13 allergen induces secretion of mediators and degradation of occludin protein of human lung epithelial cells. Allergy. 2006;61(3):382-8.

Xiao C, Puddicombe SM, Field S, Haywood J, Broughton-Head V, Puxeddu I, et al.
Defective epithelial barrier function in asthma. J Allergy Clin Immunol. 2011;128(3):549-56 e112.

84. de Boer WI, Sharma HS, Baelemans SM, Hoogsteden HC, Lambrecht BN, Braunstahl GJ. Altered expression of epithelial junctional proteins in atopic asthma: possible role in inflammation. Can J Physiol Pharmacol. 2008;86(3):105-12.

85. Sweerus K, Lachowicz-Scroggins M, Gordon E, LaFemina M, Huang X, Parikh M, et al. Claudin-18 deficiency is associated with airway epithelial barrier dysfunction and asthma. J Allergy Clin Immunol. 2017;139(1):72-81 e1.

86. Hardyman MA, Wilkinson E, Martin E, Jayasekera NP, Blume C, Swindle EJ, et al. TNF-alpha-mediated bronchial barrier disruption and regulation by src-family kinase activation. J Allergy Clin Immunol. 2013;132(3):665-75 e8.

87. Antony AB, Tepper RS, Mohammed KA. Cockroach extract antigen increases bronchial airway epithelial permeability. J Allergy Clin Immunol. 2002;110(4):589-95.

88. Dong H, Hu Y, Liu L, Zou M, Huang C, Luo L, et al. Distinct roles of short and long thymic stromal lymphopoietin isoforms in house dust mite-induced asthmatic airway epithelial barrier disruption. Sci Rep. 2016;6:39559.

89. Ahdieh M, Vandenbos T, Youakim A. Lung epithelial barrier function and wound healing are decreased by IL-4 and IL-13 and enhanced by IFN-gamma. Am J Physiol Cell Physiol. 2001;281(6):C2029-38.

90. Hiemstra PS, McCray PB, Jr., Bals R. The innate immune function of airway epithelial cells in inflammatory lung disease. Eur Respir J. 2015;45(4):1150-62.

91. Asokananthan N, Graham PT, Fink J, Knight DA, Bakker AJ, McWilliam AS, et al. Activation of protease-activated receptor (PAR)-1, PAR-2, and PAR-4 stimulates IL-6, IL-8, and prostaglandin E2 release from human respiratory epithelial cells. J Immunol. 2002;168(7):3577-85.

92. Chu DK, Llop-Guevara A, Walker TD, Flader K, Goncharova S, Boudreau JE, et al. IL-33, but not thymic stromal lymphopoietin or IL-25, is central to mite and peanut allergic sensitization. J Allergy Clin Immunol. 2013;131(1):187-200 e1-8.

93. Hoshino A, Tanaka Y, Akiba H, Asakura Y, Mita Y, Sakurai T, et al. Critical role for OX40 ligand in the development of pathogenic Th2 cells in a murine model of asthma. Eur J Immunol. 2003;33(4):861-9.

94. Zhou Q, Ho AW, Schlitzer A, Tang Y, Wong KH, Wong FH, et al. GM-CSF-licensed CD11b+ lung dendritic cells orchestrate Th2 immunity to Blomia tropicalis. J Immunol. 2014;193(2):496-509.

95. Sheih A, Parks WC, Ziegler SF. GM-CSF produced by the airway epithelium is required for sensitization to cockroach allergen. Mucosal Immunol. 2017;10(3):705-15.

96. Moro K, Yamada T, Tanabe M, Takeuchi T, Ikawa T, Kawamoto H, et al. Innate production of T(H)2 cytokines by adipose tissue-associated c-Kit(+)Sca-1(+) lymphoid cells. Nature. 2010;463(7280):540-4.

97. Camelo A, Rosignoli G, Ohne Y, Stewart RA, Overed-Sayer C, Sleeman MA, et al. IL-33, IL-25, and TSLP induce a distinct phenotypic and activation profile in human type 2 innate lymphoid cells. Blood Adv. 2017;1(10):577-89. 98. Poulsen LK, Hummelshoj L. Triggers of IgE class switching and allergy development. Ann Med. 2007;39(6):440-56.

99. Kang JW, Lee YH, Kang MJ, Lee HJ, Oh R, Min HJ, et al. Synergistic mucus secretion by histamine and IL-4 through TMEM16A in airway epithelium. Am J Physiol Lung Cell Mol Physiol. 2017;313(3):L466-L76.

100. van der Velden VH, Naber BA, Wierenga-Wolf AF, Debets R, Savelkoul HF, Overbeek SE, et al. Interleukin 4 receptors on human bronchial epithelial cells. An in vivo and in vitro analysis of expression and function. Cytokine. 1998;10(10):803-13.

101. White SR, Martin LD, Stern R, Laxman B, Marroquin BA. Expression of IL-4/IL-13 receptors in differentiating human airway epithelial cells. Am J Physiol Lung Cell Mol Physiol. 2010;299(5):L681-93.

102. Taube C, Hiemstra PS. IL-13 and the Airway Epithelium. It Is All in the Genes. Am J Respir Crit Care Med. 2016;193(4):347-8.

103. Hatzivlassiou M, Grainge C, Kehagia V, Lau L, Howarth PH. The allergen specificity of the late asthmatic reaction. Allergy. 2010;65(3):355-8.

104. Dyer KD, Percopo CM, Rosenberg HF. IL-33 promotes eosinophilia in vivo and antagonizes IL-5-dependent eosinophil hematopoiesis ex vivo. Immunol Lett. 2013;150(1-2):41-7.

105. Hurst SD, Muchamuel T, Gorman DM, Gilbert JM, Clifford T, Kwan S, et al. New IL-17 family members promote Th1 or Th2 responses in the lung: in vivo function of the novel cytokine IL-25. J Immunol. 2002;169(1):443-53.

106. Greenfeder S, Umland SP, Cuss FM, Chapman RW, Egan RW. Th2 cytokines and asthma. The role of interleukin-5 in allergic eosinophilic disease. Respir Res. 2001;2(2):71-9.

107. McMillan SJ, Xanthou G, Lloyd CM. Manipulation of allergen-induced airway remodeling by treatment with anti-TGF-beta antibody: effect on the Smad signaling pathway. J Immunol. 2005;174(9):5774-80.

108. Perng DW, Wu YC, Chang KT, Wu MT, Chiou YC, Su KC, et al. Leukotriene C4 induces TGF-beta1 production in airway epithelium via p38 kinase pathway. Am J Respir Cell Mol Biol. 2006;34(1):101-7.

109. Trian T, Allard B, Dupin I, Carvalho G, Ousova O, Maurat E, et al. House dust mites induce proliferation of severe asthmatic smooth muscle cells via an epithelium-dependent pathway. Am J Respir Crit Care Med. 2015;191(5):538-46.

110. Ramos C, Becerril C, Montano M, Garcia-De-Alba C, Ramirez R, Checa M, et al. FGF-1 reverts epithelial-mesenchymal transition induced by TGF-{beta}1 through MAPK/ERK kinase pathway. Am J Physiol Lung Cell Mol Physiol. 2010;299(2):L222-31.

111. Batra V, Musani AI, Hastie AT, Khurana S, Carpenter KA, Zangrilli JG, et al. Bronchoalveolar lavage fluid concentrations of transforming growth factor (TGF)-beta1, TGF-beta2, interleukin (IL)-4 and IL-13 after segmental allergen challenge and their effects on alpha-smooth muscle actin and collagen III synthesis by primary human lung fibroblasts. Clin Exp Allergy. 2004;34(3):437-44.

112. Ohbayashi H, Shimokata K. Matrix metalloproteinase-9 and airway remodeling in asthma. Curr Drug Targets Inflamm Allergy. 2005;4(2):177-81.

113. Lee CG, Homer RJ, Zhu Z, Lanone S, Wang X, Koteliansky V, et al. Interleukin-13 induces tissue fibrosis by selectively stimulating and activating transforming growth factor beta(1). J Exp Med. 2001;194(6):809-21.

114. Oprian DD. The ligand-binding domain of rhodopsin and other G protein-linked receptors. J Bioenerg Biomembr. 1992;24(2):211-7.

115. Seibert C, Harteneck C, Ernst OP, Schultz G, Hofmann KP. Activation of the rod Gprotein Gt by the thrombin receptor (PAR1) expressed in Sf9 cells. Eur J Biochem. 1999;266(3):911-6.

116. Vu TK, Hung DT, Wheaton VI, Coughlin SR. Molecular cloning of a functional thrombin receptor reveals a novel proteolytic mechanism of receptor activation. Cell. 1991;64(6):1057-68.

117. Dery O, Corvera CU, Steinhoff M, Bunnett NW. Proteinase-activated receptors: novel mechanisms of signaling by serine proteases. Am J Physiol. 1998;274(6 Pt 1):C1429-52.

118. Soh UJ, Dores MR, Chen B, Trejo J. Signal transduction by protease-activated receptors.Br J Pharmacol. 2010;160(2):191-203.

119. Al-Ani B, Saifeddine M, Kawabata A, Hollenberg MD. Proteinase activated receptor 2:
Role of extracellular loop 2 for ligand-mediated activation. Br J Pharmacol. 1999;128(5):110513.

120. Nystedt S, Emilsson K, Wahlestedt C, Sundelin J. Molecular cloning of a potential proteinase activated receptor. Proc Natl Acad Sci U S A. 1994;91(20):9208-12.

121. Bohm SK, Khitin LM, Grady EF, Aponte G, Payan DG, Bunnett NW. Mechanisms of desensitization and resensitization of proteinase-activated receptor-2. J Biol Chem. 1996;271(36):22003-16.

122. Corvera CU, Dery O, McConalogue K, Bohm SK, Khitin LM, Caughey GH, et al. Mast cell tryptase regulates rat colonic myocytes through proteinase-activated receptor 2. J Clin Invest. 1997;100(6):1383-93.

123. Camerer E, Huang W, Coughlin SR. Tissue factor- and factor X-dependent activation of protease-activated receptor 2 by factor VIIa. Proc Natl Acad Sci U S A. 2000;97(10):5255-60.

124. Smith R, Jenkins A, Lourbakos A, Thompson P, Ramakrishnan V, Tomlinson J, et al. Evidence for the activation of PAR-2 by the sperm protease, acrosin: expression of the receptor on oocytes. FEBS Lett. 2000;484(3):285-90.

125. Cottrell GS, Amadesi S, Grady EF, Bunnett NW. Trypsin IV, a novel agonist of proteaseactivated receptors 2 and 4. J Biol Chem. 2004;279(14):13532-9.

126. Dulon S, Leduc D, Cottrell GS, D'Alayer J, Hansen KK, Bunnett NW, et al. Pseudomonas aeruginosa elastase disables proteinase-activated receptor 2 in respiratory epithelial cells. Am J Respir Cell Mol Biol. 2005;32(5):411-9.

127. Oikonomopoulou K, Hansen KK, Saifeddine M, Tea I, Blaber M, Blaber SI, et al. Proteinase-activated receptors, targets for kallikrein signaling. J Biol Chem. 2006;281(43):32095-112.

128. Ramsay AJ, Dong Y, Hunt ML, Linn M, Samaratunga H, Clements JA, et al. Kallikreinrelated peptidase 4 (KLK4) initiates intracellular signaling via protease-activated receptors (PARs). KLK4 and PAR-2 are co-expressed during prostate cancer progression. J Biol Chem. 2008;283(18):12293-304.

129. Al-Ani B, Saifeddine M, Kawabata A, Renaux B, Mokashi S, Hollenberg MD. Proteinase-activated receptor 2 (PAR(2)): development of a ligand-binding assay correlating

with activation of PAR(2) by PAR(1)- and PAR(2)-derived peptide ligands. J Pharmacol Exp Ther. 1999;290(2):753-60.

130. Scott G, Leopardi S, Parker L, Babiarz L, Seiberg M, Han R. The proteinase-activated receptor-2 mediates phagocytosis in a Rho-dependent manner in human keratinocytes. J Invest Dermatol. 2003;121(3):529-41.

131. DeFea KA, Zalevsky J, Thoma MS, Dery O, Mullins RD, Bunnett NW. beta-arrestindependent endocytosis of proteinase-activated receptor 2 is required for intracellular targeting of activated ERK1/2. J Cell Biol. 2000;148(6):1267-81.

132. Kanke T, Macfarlane SR, Seatter MJ, Davenport E, Paul A, McKenzie RC, et al. Proteinase-activated receptor-2-mediated activation of stress-activated protein kinases and inhibitory kappa B kinases in NCTC 2544 keratinocytes. J Biol Chem. 2001;276(34):31657-66.

133. Stalheim L, Ding Y, Gullapalli A, Paing MM, Wolfe BL, Morris DR, et al. Multiple independent functions of arrestins in the regulation of protease-activated receptor-2 signaling and trafficking. Mol Pharmacol. 2005;67(1):78-87.

134. Hyun E, Andrade-Gordon P, Steinhoff M, Vergnolle N. Protease-activated receptor-2 activation: a major actor in intestinal inflammation. Gut. 2008;57(9):1222-9.

135. Lohman RJ, Cotterell AJ, Suen J, Liu L, Do AT, Vesey DA, et al. Antagonism of protease-activated receptor 2 protects against experimental colitis. J Pharmacol Exp Ther. 2012;340(2):256-65.

136. Seeliger S, Derian CK, Vergnolle N, Bunnett NW, Nawroth R, Schmelz M, et al. Proinflammatory role of proteinase-activated receptor-2 in humans and mice during cutaneous inflammation in vivo. FASEB J. 2003;17(13):1871-85.

137. Steinhoff M, Corvera CU, Thoma MS, Kong W, McAlpine BE, Caughey GH, et al. Proteinase-activated receptor-2 in human skin: tissue distribution and activation of keratinocytes by mast cell tryptase. Exp Dermatol. 1999;8(4):282-94.

138. Knight V, Tchongue J, Lourensz D, Tipping P, Sievert W. Protease-activated receptor 2 promotes experimental liver fibrosis in mice and activates human hepatic stellate cells. Hepatology. 2012;55(3):879-87.

139. Xue M, Chan YK, Shen K, Dervish S, March L, Sambrook PN, et al. Protease-activated receptor 2, rather than protease-activated receptor 1, contributes to the aggressive properties of synovial fibroblasts in rheumatoid arthritis. Arthritis Rheum. 2012;64(1):88-98.

140. Bunnett NW. Protease-activated receptors: how proteases signal to cells to cause inflammation and pain. Semin Thromb Hemost. 2006;32 Suppl 1:39-48.

141. Schmidlin F, Amadesi S, Dabbagh K, Lewis DE, Knott P, Bunnett NW, et al. Proteaseactivated receptor 2 mediates eosinophil infiltration and hyperreactivity in allergic inflammation of the airway. J Immunol. 2002;169(9):5315-21.

142. Asaduzzaman M, Nadeem A, Arizmendi N, Davidson C, Nichols HL, Abel M, et al. Functional inhibition of PAR2 alleviates allergen-induced airway hyperresponsiveness and inflammation. Clin Exp Allergy. 2015;45(12):1844-55.

143. Wygrecka M, Kwapiszewska G, Jablonska E, von Gerlach S, Henneke I, Zakrzewicz D, et al. Role of protease-activated receptor-2 in idiopathic pulmonary fibrosis. Am J Respir Crit Care Med. 2011;183(12):1703-14.

144. Kwapiszewska G, Markart P, Dahal BK, Kojonazarov B, Marsh LM, Schermuly RT, et al. PAR-2 inhibition reverses experimental pulmonary hypertension. Circ Res. 2012;110(9):1179-91.

145. Sun G, Stacey MA, Schmidt M, Mori L, Mattoli S. Interaction of mite allergens Der p3 and Der p9 with protease-activated receptor-2 expressed by lung epithelial cells. J Immunol. 2001;167(2):1014-21.

146. Hong JH, Lee SI, Kim KE, Yong TS, Seo JT, Sohn MH, et al. German cockroach extract activates protease-activated receptor 2 in human airway epithelial cells. J Allergy Clin Immunol. 2004;113(2):315-9.

147. Matsuwaki Y, Wada K, White T, Moriyama H, Kita H. Alternaria fungus induces the production of GM-CSF, interleukin-6 and interleukin-8 and calcium signaling in human airway epithelium through protease-activated receptor 2. Int Arch Allergy Immunol. 2012;158 Suppl 1:19-29.

148. Chiu LL, Perng DW, Yu CH, Su SN, Chow LP. Mold allergen, pen C 13, induces IL-8 expression in human airway epithelial cells by activating protease-activated receptor 1 and 2. J Immunol. 2007;178(8):5237-44.

149. Cocks TM, Fong B, Chow JM, Anderson GP, Frauman AG, Goldie RG, et al. A protective role for protease-activated receptors in the airways. Nature. 1999;398(6723):156-60.

150. Lindner JR, Kahn ML, Coughlin SR, Sambrano GR, Schauble E, Bernstein D, et al. Delayed onset of inflammation in protease-activated receptor-2-deficient mice. J Immunol. 2000;165(11):6504-10.

151. Takizawa T, Tamiya M, Hara T, Matsumoto J, Saito N, Kanke T, et al. Abrogation of bronchial eosinophilic inflammation and attenuated eotaxin content in protease-activated receptor 2-deficient mice. J Pharmacol Sci. 2005;98(1):99-102.

152. Page K, Ledford JR, Zhou P, Dienger K, Wills-Karp M. Mucosal sensitization to German cockroach involves protease-activated receptor-2. Respir Res. 2010;11:62.

153. Arizmendi NG, Abel M, Mihara K, Davidson C, Polley D, Nadeem A, et al. Mucosal allergic sensitization to cockroach allergens is dependent on proteinase activity and proteinase-activated receptor-2 activation. J Immunol. 2011;186(5):3164-72.

154. Davidson CE, Asaduzzaman M, Arizmendi NG, Polley D, Wu Y, Gordon JR, et al. Proteinase-activated receptor-2 activation participates in allergic sensitization to house dust mite allergens in a murine model. Clin Exp Allergy. 2013;43(11):1274-85.

155. Asaduzzaman M, Davidson C, Nahirney D, Fiteih Y, Puttagunta L, Vliagoftis H. Proteinase-activated receptor-2 blockade inhibits changes seen in a chronic murine asthma model. Allergy. 2018;73(2):416-20.

156. Schmidlin F, Amadesi S, Vidil R, Trevisani M, Martinet N, Caughey G, et al. Expression and function of proteinase-activated receptor 2 in human bronchial smooth muscle. Am J Respir Crit Care Med. 2001;164(7):1276-81.

157. Miotto D, Hollenberg MD, Bunnett NW, Papi A, Braccioni F, Boschetto P, et al. Expression of protease activated receptor-2 (PAR-2) in central airways of smokers and non-smokers. Thorax. 2002;57(2):146-51.

158. Matsushima R, Takahashi A, Nakaya Y, Maezawa H, Miki M, Nakamura Y, et al. Human airway trypsin-like protease stimulates human bronchial fibroblast proliferation in a protease-activated receptor-2-dependent pathway. Am J Physiol Lung Cell Mol Physiol. 2006;290(2):L385-95. 159. Bock A, Tucker N, Kelher MR, Khan SY, Gonzalez E, Wohlauer M, et al. alpha-Enolase Causes Proinflammatory Activation of Pulmonary Microvascular Endothelial Cells and Primes Neutrophils Through Plasmin Activation of Protease-Activated Receptor 2. Shock. 2015;44(2):137-42.

160. Jin E, Fujiwara M, Pan X, Ghazizadeh M, Arai S, Ohaki Y, et al. Protease-activated receptor (PAR)-1 and PAR-2 participate in the cell growth of alveolar capillary endothelium in primary lung adenocarcinomas. Cancer. 2003;97(3):703-13.

161. Carvalho RF, Nilsson G, Harvima IT. Increased mast cell expression of PAR-2 in skin inflammatory diseases and release of IL-8 upon PAR-2 activation. Exp Dermatol. 2010;19(2):117-22.

162. Jiang B, Grage-Griebenow E, Csernok E, Butherus K, Ehlers S, Gross WL, et al. The role of proteinase 3 (PR3) and the protease-activated receptor-2 (PAR-2) pathway in dendritic cell (DC) maturation of human-DC-like monocytes and murine DC. Clin Exp Rheumatol. 2010;28(1 Suppl 57):56-61.

163. Colognato R, Slupsky JR, Jendrach M, Burysek L, Syrovets T, Simmet T. Differential expression and regulation of protease-activated receptors in human peripheral monocytes and monocyte-derived antigen-presenting cells. Blood. 2003;102(7):2645-52.

164. Bolton SJ, McNulty CA, Thomas RJ, Hewitt CR, Wardlaw AJ. Expression of and functional responses to protease-activated receptors on human eosinophils. J Leukoc Biol. 2003;74(1):60-8.

165. Crilly A, Burns E, Nickdel MB, Lockhart JC, Perry ME, Ferrell PW, et al. PAR(2) expression in peripheral blood monocytes of patients with rheumatoid arthritis. Ann Rheum Dis. 2012;71(6):1049-54.

166. Zheng J, Liu W, Fan Y, Ye X, Xia W, Wang H, et al. Suppression of connexin 26 is related to protease-activated receptor 2-mediated pathway in patients with allergic rhinitis. Am J Rhinol Allergy. 2012;26(1):e5-9.

167. Heijink IH, van Oosterhout A, Kapus A. Epidermal growth factor receptor signalling contributes to house dust mite-induced epithelial barrier dysfunction. Eur Respir J. 2010;36(5):1016-26.

168. Lewkowich IP, Day SB, Ledford JR, Zhou P, Dienger K, Wills-Karp M, et al. Proteaseactivated receptor 2 activation of myeloid dendritic cells regulates allergic airway inflammation. Respir Res. 2011;12:122.

169. Wang H, Yi T, Zheng Y, He S. Induction of monocyte chemoattractant protein-1 release from A549 cells by agonists of protease-activated receptor-1 and -2. Eur J Cell Biol. 2007;86(4):233-42.

170. Vliagoftis H, Befus AD, Hollenberg MD, Moqbel R. Airway epithelial cells release eosinophil survival-promoting factors (GM-CSF) after stimulation of proteinase-activated receptor 2. J Allergy Clin Immunol. 2001;107(4):679-85.

171. Kale SL, Arora N. Per a 10 activates human derived epithelial cell line in a protease dependent manner via PAR-2. Immunobiology. 2015;220(4):525-32.

172. Kouzaki H, O'Grady SM, Lawrence CB, Kita H. Proteases induce production of thymic stromal lymphopoietin by airway epithelial cells through protease-activated receptor-2. J Immunol. 2009;183(2):1427-34.

173. Kouzaki H, Tojima I, Kita H, Shimizu T. Transcription of interleukin-25 and extracellular release of the protein is regulated by allergen proteases in airway epithelial cells. Am J Respir Cell Mol Biol. 2013;49(5):741-50.

174. Kauffman HF, Tamm M, Timmerman JA, Borger P. House dust mite major allergens Der p 1 and Der p 5 activate human airway-derived epithelial cells by protease-dependent and protease-independent mechanisms. Clin Mol Allergy. 2006;4:5.

175. Lee MF, Wang NM, Liu SW, Lin SJ, Chen YH. Induction of interleukin 8 by American cockroach allergens from human airway epithelial cells via extracellular signal regulatory kinase and jun N-terminal kinase but not p38 mitogen-activated protein kinase. Ann Allergy Asthma Immunol. 2010;105(3):234-40.

176. Asokananthan N, Graham PT, Stewart DJ, Bakker AJ, Eidne KA, Thompson PJ, et al. House dust mite allergens induce proinflammatory cytokines from respiratory epithelial cells: the cysteine protease allergen, Der p 1, activates protease-activated receptor (PAR)-2 and inactivates PAR-1. J Immunol. 2002;169(8):4572-8.

177. Sumi Y, Hamid Q. Airway remodeling in asthma. Allergol Int. 2007;56(4):341-8.

178. Vliagoftis H, Schwingshackl A, Milne CD, Duszyk M, Hollenberg MD, Wallace JL, et al. Proteinase-activated receptor-2-mediated matrix metalloproteinase-9 release from airway epithelial cells. J Allergy Clin Immunol. 2000;106(3):537-45.

179. Page K, Strunk VS, Hershenson MB. Cockroach proteases increase IL-8 expression in human bronchial epithelial cells via activation of protease-activated receptor (PAR)-2 and extracellular-signal-regulated kinase. J Allergy Clin Immunol. 2003;112(6):1112-8.

180. Page K, Hughes VS, Bennett GW, Wong HR. German cockroach proteases regulate matrix metalloproteinase-9 in human bronchial epithelial cells. Allergy. 2006;61(8):988-95.

181. Lieber M, Smith B, Szakal A, Nelson-Rees W, Todaro G. A continuous tumor-cell line from a human lung carcinoma with properties of type II alveolar epithelial cells. Int J Cancer. 1976;17(1):62-70.

182. Reddel RR, Yang K, Rhim JS, Brash D, Su RT, Lechner JF, et al. Immortalized human bronchial epitherial mesothelial cell lines. Google Patents; 1989.

183. Allard B, Bara I, Gilbert G, Carvalho G, Trian T, Ozier A, et al. Protease activated receptor-2 expression and function in asthmatic bronchial smooth muscle. PLoS One. 2014;9(2):e86945.

184. Aubier M, Thabut G, Hamidi F, Guillou N, Brard J, Dombret MC, et al. Airway smooth muscle enlargement is associated with protease-activated receptor 2/ligand overexpression in patients with difficult-to-control severe asthma. J Allergy Clin Immunol. 2016;138(3):729-39 e11.

185. Knight DA, Lim S, Scaffidi AK, Roche N, Chung KF, Stewart GA, et al. Proteaseactivated receptors in human airways: upregulation of PAR-2 in respiratory epithelium from patients with asthma. J Allergy Clin Immunol. 2001;108(5):797-803.

186. Ferrell WR, Lockhart JC, Kelso EB, Dunning L, Plevin R, Meek SE, et al. Essential role for proteinase-activated receptor-2 in arthritis. J Clin Invest. 2003;111(1):35-41.

187. Wei L, Yu X, Shi H, Zhang B, Lian M, Li J, et al. 15-PGDH/15-KETE plays a role in hypoxia-induced pulmonary vascular remodeling through ERK1/2-dependent PAR-2 pathway. Cell Signal. 2014;26(7):1476-88.

188. Lee JH, Kim KW, Gee HY, Lee J, Lee KH, Park HS, et al. A synonymous variation in protease-activated receptor-2 is associated with atopy in Korean children. J Allergy Clin Immunol. 2011;128(6):1326-34 e3.

189. Adams MN, Pagel CN, Mackie EJ, Hooper JD. Evaluation of antibodies directed against human protease-activated receptor-2. Naunyn Schmiedebergs Arch Pharmacol. 2012;385(9):861-73.

190. Ostrowska E, Sokolova E, Reiser G. PAR-2 activation and LPS synergistically enhance inflammatory signaling in airway epithelial cells by raising PAR expression level and interleukin-8 release. Am J Physiol Lung Cell Mol Physiol. 2007;293(5):L1208-18.

191. Ritchie E, Saka M, Mackenzie C, Drummond R, Wheeler-Jones C, Kanke T, et al. Cytokine upregulation of proteinase-activated-receptors 2 and 4 expression mediated by p38 MAP kinase and inhibitory kappa B kinase beta in human endothelial cells. Br J Pharmacol. 2007;150(8):1044-54.

192. Zhang H, Lin L, Yang H, Zhang Z, Yang X, Zhang L, et al. Induction of IL-13 production and upregulation of gene expression of protease activated receptors in P815 cells by IL-6. Cytokine. 2010;50(2):138-45.

193. Zhang H, Yang H, Zhang L, Yang X, Zhang Z, Lin Q, et al. Induction of IL-4 release and upregulated expression of protease activated receptors by GM-CSF in P815 cells. Cytokine. 2009;48(3):196-202.

194. Gruber BL, Marchese MJ, Santiago-Schwarz F, Martin CA, Zhang J, Kew RR. Proteaseactivated receptor-2 (PAR-2) expression in human fibroblasts is regulated by growth factors and extracellular matrix. J Invest Dermatol. 2004;123(5):832-9.

195. Freund-Michel V, Frossard N. Inflammatory conditions increase expression of proteaseactivated receptor-2 by human airway smooth muscle cells in culture. Fundam Clin Pharmacol. 2006;20(4):351-7.

196. Xiang Y, Masuko-Hongo K, Sekine T, Nakamura H, Yudoh K, Nishioka K, et al. Expression of proteinase-activated receptors (PAR)-2 in articular chondrocytes is modulated by IL-1beta, TNF-alpha and TGF-beta. Osteoarthritis Cartilage. 2006;14(11):1163-73.

197. Tsai SH, Sheu MT, Liang YC, Cheng HT, Fang SS, Chen CH. TGF-beta inhibits IL-1beta-activated PAR-2 expression through multiple pathways in human primary synovial cells. J Biomed Sci. 2009;16:97.

198. Nystedt S, Ramakrishnan V, Sundelin J. The proteinase-activated receptor 2 is induced by inflammatory mediators in human endothelial cells. Comparison with the thrombin receptor. J Biol Chem. 1996;271(25):14910-5.

199. Zhang H, Yang H, He S. TNF increases expression of IL-4 and PARs in mast cells. Cell Physiol Biochem. 2010;26(3):327-36.

200. Zhang Z, Zhang H, Yang H, Zhang L, Chen X, Zheng X, et al. Induction of T-helper type 2 cytokine release and up-regulated expression of protease-activated receptors on mast cells by recombinant American cockroach allergen Per a 7. Clin Exp Allergy. 2008;38(7):1160-7.

201. He S, Zhang Z, Zhang H, Wei J, Yang L, Yang H, et al. Analysis of properties and proinflammatory functions of cockroach allergens Per a 1.01s. Scand J Immunol. 2011;74(3):288-95.

202. Svensson KJ, Kucharzewska P, Christianson HC, Skold S, Lofstedt T, Johansson MC, et al. Hypoxia triggers a proangiogenic pathway involving cancer cell microvesicles and PAR-2-mediated heparin-binding EGF signaling in endothelial cells. Proc Natl Acad Sci U S A. 2011;108(32):13147-52.

203. Hamilton JR, Frauman AG, Cocks TM. Increased expression of protease-activated receptor-2 (PAR2) and PAR4 in human coronary artery by inflammatory stimuli unveils endothelium-dependent relaxations to PAR2 and PAR4 agonists. Circ Res. 2001;89(1):92-8.

204. Sokolova E, Aleshin S, Reiser G. Expression of protease-activated receptor (PAR)-2, but not other PARs, is regulated by inflammatory cytokines in rat astrocytes. Neurochem Int. 2012;60(3):276-85.

205. Zhang H, Yang X, Yang H, Zhang Z, Lin Q, Zheng Y, et al. Modulation of mast cell proteinase-activated receptor expression and IL-4 release by IL-12. Immunol Cell Biol. 2007;85(7):558-66.

206. Zhou J, Perelman JM, Kolosov VP, Zhou X. Neutrophil elastase induces MUC5AC secretion via protease-activated receptor 2. Mol Cell Biochem. 2013;377(1-2):75-85.

207. Bliss M. The history of insulin. Diabetes Care. 1993;16 Suppl 3:4-7.

208. Taniguchi CM, Emanuelli B, Kahn CR. Critical nodes in signalling pathways: insights into insulin action. Nat Rev Mol Cell Biol. 2006;7(2):85-96.

209. Cohen P. The twentieth century struggle to decipher insulin signalling. Nat Rev Mol Cell Biol. 2006;7(11):867-73.

210. Ottlecz A, Koltai M, Dekov E. Effect of insulin and alloxan diabetes on carrageenin inflammation in rats. Int Arch Allergy Appl Immunol. 1976;50(5):548-54.

211. Collier B, Dossett LA, May AK, Diaz JJ. Glucose control and the inflammatory response. Nutr Clin Pract. 2008;23(1):3-15.

212. Hyun E, Ramachandran R, Hollenberg MD, Vergnolle N. Mechanisms behind the antiinflammatory actions of insulin. Crit Rev Immunol. 2011;31(4):307-40.

213. Werner H, Weinstein D, Bentov I. Similarities and differences between insulin and IGF-I: structures, receptors, and signalling pathways. Arch Physiol Biochem. 2008;114(1):17-22.

214. Boucher J, Tseng YH, Kahn CR. Insulin and insulin-like growth factor-1 receptors act as ligand-specific amplitude modulators of a common pathway regulating gene transcription. J Biol Chem. 2010;285(22):17235-45.

215. Kaburagi Y, Yamamoto-Honda R, Tobe K, Ueki K, Yachi M, Akanuma Y, et al. The role of the NPXY motif in the insulin receptor in tyrosine phosphorylation of insulin receptor substrate-1 and Shc. Endocrinology. 1995;136(8):3437-43.

216. Isakoff SJ, Yu YP, Su YC, Blaikie P, Yajnik V, Rose E, et al. Interaction between the phosphotyrosine binding domain of Shc and the insulin receptor is required for Shc phosphorylation by insulin in vivo. J Biol Chem. 1996;271(8):3959-62.

217. Shepherd PR, Withers DJ, Siddle K. Phosphoinositide 3-kinase: the key switch mechanism in insulin signalling. Biochem J. 1998;333 (Pt 3):471-90.

218. Stephens L, Smrcka A, Cooke FT, Jackson TR, Sternweis PC, Hawkins PT. A novel phosphoinositide 3 kinase activity in myeloid-derived cells is activated by G protein beta gamma subunits. Cell. 1994;77(1):83-93.

219. Stoyanov B, Volinia S, Hanck T, Rubio I, Loubtchenkov M, Malek D, et al. Cloning and characterization of a G protein-activated human phosphoinositide-3 kinase. Science. 1995;269(5224):690-3.

220. Giorgetti S, Ballotti R, Kowalski-Chauvel A, Tartare S, Van Obberghen E. The insulin and insulin-like growth factor-I receptor substrate IRS-1 associates with and activates phosphatidylinositol 3-kinase in vitro. J Biol Chem. 1993;268(10):7358-64.

221. Nave BT, Siddle K, Shepherd PR. Phorbol esters stimulate phosphatidylinositol 3,4,5trisphosphate production in 3T3-L1 adipocytes: implications for stimulation of glucose transport. Biochem J. 1996;318 (Pt 1):203-5.

222. Alessi DR, Deak M, Casamayor A, Caudwell FB, Morrice N, Norman DG, et al. 3-Phosphoinositide-dependent protein kinase-1 (PDK1): structural and functional homology with the Drosophila DSTPK61 kinase. Curr Biol. 1997;7(10):776-89.

223. Currie RA, Walker KS, Gray A, Deak M, Casamayor A, Downes CP, et al. Role of phosphatidylinositol 3,4,5-trisphosphate in regulating the activity and localization of 3-phosphoinositide-dependent protein kinase-1. Biochem J. 1999;337 (Pt 3):575-83.

224. Stephens L, Anderson K, Stokoe D, Erdjument-Bromage H, Painter GF, Holmes AB, et al. Protein kinase B kinases that mediate phosphatidylinositol 3,4,5-trisphosphate-dependent activation of protein kinase B. Science. 1998;279(5351):710-4.

225. Sarbassov DD, Guertin DA, Ali SM, Sabatini DM. Phosphorylation and regulation of Akt/PKB by the rictor-mTOR complex. Science. 2005;307(5712):1098-101.

226. Saltiel AR, Kahn CR. Insulin signalling and the regulation of glucose and lipid metabolism. Nature. 2001;414(6865):799-806.

227. Skolnik EY, Batzer A, Li N, Lee CH, Lowenstein E, Mohammadi M, et al. The function of GRB2 in linking the insulin receptor to Ras signaling pathways. Science. 1993;260(5116):1953-5.

228. Skolnik EY, Lee CH, Batzer A, Vicentini LM, Zhou M, Daly R, et al. The SH2/SH3 domain-containing protein GRB2 interacts with tyrosine-phosphorylated IRS1 and Shc: implications for insulin control of ras signalling. Embo J. 1993;12(5):1929-36.

229. Pronk GJ, Bos JL. The role of p21ras in receptor tyrosine kinase signalling. Biochim Biophys Acta. 1994;1198(2-3):131-47.

230. Dent P, Haser W, Haystead TA, Vincent LA, Roberts TM, Sturgill TW. Activation of mitogen-activated protein kinase kinase by v-Raf in NIH 3T3 cells and in vitro. Science. 1992;257(5075):1404-7.

231. Crews CM, Alessandrini A, Erikson RL. The primary structure of MEK, a protein kinase that phosphorylates the ERK gene product. Science. 1992;258(5081):478-80.

232. Plotnikov A, Zehorai E, Procaccia S, Seger R. The MAPK cascades: signaling components, nuclear roles and mechanisms of nuclear translocation. Biochim Biophys Acta. 2011;1813(9):1619-33.

233. Sun Y, Liu WZ, Liu T, Feng X, Yang N, Zhou HF. Signaling pathway of MAPK/ERK in cell proliferation, differentiation, migration, senescence and apoptosis. J Recept Signal Transduct Res. 2015;35(6):600-4.

234. Mebratu Y, Tesfaigzi Y. How ERK1/2 activation controls cell proliferation and cell death: Is subcellular localization the answer? Cell Cycle. 2009;8(8):1168-75.

235. Ghanim H, Korzeniewski K, Sia CL, Abuaysheh S, Lohano T, Chaudhuri A, et al. Suppressive effect of insulin infusion on chemokines and chemokine receptors. Diabetes Care. 2010;33(5):1103-8.

236. Aljada A, Saadeh R, Assian E, Ghanim H, Dandona P. Insulin inhibits the expression of intercellular adhesion molecule-1 by human aortic endothelial cells through stimulation of nitric oxide. J Clin Endocrinol Metab. 2000;85(7):2572-5.

237. Takeda K, Akira S. Toll-like receptors in innate immunity. Int Immunol. 2005;17(1):1-14.

238. Fajas L, Debril MB, Auwerx J. Peroxisome proliferator-activated receptor-gamma: from adipogenesis to carcinogenesis. J Mol Endocrinol. 2001;27(1):1-9.

239. Devaraj S, Dasu MR, Rockwood J, Winter W, Griffen SC, Jialal I. Increased toll-like receptor (TLR) 2 and TLR4 expression in monocytes from patients with type 1 diabetes: further evidence of a proinflammatory state. J Clin Endocrinol Metab. 2008;93(2):578-83.

240. Satomi N, Sakurai A, Haranaka K. Relationship of hypoglycemia to tumor necrosis factor production and antitumor activity: role of glucose, insulin, and macrophages. J Natl Cancer Inst. 1985;74(6):1255-60.

241. Jeschke MG, Klein D, Bolder U, Einspanier R. Insulin attenuates the systemic inflammatory response in endotoxemic rats. Endocrinology. 2004;145(9):4084-93.

242. Aljada A, Ghanim H, Saadeh R, Dandona P. Insulin inhibits NFkappaB and MCP-1 expression in human aortic endothelial cells. J Clin Endocrinol Metab. 2001;86(1):450-3.

243. Dandona P, Aljada A, Mohanty P, Ghanim H, Hamouda W, Assian E, et al. Insulin inhibits intranuclear nuclear factor kappaB and stimulates IkappaB in mononuclear cells in obese subjects: evidence for an anti-inflammatory effect? J Clin Endocrinol Metab. 2001;86(7):3257-65.

244. Li Q, Verma IM. NF-kappaB regulation in the immune system. Nat Rev Immunol. 2002;2(10):725-34.

245. Carter ME, Brunet A. FOXO transcription factors. Curr Biol. 2007;17(4):R113-4.

246. Tzivion G, Dobson M, Ramakrishnan G. FoxO transcription factors; Regulation by AKT and 14-3-3 proteins. Biochim Biophys Acta. 2011;1813(11):1938-45.

247. Huang H, Regan KM, Wang F, Wang D, Smith DI, van Deursen JM, et al. Skp2 inhibits FOXO1 in tumor suppression through ubiquitin-mediated degradation. Proc Natl Acad Sci U S A. 2005;102(5):1649-54.

Puigserver P, Rhee J, Donovan J, Walkey CJ, Yoon JC, Oriente F, et al. Insulin-regulated hepatic gluconeogenesis through FOXO1-PGC-1alpha interaction. Nature. 2003;423(6939):550-5.

249. Fan W, Morinaga H, Kim JJ, Bae E, Spann NJ, Heinz S, et al. FoxO1 regulates Tlr4 inflammatory pathway signalling in macrophages. EMBO J. 2010;29(24):4223-36.

250. Ito Y, Daitoku H, Fukamizu A. Foxo1 increases pro-inflammatory gene expression by inducing C/EBPbeta in TNF-alpha-treated adipocytes. Biochem Biophys Res Commun. 2009;378(2):290-5.

251. Martins R, Lithgow GJ, Link W. Long live FOXO: unraveling the role of FOXO proteins in aging and longevity. Aging Cell. 2016;15(2):196-207.

252. Ponugoti B, Xu F, Zhang C, Tian C, Pacios S, Graves DT. FOXO1 promotes wound healing through the up-regulation of TGF-beta1 and prevention of oxidative stress. J Cell Biol. 2013;203(2):327-43.

253. Zhang C, Lim J, Jeon HH, Xu F, Tian C, Miao F, et al. FOXO1 deletion in keratinocytes improves diabetic wound healing through MMP9 regulation. Sci Rep. 2017;7(1):10565.

254. Haines RJ, Beard RS, Jr., Eitner RA, Chen L, Wu MH. TNFalpha/IFNgamma Mediated Intestinal Epithelial Barrier Dysfunction Is Attenuated by MicroRNA-93 Downregulation of PTK6 in Mouse Colonic Epithelial Cells. PLoS One. 2016;11(4):e0154351. 255. Seiler F, Hellberg J, Lepper PM, Kamyschnikow A, Herr C, Bischoff M, et al. FOXO transcription factors regulate innate immune mechanisms in respiratory epithelial cells. J Immunol. 2013;190(4):1603-13.

256. Emanuelli B, Peraldi P, Filloux C, Chavey C, Freidinger K, Hilton DJ, et al. SOCS-3 inhibits insulin signaling and is up-regulated in response to tumor necrosis factor-alpha in the adipose tissue of obese mice. J Biol Chem. 2001;276(51):47944-9.

257. Rui L, Yuan M, Frantz D, Shoelson S, White MF. SOCS-1 and SOCS-3 block insulin signaling by ubiquitin-mediated degradation of IRS1 and IRS2. J Biol Chem. 2002;277(44):42394-8.

258. Rotter V, Nagaev I, Smith U. Interleukin-6 (IL-6) induces insulin resistance in 3T3-L1 adipocytes and is, like IL-8 and tumor necrosis factor-alpha, overexpressed in human fat cells from insulin-resistant subjects. J Biol Chem. 2003;278(46):45777-84.

259. Jager J, Gremeaux T, Cormont M, Le Marchand-Brustel Y, Tanti JF. Interleukin-1betainduced insulin resistance in adipocytes through down-regulation of insulin receptor substrate-1 expression. Endocrinology. 2007;148(1):241-51.

260. Stephens JM, Lee J, Pilch PF. Tumor necrosis factor-alpha-induced insulin resistance in 3T3-L1 adipocytes is accompanied by a loss of insulin receptor substrate-1 and GLUT4 expression without a loss of insulin receptor-mediated signal transduction. J Biol Chem. 1997;272(2):971-6.

261. Hirosumi J, Tuncman G, Chang L, Gorgun CZ, Uysal KT, Maeda K, et al. A central role for JNK in obesity and insulin resistance. Nature. 2002;420(6913):333-6.

262. De Fea K, Roth RA. Protein kinase C modulation of insulin receptor substrate-1 tyrosine phosphorylation requires serine 612. Biochemistry. 1997;36(42):12939-47.

263. Lim J, Iyer A, Liu L, Suen JY, Lohman RJ, Seow V, et al. Diet-induced obesity, adipose inflammation, and metabolic dysfunction correlating with PAR2 expression are attenuated by PAR2 antagonism. FASEB J. 2013;27(12):4757-67.

264. Badeanlou L, Furlan-Freguia C, Yang G, Ruf W, Samad F. Tissue factor-proteaseactivated receptor 2 signaling promotes diet-induced obesity and adipose inflammation. Nat Med. 2011;17(11):1490-7. 265. Li M, Yang X, Zhang Y, Chen L, Lu H, Li X, et al. Activation of proteaseactivated receptor2 is associated with increased expression of inflammatory factors in the adipose tissues of obese mice. Mol Med Rep. 2015;12(4):6227-34.

266. McArdle MA, Finucane OM, Connaughton RM, McMorrow AM, Roche HM. Mechanisms of obesity-induced inflammation and insulin resistance: insights into the emerging role of nutritional strategies. Front Endocrinol (Lausanne). 2013;4:52.

267. Howarth PH, Babu KS, Arshad HS, Lau L, Buckley M, McConnell W, et al. Tumour necrosis factor (TNFalpha) as a novel therapeutic target in symptomatic corticosteroid dependent asthma. Thorax. 2005;60(12):1012-8.

268. Neveu WA, Allard JL, Raymond DM, Bourassa LM, Burns SM, Bunn JY, et al. Elevation of IL-6 in the allergic asthmatic airway is independent of inflammation but associates with loss of central airway function. Respir Res. 2010;11:28.

269. Tillie-Leblond I, Pugin J, Marquette CH, Lamblin C, Saulnier F, Brichet A, et al. Balance between proinflammatory cytokines and their inhibitors in bronchial lavage from patients with status asthmaticus. Am J Respir Crit Care Med. 1999;159(2):487-94.

270. Borish L, Mascali JJ, Dishuck J, Beam WR, Martin RJ, Rosenwasser LJ. Detection of alveolar macrophage-derived IL-1 beta in asthma. Inhibition with corticosteroids. J Immunol. 1992;149(9):3078-82.

271. Loffredo LF, Abdala-Valencia H, Anekalla KR, Cuervo-Pardo L, Gottardi CJ, Berdnikovs S. Beyond epithelial-to-mesenchymal transition: Common suppression of differentiation programs underlies epithelial barrier dysfunction in mild, moderate, and severe asthma. Allergy. 2017.

272. Hyun E, Ramachandran R, Cenac N, Houle S, Rousset P, Saxena A, et al. Insulin modulates protease-activated receptor 2 signaling: implications for the innate immune response. J Immunol. 2010;184(5):2702-9.

273. Possa SS, Leick EA, Prado CM, Martins MA, Tiberio IF. Eosinophilic inflammation in allergic asthma. Front Pharmacol. 2013;4:46.

274. Monteseirin J. Neutrophils and asthma. J Investig Allergol Clin Immunol. 2009;19(5):340-54.

275. Chung KF, Barnes PJ. Cytokines in asthma. Thorax. 1999;54(9):825-57.
276. Sahiner UM, Birben E, Erzurum S, Sackesen C, Kalayci O. Oxidative stress in asthma. World Allergy Organ J. 2011;4(10):151-8.

277. Polosukhin VV, Lawson WE, Milstone AP, Egunova SM, Kulipanov AG, Tchuvakin SG, et al. Association of progressive structural changes in the bronchial epithelium with subepithelial fibrous remodeling: a potential role for hypoxia. Virchows Arch. 2007;451(4):793-803.

278. Polosukhin VV, Cates JM, Lawson WE, Milstone AP, Matafonov AG, Massion PP, et al. Hypoxia-inducible factor-1 signalling promotes goblet cell hyperplasia in airway epithelium. J Pathol. 2011;224(2):203-11.

279. Lee SY, Kwon S, Kim KH, Moon HS, Song JS, Park SH, et al. Expression of vascular endothelial growth factor and hypoxia-inducible factor in the airway of asthmatic patients. Ann Allergy Asthma Immunol. 2006;97(6):794-9.

280. Davydova ON, Yakovlev AA, Lyzhin AA, Khaspekov LG, Gulyaeva NV. Growth Factors Deprivation Induces a Specific Increase in PAR2 Receptor mRNA Expression in Primary Cerebellar Cultures. Neurochem J+. 2010;4(4):279-83.

# Chapter 2: Insulin decreases expression of the pro-inflammatory receptor proteinase-activated receptor-2 on human airway epithelial cells

Vivek D. Gandhi, M.Sc., Nami Shrestha Palikhe, Ph.D., Shereen M. Hamza, Ph.D., Jason R.B. Dyck, Ph.D., Jean Buteau, Ph.D., and Harissios Vliagoftis, M.D.

A version of this chapter has been accepted for publication.

DOI: https://doi.org/10.1016/j.jaci.2018.04.040

The chapter here has been expanded and supplemental data and methods have been included in the main manuscript for clarity.

# 2.1 Abstract

**Background:** Proteinase-Activated Receptor-2 (PAR-2) is implicated in the pathogenesis of allergic asthma. PAR-2 expression is increased on the airway epithelium of asthmatic individuals, but the factors responsible and the consequences of increased PAR-2 expression are poorly studied. Recently, decreased insulin signaling has been linked with airway epithelial changes in asthma.

**Objective:** To understand the role of insulin in PAR-2 regulation in the lungs *in vivo* and *in vitro*.

**Methods:** PAR-2 expression was studied in the lungs of mice with obesity following high fat diet (HFD). Submerged and air-liquid interface cultures of primary human airway epithelial cells (AECs) were used to elucidate the mechanisms underlying regulation of PAR-2 expression on AECs by insulin.

**Results:** HFD-fed mice that are characterized by increased levels of circulating insulin expressed lower PAR-2 levels in the lungs. Insulin deficiency in human AECs resulted in increased expression of PAR-2 mRNA and protein. PAR-2 upregulation increased PAR-2-dependent pro-inflammatory potential of AECs. PI3K inhibition prevented insulin-mediated regulation of PAR-

2 expression. Insulin, through PI3K, inhibits FOXO-1 transcription factor. Expression of constitutively active FOXO-1 resulted in increased PAR-2 expression in the presence of insulin and inhibition of FOXO-1 prevented insulin deprivation-induced PAR-2 upregulation.

**Conclusion:** Insulin may have an anti-inflammatory role in the airways by controlling PAR-2 expression on AECs through regulation of FOXO-1 activity. Alteration of insulin signaling through decreased insulin receptor expression or insulin resistance may be responsible for PAR-2 upregulation in asthmatic airway epithelium, an effect that may exacerbate proteinase-mediated airway inflammation.

#### **2.2 Introduction**

Proteinase-Activated Receptor-2 (PAR-2) is a G-protein coupled receptor activated by serine proteinases (1). PAR-2 expression is widespread on cells in the gut, skin, liver and other tissues and its activation induces inflammation (2-5). In lungs, PAR-2 is expressed on bronchial smooth muscle cells (6, 7), fibroblasts (8) and airway epithelial cells (AECs) (7), in addition to various immune cells. PAR-2 has been shown to be involved in the pathogenesis of a number of lung diseases such as allergic asthma (9, 10), idiopathic pulmonary fibrosis (IPF) (11) and pulmonary arterial hypertension (12).

PAR-2 activation on AECs induces the release of mediators that play important roles in the development of allergic sensitization (13, 14), recruitment of inflammatory cells (15, 16), eosinophil survival (16) and tissue remodeling (17). PAR-2 activation also affects epithelial barrier properties (18) resulting in increased interactions of inhaled allergens and pollutants with immune cells. These observations indicate that PAR-2 plays an important role in the pathogenesis of asthma.

PAR-2 expression increases in various disease states (3, 11, 19-21) including asthma (22-24), and PAR-2 over-expression can enhance inflammation (9). It is believed that the inflammatory milieu upregulates PAR-2 expression in these disease states, but this has not been proven. Inflammatory stimuli (25-27) and hypoxia (12, 28) can regulate PAR-2 expression on different cells but there have been no systematic studies addressing the detailed mechanisms of PAR-2 regulation. Our main interest has been to understand the role and regulation of PAR-2 in asthma, and especially on AECs, the first line of defense against inhaled inflammatory triggers and an important participant in the pathogenesis of inflammatory lung diseases.

Insulin has anti-inflammatory properties (29) but the role of these mechanisms in regulating airway inflammation in patients with asthma is not known. Insulin resistance, due to decreased expression of insulin receptor and/or blockade in insulin signaling, limits insulin-mediated anti-inflammatory activities. Insulin resistance is associated with allergic sensitization and allergic asthma in humans (30) and also in animal models (31, 32). Recent evidence also shows decreased expression of insulin receptor in the airway epithelium in asthmatic individuals (33),

which may render asthmatic epithelium insulin resistant. These results suggest that airway inflammation in asthma could be, at least in part, the result of decreased anti-inflammatory effects of insulin. Expression of the pro-inflammatory receptor PAR-2 increases in adipose tissue during insulin resistance, an increase that results in exaggerated PAR-2-mediated inflammation (34-36), while improving insulin sensitivity decreases PAR-2 expression and PAR-2-mediated inflammation (34).

We hypothesize that insulin-mediated regulation of PAR-2 expression explains the observed increase in PAR-2 expression on AECs of asthmatic individuals in a background of decreased insulin signaling. Here we studied the effects of insulin on PAR-2 expression in whole lungs and AECs using a mouse model and primary cultures of human bronchial epithelial cells, respectively. We show that PAR-2 expression increases with decreasing insulin concentrations. PAR-2 upregulation in the absence of insulin leads to increased calcium release upon PAR-2 activation, indicating that the pro-inflammatory potential of AECs may increase with decreasing insulin signaling. We also show that insulin-mediated regulation of PAR-2 expression in AECs is phosphoinositide 3-kinase (PI3K)-Akt- Forkhead box protein O1 (FOXO-1) pathway dependent.

## 2.3 Materials and Methods

#### Animal study

C57BL6 male mice (6-7 weeks old) were maintained on high fat diet (HFD) (60% kcal fat, Research Diets, New Brunswick, NJ) or standard chow diet (4% kcal fat) *ad libitum* for 30 weeks. Mice were euthanized with an overdose of sodium pentobarbital and blood was collected by cardiac puncture. Lungs were snap frozen in liquid nitrogen. Plasma and lung tissue were stored at -80°C until further processing. Experiments were approved by the University of Alberta Health Sciences Laboratory Animal Ethics committee.

#### Insulin measurement

Insulin levels were measured in mouse plasma using the Rodent Insulin Chemiluminescence ELISA kit (ALPCO, Salem, NH).

## Cell culture

Primary AECs (normal human bronchial epithelial cells), from five donors, were purchased from Lonza (Walkersville, MD); experiments were done on cells at passages 3-5 after receiving. The cells were cultured on tissue culture flasks or plates (Falcon®, Corning, NY) coated with 30 µg/mL bovine collagen type I (Life Technologies, Burlington, ON) dissolved in bronchial epithelial basal media (BEBM, Lonza). Cells were cultured in bronchial epithelial growth media (BEGM, Lonza) i.e. BEBM supplemented with bovine pituitary extract, hydrocortisone, epidermal growth factor, epinephrine, insulin, transferrin, triiodothyronine, retinoic acid, gentamycin-amphotericin-B and Penicillin-Streptomycin (Life Technologies).

Air-liquid interface (ALI) cultures of AECs were done as described (37). In summary, 90,000 cells /insert were seeded on 0.4  $\mu$ m inserts (Corning Inc., Corning, NY). The differentiation media contained 50% DMEM (high glucose, Life Technologies) and 50% BEGM with all the mentioned growth supplements except gentamicin-amphotericin B, triiodothyronine and retinoic acid. Gentamycin at final concentration of 50  $\mu$ g/mL (Life Technologies) was added separately. All-trans retinoic acid (Sigma) was added fresh every time at final concentration of 50 nM. For the first 7 days, the differentiation media was added to apical and basolateral sides. After 7 days,

apical media was removed to establish ALI and new media was added to only basolateral side. Cells were further cultured for 2 weeks to allow differentiation.

BEAS-2B cells (virus transformed AECs) were purchased from ATCC (Manassas, VA). The cells were cultured on tissue culture flasks or plates coated with a mixture of 2  $\mu$ g/mL fibronectin (Life Technologies), 30  $\mu$ g/mL bovine collagen type I and 10  $\mu$ g/mL BSA (Sigma-Aldrich, St. Louis, MO) dissolved in BEBM. BEGM media without gentamycin - amphotericin and retinoic acid was used to culture cells.

For all the cell culture procedure, cells were cultured at  $37^{\circ}$ C in 5% CO<sub>2</sub> and 90% humidity and media was changed 3 times /week. All cell cultures maintained in our lab were tested monthly for mycoplasma using the MycoAlert kit (Lonza) and only negative cultures were used.

To understand the role of insulin signaling pathway in PAR-2 regulation, AECs were treated with PI3K inhibitors LY294002 (20  $\mu$ M, (38, 39)) (Tocris, Minneapolis, MN) and wortmannin (5  $\mu$ M, (40)) (Cayman, Ann-Arbor, MI) and Forkhead box protein O1 (FOXO-1) inhibitor AS1842856 (1  $\mu$ M, (41)) (Calbiochem, San Diego, CA). All the inhibitors were dissolved in DMSO (Sigma-Aldrich) so that the final concentration of DMSO added to cells was less than 0.1%. DMSO at 0.1% concentration had no effect on PAR-2 expression (appendix-1).

# RNA extraction, reverse transcription and quantitative RT-PCR

RNA extraction was performed using RNeasy kit (Qiagen, Germantown, MD); 0.5 µg RNA was reverse transcribed using, 0.5 µg Oligo (dt) 12-18 (Life Technologies) and 200U M-MLV reverse transcriptase (Life Technologies) in 20 µl final volume.

Human *PAR-2* mRNA was quantified using a custom 6FAM-labeled DTAM probe (5'-TAA GGT TGA TGG CAC ATC CCA CGT CAC TGG -3') (ThermoFisher Scientific) and the following primers *PAR-2-*F (5'-TGC TAG CAG CCT CTC TCT CC -3') and *PAR-2-*R (5'-CCA GTG AGG ACA GAT GCA GA-3'). *GAPDH* mRNA was used as an internal control and was quantified using a custom 6FAM-labeled TAMRA probe (5'-AAA TCC CAT CAC CAT CTT CCA GGA GCG A-3') (ThermoFisher Scientific) and the following primers *GAPDH-*F (5'-CTG AGA ACG GGA AGC TTG TCA -3') and *GAPDH-*R (5'-GCA AAT GAG CCC CAG CCT T-

3'). The PCR protocol consisted of 10 minutes at 94°C followed by 40 cycles of 15 seconds at 94°C, 25 seconds at 62°C and 15 seconds at 72°C. All samples were run in triplicate. The *PAR-2* and *GAPDH* PCR products were cloned into the pCR2.1-TOPO vector (Life Technologies), quantified and serial dilutions of linearized plasmid were used to generate a standard curve of known copy numbers for each gene. Data were calculated and represented *PAR-2* copies /1000 *GAPDH* copies. PAR-2 expression was also normalized to a second housekeeping gene, *Cyclophilin A (PPIA)*. For this Taqman gene expression assays were used for *PAR-2* (Hs00608346\_m1, ThermoFisher Scientific) and housekeeping gene *Cyclophilin A* (Hs04194521\_s1, ThermoFisher Scientific).

Mouse *PAR-2* expression was measured using Taqman gene expression assay (Mm0043360\_m1, ThermoFisher Scientific, Waltham, MA) as per manufacturer's instructions; *HPRT* (Mm03024075\_m1, ThermoFisher Scientific) was used as a house keeping gene.

# Confocal microscopy

AECs were cultured on collagen coated glass coverslips and subjected to various treatments. Cells were then rinsed with PBS and fixed with 4% paraformaldehyde-PBS for 15 minutes (min) at room temperature, rinsed (with PBS 3X 5 min), permeabilized using 0.5% triton X100-PBS for 10 minutes and then rinsed to remove triton X100. Cells were then blocked using 1% BSA-PBS for 30 min at room temperature to decrease nonspecific binding of antibodies; excess BSA was removed by rinsing cells. Cells were then incubated overnight at 4°C with 2 µg anti-PAR-2 monoclonal primary antibody (SAM-11, Santa Cruz Biotechnology, Dallas, TX) in 1% BSA-PBS. Cells were then rinsed to remove unbound antibody and incubated with 4 µg secondary antibody (Alexa Fluor 488 goat anti-mouse IgG2a, Molecular Probes) in 1% BSA-PBS for 90 min at room temperature. Cells stained with only secondary antibody were used as negative control. Cells were again rinsed to remove unbound antibody and incubated with 5 µM DAPI (Sigma-Aldrich) for 10 min in dark at room temperature. Cells were rinsed with water and mounted in prolong gold without DAPI (Molecular Probes) and allowed to cure overnight at room temperature in dark. Images were obtained using a 40X objective on a Leica SP5 confocal microscope. Images were analysed using ImageJ software (NIH) to calculate Corrected Total Cell Fluorescence (CTCF).

CTCF = Integrated density - (Area of selected cell X Mean fluorescence of background reading)

In the case of cell transfection with FOXO-1-GFP, live imaging was performed for up to 40 min (picture captured at every 10 min interval) using 20X objective on WaveFX spinning disk confocal microscope (Quorum Technologies).

# $Ca^{++}$ release assay

PAR-2-mediated Ca<sup>++</sup> release from intracellular stores, endoplasmic reticulum (ER), was measured using Fluo-4 NW Calcium Assay Kit (Life Technologies) as per manufacturer's instructions. In summary, AECs were cultured in 96 well plates (Corning® Clear Bottom Black plates) and, when 85% confluent, were treated as required for each experiment. At the end of the treatment, cells were incubated with 100 µl of dye loading solution containing probenecid for 30 min each at 37° C and at room temperature. Florescence measurements representing cytoplasmic Ca<sup>++</sup> levels were recorded by setting the instrument at 485 nm wavelength for excitation and 516 nm wavelength for emission. The baseline fluorescence was first recorded for 100 seconds. After this 10 µl of PAR-2 activating peptide (PAR-2 AP) or control peptide (PAR-2 CP) (50 µM final concentration) were added and fluorescence was recorded for up to 180 seconds. To study the PAR-2-mediated  $Ca^{++}$  release from intracellular store, endoplasmic reticulum (ER), we performed the assay in calcium, magnesium free HBSS buffer (Fisher Scientific). This ensured that the increase in cytoplasmic  $Ca^{++}$  is due to the release of  $Ca^{++}$  from intracellular stores and not due to entry of Ca<sup>++</sup> from the assay buffer. After PAR-2 AP or CP, 1 µM Thapsigargin (Sigma-Aldrich) was added and fluorescence was measured up to 300 seconds. Thapsigargin induces Ca<sup>++</sup> release from ER (42). PAR-2-mediated Ca<sup>++</sup> release from ER was calculated as % of thapsigargin-induced release from ER.

# Preparation of PAR-2 reporter plasmid to study PAR-2 promoter activity

The upstream region of transcription start site is considered as a putative promoter region. We used human genomic DNA as a template to amplify a 670 base pair (bp) fragment covering exon-1 (288 bp) and part of the upstream region of Exon-1 of PAR-2 gene using forward primer 5'-CTCCTCCTGCAGGGTCCACC-3', and reverse primer 5'-AGAGCCCTCCTTGGCCAGGT-3'. A kpn 1 restriction digestion site was added to the forward

primer and a Xho 1 restriction digestion site was added to the reverse primer. The PCR product was gel purified with a gel purification kit (Qiagen) and then ligated into the TOPO vector (Life Technologies) using TA cloning. The vector was then sequenced and confirmed for the orientation of the 670 bp DNA fragment. The TOPO vectors were then digested with kpn I and Xho I restriction enzymes to recover the 670 bp fragment DNA. The pGL3-Basic vector (Promega, Madison, WI) was also digested with kpn I and Xho I restriction enzymes. The recovered 670 bp DNA fragment was ligated into the digested pGL3-Basic vector upstream of the firefly luciferase gene to prepare a PAR-2 reporter plasmid. The plasmid DNA was purified using the plasmid maxi kit (Qiagen). The plasmid concentration and purity were measured by Nanodrop 2000c (Thermo Scientific). The size of 670 bp construct provided us with a small size reporter plasmid to increase transfection efficiency and also enabled us to design specific primers to amplify the fragment of interest. However, further different sizes of promoter constructs need to be studied to fully understand the PAR-2 promoter region.

#### Transfection studies

A pBluescript containing the coding sequence for a constitutively active form of transcription factor FOXO-1 (FOXO-1 CA) was used (43). AECs were nucleofected with 2-4  $\mu$ g of FOXO-1 CA or GFP-encoding plasmid (pmax-GFP) or empty pBluescript according to the manufacturer's instruction using Amaxa Basic Nucleofector kit for Primary Mammalian Epithelial cells (Lonza), or Ingenio Electroporation kit (Mirus Bio, Madison, WI) and Amaxa Nucleofector 2s (W-005 program, Lonza).

AECs were transfected with a previously described FOXO-1-GFP plasmid (43) using Mirus TransIT-X2 reagent as per manufacturer's instructions (Please see Appendix-2 for transfection optimization procedure with 3 different transfection reagents from Mirus). In summary, AECs were cultured in 35mm tissue culture dishes (Falcon®, Fisher Scientific) until 85% confluent. Cells were then rinsed with PBS and 2 mL of BEGM media without antibiotics was added. Transfection complexes were formed by mixing plasmid DNA (2.5  $\mu$ g) and transfection reagent (5  $\mu$ L) in 250  $\mu$ l BEBM media for 30 min at room temperature and added drop wise to the culture dishes. The next day, transfected cells were rinsed and incubated with 0.5  $\mu$ g/mL Hoechst stain at 37°C for 10 min to stain nuclei. Cells were then rinsed and cultured in insulin-deficient

media for 2-4h to allow nuclear localization of FOXO-1-GFP. Insulin (5 µg/mL) was then added and the movement of FOXO-1-GFP was studied by live imaging for up to 40 min using confocal microscopy.

For promoter studies, BEAS-2B cells were seeded in 12-well plates (Falcon) and cultured until 85% confluent. On the day of transfection, cells containing media was replaced with BEBM media (400  $\mu$ L) after rinsing the cells with PBS. The transfection complexes were formed by mixing 1  $\mu$ g PAR-2 reporter plasmid DNA or empty pGL3-Basic vector (negative control), 5 ng Renilla plasmid DNA and 2  $\mu$ g of Lipofectamine 2000 transfection reagent (Life Technologies) in 100  $\mu$ L BEBM. They were incubated for 30 min at room temperature, after which 100  $\mu$ L of each transfection complexes were then removed and growth media was added for 24h. The next day, to study the effect of insulin on PAR-2 promoter, cells were cultured with or without insulin for 24h and promoter activity was assessed by performing luciferase reporter assay.

# Luciferase Reporter Assay

At the end of the incubation period, cells were harvested using a cell scraper, pelleted and lysed with 60  $\mu$ L passive lysis buffer (Promega) per well. From the 60  $\mu$ L lysate, a 50  $\mu$ L aliquot was assayed for luciferase activity using Dual –Luciferase Reporter Assay Kit (Promega) and the signal was measured using Turner BioSystems 20/20<sup>n</sup> Luminometer. The measured luciferase activity is proportional to the amount of produced firefly luciferase enzyme and thus proportional to the promoter activity of the cloned region. To control for any effect of the plasmid itself, one set of cells were transfected with empty pGL3-Basic vector. Renilla is a constitutive active luciferase value of each sample was normalized by Renilla luciferase of the same sample (firefly luciferase / Renilla luciferase), which gave us the promoter activity of the PAR-2 promoter activity in cells transfected with empty pGL3-Basic vector). To study the effect of insulin deprivation on PAR-2 promoter activity, results were expressed as fold change in promoter activity over cells cultured with insulin.

## Western blot

To study the presence of phospho-Akt in mouse lungs, we used rabbit anti-Akt and rabbit antiphospho-Akt (Ser473) (Cell Signaling, Danvers, MA) primary antibodies, followed by HRP labelled secondary antibody (goat anti-rabbit IgG, Santa Cruz) and the membranes were developed with ECL reagents (ThermoFisher Scientific). To study insulin-mediated FOXO-1 phosphorylation in AECs, we used rabbit anti-FOXO-1 and rabbit anti-Phospho-FOXO-1 (Ser256) (Cell Signaling) primary antibodies, followed by IRdye® conjugated secondary antibody (goat anti-rabbit IgG, LI-COR, Lincoln, NE) and membranes were imaged using Odyssey Infrared Imager (LI-COR). Images were analyzed using Image Studio Lite software (LI-COR) and results were expressed as a ratio of Phospho-Akt/total Akt and Phospho-FOXO-1/total FOXO-1.

## **Statistics**

The results are expressed as Mean  $\pm$  SEM. In case of comparisons between two groups, twotailed paired *t*-test was performed, except for the mouse experiments where two-tailed Mann Whitney test was used. In case of three or more groups, repeated measures ANOVA was used with Tukey's multiple comparison test. The *p*-value of 0.05 or less was considered statistically significant.

## 2.4 Results

## PAR-2 expression in lungs of HFD-fed mice

HFD induces insulin resistance (44) in skeletal muscle, liver and adipose tissue. Insulin resistance modulates PAR-2 expression in adipose tissue (34-36) indicating that insulin regulates PAR-2 expression. We hypothesized that insulin also regulates PAR-2 expression in the lungs of HFD-fed mice. As expected, plasma insulin levels were higher in mice fed with HFD compared to mice fed a control diet (Fig. 2.1A) but there was no significant difference in downstream insulin signaling as measured by phospho-Akt levels in whole lungs (Fig. 2.1B and 2.1C), indicating absence of altered insulin signaling in the lungs of HFD-fed mice. Finally, PAR-2 mRNA expression was decreased in lungs of HFD-fed mice (Fig. 2.1D). From these results we hypothesized that increased circulating insulin levels suppressed PAR-2 expression in lungs of HFD-fed mice. This hypothesis may explain the observed PAR-2 upregulation on AECs of asthmatic individuals (24) in a background of decreased insulin signaling (33).



Fig. 2.1 Plasma insulin levels, Akt phosphorylation in lungs and PAR-2 expression in lungs of HFD- and control diet-fed mice (A) Plasma insulin levels in HFD- and control diet-fed mice, (B) a western blot and (C) densitometric analysis of Akt phosphorylation in lungs of HFD- and control diet-fed mice and (D) PAR-2 expression in lungs of HFD- and control diet-fed mice. Data are presented as Mean  $\pm$  SEM from n=9 for HFD-fed and n=5 for control diet-fed mice; ns (not significant), \*\**P*<0.01.

## Insulin regulates PAR-2 mRNA expression in primary human AECs

To explore the role of insulin in regulating PAR-2 expression in primary human AECs, we cultured cells in media with all the growth supplements (media with insulin) or media with all the growth supplements except insulin (media without insulin) for up to 48h and PAR-2 expression was studied by qRT-PCR. Insulin deficiency resulted in PAR-2 mRNA upregulation at all time points compared to cells cultured in media with insulin (Supplementary Fig. 2.E1A and 2.E1B), while PAR-2 expression remained constant at all time points in cells cultured in media with insulin. PAR-2 mRNA upregulation in the absence of insulin was confirmed using

another housekeeping gene, cyclophilin A (Supplementary Fig. 2.E1C). PAR-2 mRNA upregulation in the absence of insulin plateaued at 24h (Fig. 2.2A). Subsequent studies to define the mechanism of insulin-mediated PAR-2 regulation were performed with cells cultured without insulin for 24h. Addition of insulin reversed insulin deficiency-induced PAR-2 upregulation (Fig. 2.2B), indicating that this is a dynamic change in PAR-2 expression and not the effect of permanent alterations in cellular biology due to the absence of insulin. To study the cause and effect relationship between the levels of insulin and PAR-2 expression, we cultured AECs with various concentration of insulin, in the absence of all other growth supplements. Insulin concentrations between  $0.2 \mu g/mL$  and  $125 \mu g/mL$  dose dependently regulated PAR-2 expression (Fig. 2.2C).

We then studied whether insulin deficiency-induced PAR-2 mRNA upregulation is the result of increased PAR-2 mRNA stability or increased PAR-2 gene transcription. The rate of PAR-2 mRNA degradation in the presence of actinomycin D, was the same whether cells were previously incubated for 24h in the presence or absence of insulin, indicated that the absence of insulin did not alter PAR-2 mRNA stability (Fig. 2.2D). To study whether insulin directly affects PAR-2 mRNA transcription we used a 670 bp PAR-2 reporter plasmid that we have generated. Since primary AECs are difficult to transfect we used the BEAS-2B cell line (45) for these experiments. Insulin deficiency resulted in PAR-2 mRNA upregulation in BEAS-2B cells, similar to primary AECs (Fig. 2.2E). Insulin deficiency in BEAS-2B cells transfected with the PAR-2 reporter construct increased PAR-2 promoter activity (Fig. 2.2F) indicating that insulin regulates PAR-2 gene transcription.



Fig. 2.2 Insulin transcriptionally regulates PAR-2 mRNA expression in AECs. (A) Fold change in PAR-2 mRNA expression in cells cultured in the absence of insulin compared to cells cultured in the presence of insulin for up to 48h (n=3-6). (B) Reversibility of insulin deficiency-induced upregulation of PAR-2 mRNA expression by insulin (n=4). After the first 24h of culture in the presence or absence of insulin (left part of the graph), cells cultured in the absence of insulin were further cultured with or without insulin for next 24h (right part of the graph). (C) PAR-2 mRNA expression in AECs cultured in the presence of various concentrations of insulin for 24h (n=3). (D) The rate of PAR-2 mRNA degradation for 8h after the addition of actinomycin D in cells cultured in the presence (Black) or absence (Red) of insulin for 24h (n=4). PAR-2 levels are

expressed as % PAR-2 expression compared to time "0". (E) PAR-2 mRNA expression in BEAS-2B cells cultured in the presence or absence of insulin for 24h (n=3). (F) PAR-2 promoter activity in BEAS-2B cells transfected with the PAR-2 reporter plasmid and cultured in the presence or absence of insulin for 24h (n=8). Data are presented as Mean  $\pm$  SEM; ns (not significant), \**P*<0.05, \*\**P*<0.01, \*\*\**P*<0.001 between the groups indicated by horizontal lines.

#### **Insulin regulates functional PAR-2 expression in AECs**

Insulin deficiency also increased PAR-2 protein expression in AECs after 24h, as shown by confocal microscopy (Fig. 2.3A-C). It appears that during insulin deficient condition number of cells expressing PAR-2 is also increased (Fig. 2.3B). However, further studies are required where PAR-2-positive cells from multiple field of view are calculated to reach this conclusion.

As a Gq-coupled receptor, PAR-2 activation leads to release of Ca<sup>++</sup> from intracellular stores ER, which activates calcium release-activated calcium channels and induces release of inflammatory mediators from AECs (46). To study if PAR-2 upregulation increases PAR-2-mediated calcium signaling, we measured PAR-2-mediated Ca<sup>++</sup> release from ER using PAR-2 activation peptide (PAR-2 AP) or control peptide (PAR-2 CP). PAR-2 AP induced Ca<sup>++</sup> release in AECs while PAR-2 CP did not (Supplementary Fig. 2.E2A). PAR-2 AP did not induce further Ca<sup>++</sup> release in cells activated with thapsigargin indicating PAR-2-mediated Ca<sup>++</sup> release is from ER (Supplementary Fig. 2.E2B). PAR-2 AP-mediated activation of cells cultured in the absence of insulin led to increased Ca<sup>++</sup> release compared to activation of cells cultured in the presence of insulin (Fig. 2.4A and 2.4B) indicating that insulin-mediated regulation of PAR-2 expression may affect PAR-2-mediated activation of AECs.



Fig. 2.3 Insulin regulates PAR-2 protein expression in AECs. (A-B) Images of PAR-2 expression (Green) in (A) a representative individual cell and (B) a representative field of view displaying multiple cells cultured in the presence (left image) or absence (right image) of insulin for 24h. Nuclei are stained with DAPI (Blue). (C) Corrected Total Cell Fluorescence intensity of PAR-2 expression in AECs cultured in the presence or absence of insulin (Fluorescence was calculated for 30 cells / experiment, n=3). Data are presented as Mean  $\pm$  SEM; \*\*\**P*<0.001.



Fig. 2.4 Insulin regulates PAR-2-mediated activation of AECs. (A) PAR-2-mediated Ca<sup>++</sup> release from ER in cells cultured in the presence (Blue line) and absence of insulin (Red line) for 24h. (B) Quantification of the intracellular Ca<sup>++</sup> release by PAR-2 as a percentage of the release induced by thapsigargin (n=7). Data are presented as Mean  $\pm$  SEM; \*\**P*<0.01.

# Insulin regulates PAR-2 expression through PI3K-Akt pathway

Upon binding to insulin receptor, insulin activates the PI3K-Akt signaling pathway which mediates many of the insulin effects (47). PI3K inhibition using LY294002 in cells cultured in the presence of insulin showed increase in PAR-2 expression compared to cells cultured in the presence of insulin without the inhibitor (Fig. 2.5A). Since insulin in the absence of all other growth supplements regulates PAR-2 expression (Fig. 2.2C), we then cultured cells in the presence of insulin or in media without insulin with or without an irreversible PI3K inhibitor wortmannin. Again, PI3K inhibition in the presence of insulin resulted in increased PAR-2 expression (Fig. 2.5B). These results indicate that insulin regulates PAR-2 through PI3K-Akt pathway. To investigate the role of PI3K-Akt pathway in the observed insulin-mediated reversibility of upregulated PAR-2 expression (Fig. 2.2B), we cultured cells in the absence of insulin to upregulate PAR-2 and then added insulin with or without PI3K inhibitors. PI3K inhibition with LY294002 or wortmannin prevented insulin from downregulating PAR-2 expression (Fig. 2.5C).



Fig. 2.5 Insulin regulates PAR-2 expression through PI3K activation. PAR-2 mRNA expression in cells cultured in the presence or absence of insulin with or without the PI3K inhibitor (A) LY294002 (n=3) or (B) Wortmannin (n=4) for 24h. PAR-2 mRNA expression in cells that were first cultured in the absence of insulin for 24h and then further cultured in the absence of insulin or in the presence of insulin with PI3K inhibitor (C) LY294002 (n=6) and (D) Wortmannin (n=4) for another 24h. Data are presented as Mean  $\pm$  SEM; ns (not significant), \**P*<0.05, \*\**P*<0.01, \*\*\**P*<0.001 between the groups indicated by horizontal lines.

#### Insulin regulates PAR-2 expression through the transcription factor FOXO-1

FOXO-1 transcription factor is a direct Akt target (48). Insulin, through Akt, phosphorylates FOXO-1 on 3 sites (T24, S253, S316) leading to nuclear exclusion and decreased transcription factor activity (43, 49). Our in silico analysis of the 670 bp PAR-2 promoter indicated that the promoter contains FOXO-1 binding sites (Appendix-3). We therefore hypothesized that FOXO-1 becomes active during insulin deficiency and induces transcription of PAR-2 in AECs.

We first studied if insulin phosphorylates FOXO-1 and induces its nuclear exclusion in primary AECs. Addition of insulin to AECs cultured in the absence of insulin resulted in phosphorylation of endogenous FOXO-1 (Fig. 2.6A and 2.6B). To study if insulin induces nuclear exclusion of FOXO-1, cells were transfected with a FOXO-1-GFP plasmid. FOXO-1-GFP localized to AEC nucleus in the absence of insulin (Fig. 2.6C, left image) and addition of insulin induced nuclear exclusion of FOXO-1-GFP (Fig. 2.6C, right image and Supplementary Video 1). These results indicate that insulin inhibits FOXO-1 activity in primary AECs.

Since insulin was able to inhibit FOXO-1 in AECs, we next studied if FOXO-1 can induce PAR-2 expression in AECs. Cells were transfected with a plasmid coding for constitutively active FOXO-1, where 3 phosphorylation sites are mutated (T24A, S253D, S316A) rendering FOXO-1 insensitive to insulin-Akt-mediated phosphorylation and inhibition. Transfection of AECs with the constitutively active FOXO-1 CA upregulated PAR-2 expression in cells cultured in the presence of insulin (Fig. 2.6D) indicating that FOXO-1 can mediate PAR-2 expression. Finally, to study the role of FOXO-1 in insulin deficiency-induced PAR-2 upregulation, we inhibited FOXO-1 using an inhibitor AS1842856 in cells cultured in the media without insulin. FOXO-1 inhibition prevented insulin deficiency-induced upregulation of PAR-2 (Fig. 2.6E).



Fig. 2.6 Insulin regulates PAR-2 expression by controlling FOXO-1 transcription factor in AECs. AECs were incubated in media without insulin for 4-6h and then insulin (5  $\mu$ g/ml) was added for the indicated time. (A) A representative blot and (B) densitometric analysis of FOXO-1 phosphorylation by insulin at indicated time points (n=3). (C) Representative images of insulin-induced nuclear exclusion of FOXO-1 (Green). FOXO-1-GFP transfected cells cultured without insulin (left image) and after addition of insulin (right image) (D). PAR-2 mRNA expression in AECs nucleofected with FOXO-1 CA or pmaxGFP or empty pBluescript (n=5) and cultured in the presence of insulin for 24 h. (E) PAR-2 mRNA expression in AECs cultured in the presence of insulin with or without FOXO-1 inhibitor AS1842856 for 24h. Data are presented as Mean  $\pm$  SEM; ns (not significant), \**P*<0.05, \*\**P*<0.01, \*\*\**P*<0.001 between the groups indicated by horizontal lines.

#### ALI cultures of AECs show similar insulin-mediated PAR-2 regulation

All studies reported above were performed with submerged cultures of AECs. However, submerged cultures are not differentiated and polarized as *in vivo*. We therefore, studied the effect of insulin on ALI cultures. ALI cultures showed the presence of ciliated cells demonstrating differentiation of AECs (Fig. 2.7A). Insulin deficiency in ALI cultures resulted in PAR-2 upregulation (Fig. 2.7B) similar to that observed in submerged cell cultures. In addition, as in submerged cultures, FOXO-1 inhibition prevented insulin deficiency-induced PAR-2 upregulation in ALI cultures (Fig. 2.7C).



Fig. 2.7 Differentiated cultures of AECs show similar insulin-mediated PAR-2 regulation. (A) Haematoxylin and Eosin (H&E) staining of AECs cultured at Air-Liquid Interface (ALI) to allow cellular differentiation. (B) PAR-2 mRNA expression in differentiated AECs in the presence or absence of insulin for up to 72h (n=4). (C) PAR-2 mRNA expression in differentiated AECs in the presence or absence of insulin with or without FOXO-1 inhibitor AS1842856 for 24h (n=3). Data are presented as Mean  $\pm$  SEM; ns (not significant), \**P*<0.05, \*\*\**P*<0.001 between the groups indicated by horizontal lines.

#### **2.5 Discussion**

The serine proteinase receptor PAR-2 is involved in the pathogenesis of allergic asthma (9, 10). In addition, PAR-2 expression is increased in AECs of patients with asthma (24), but the regulation of PAR-2 expression in AECs is poorly studied. We show that insulin regulates expression of PAR-2 on primary AECs through a PI3K-Akt-FOXO-1 pathway. Decreasing concentrations of insulin lead to upregulation of PAR-2 expression and increased potential for PAR-2-mediated activation of epithelial cells. To our knowledge, this is the first report providing a detailed mechanism of PAR-2 regulation in primary human bronchial epithelial cells. In addition, this is the first evidence showing regulation of PAR-2 by a hormone with anti-inflammatory activities.

Recently Loffredo et al. (33) analyzed published microarray data and found decreased expression of insulin receptor in AECs obtained from asthmatic individuals. The authors also noted that decreased insulin receptor expression was associated with downregulation of genes involved in epithelial differentiation and validated this observation ex vivo. For their experiments they excluded insulin from ALI cultures of normal AECs throughout the differentiation period and showed that by decreasing insulin signaling they were able to mimic the decreased differentiation of AECs observed in asthma in vivo. In our experiments, we cultured AECs with insulin during the differentiation phase of the ALI cultures, and removed insulin after the cultures were fully differentiated. We found that insulin deficiency resulted in PAR-2 upregulation indicating that disruption in insulin signaling could result in increased PAR-2 expression in fully differentiated normal human AECs. We did not validate our confocal microscopy results with another protein detection method for a number of reasons as discussed below. AECs are strongly adherent cells and trypsin is required to detach cells from the cell culture vessels. Trypsin is a PAR-2 agonist and may result in the loss of cell surface PAR-2 expression and recognition of the receptor by the antibody (50) we have been using for detection in flow cytometry in other studies (51). Hence flow cytometry will not be helpful at answering if cell surface PAR-2 expression is increased in the absence of insulin. We also did not use western blot for analysing total PAR-2 expression in the presence or absence of insulin. The molecular weight of PAR-2, based on its amino acid sequence, is 44 kilodalton (kDA) (52). Based on the

post-translational modifications, PAR-2 molecular weight ranges from 44-100 kDA, which results in multiple bands, on a western blot, representing different post-translationally modified forms of PAR-2 (53); multiple bands make difficult to compare and conclude if PAR-2 expression is increased in the absence of insulin.

PAR-2 activation in the airways plays a role in the development of allergic sensitization (54), allergic airway inflammation (10) and airway remodeling (55), while the degree of airway inflammation in these cases may depend on the levels of PAR-2 expression (9). These effects are mediated by the plethora of inflammatory mediators released by PAR-2-activated AECs (13-17); mediator release that in most cases depends on PAR-2-mediated calcium signaling (46). We show here that AECs grown in the absence of insulin exhibit increased calcium signaling following PAR-2-mediated activation, which may result in higher inflammatory mediator release. Thus insulin, by regulating PAR-2 expression, could regulate PAR-2-mediated inflammation.

Insulin has indirect anti-inflammatory properties by preventing hyperglycemia; however, there is increasing evidence that insulin has also direct anti-inflammatory properties by regulating the activity of transcription factors and controlling expression of chemokine receptors, endothelial adhesion molecules and Toll-like receptors (29). There is limited information on the effects of insulin on AECs. Insulin regulates expression of surfactants (56), glucose uptake and barrier function (57) in AECs through the PI3K-Akt pathway. Older studies also suggested that insulin plays a role in wound healing by being a direct chemotactic factor for AECs (58, 59) and/or by regulating expression of fibronectin in AECs (60). Our results demonstrate another anti-inflammatory effect of insulin on AECs through inhibition of the expression of a pro-inflammatory receptor, PAR-2. Thus, alteration in insulin signaling may increase the possibilities for the development of PAR-2-mediated inflammation.

Insulin has also anti-inflammatory effects by regulating the activity of transcription factors and thus inhibiting release of inflammatory mediators. Insulin can inhibit NF- $\kappa$ B transcription factor indirectly by increasing the expression of I $\kappa$ B (61). The FOXO family of transcription factors is under the direct negative regulation of insulin-mediated PI3K-Akt signaling (48). There is limited information on the role of FOXO transcription factors in AEC biology. FOXO-1 is

activated upon bacterial infection of AECs and induces inflammatory mediators and defensins, an effect that is negatively regulated by insulin (62). Our results indicate that with decreasing insulin signaling, FOXO-1 becomes activated and induces PAR-2 transcription leading to increased PAR-2 expression and increased pro-inflammatory potential of the epithelial cells.

An important question is whether decreased *in vivo* insulin signaling in asthma is only the result of decreased receptor expression (33) or there is also an element of insulin resistance. There are two basic animal models to study the effect of insulin *in vivo*; insulin deficient mice (by destruction of pancreatic  $\beta$  cells) and insulin resistant mice. PAR-2-mediated acute paw inflammation is higher in insulin deficient mice (63). Although alterations in PAR-2 expression were not studied in these mice, the effect described could be the result of increased PAR-2 expression in the absence of insulin. HFD-induced obesity induces insulin resistance in mouse adipose tissue, and has been associated with increased PAR-2 expression in adipose tissue and increased PAR-2-mediated adipose tissue inflammation (34-36). However, the authors did not study whether there is a direct pathway between insulin resistance and PAR-2 expression or increased PAR-2 expression is the result of alterations in other inflammatory pathways in HFDfed mice. Using a similar model of HFD-induced obesity, we did not find alteration in insulin signaling in lung tissue but we found that PAR-2 expression was decreased. We hypothesized that hyperinsulinemia in these mice suppresses PAR-2 expression on AECs. Our *in vitro* data, which show that the higher the concentration of insulin in the media, the lower the PAR-2 expression in AECs, support this observation.

Our results predict that PAR-2 expression should increase in the presence of insulin resistance in airway epithelium. Whether this phenomenon underlies the increased PAR-2 expression seen on the airway epithelium in asthma (24), IPF (11) or inflammatory conditions in other tissues (3, 19-21), requires further study. Inflammatory mediators such as TNF, IL-6 and IL-1 $\beta$  induce insulin resistance in adipose tissue (64). These mediators are also increased in the airway of asthmatics (65-68). Thus, it is possible that increased PAR-2 expression in asthma is the result of localized insulin resistance in the airway epithelium induced by the inflammatory milieu rather than an effect of an individual inflammatory mediator as shown in various *in vitro* studies (25-27). To answer this question, mucosal biopsies from asthmatic patients could be studied for the presence

of insulin resistance markers such as decreased insulin receptor expression (69) or serine phosphorylation of IRS-1 component of insulin receptor signaling cascade (70-73).

In conclusion, our data suggest a potential mechanism for the increased expression of PAR-2 on epithelial cells in asthma and possibly other inflammatory conditions, where PAR-2 plays a role in the disease pathogenesis. Understanding insulin-mediated regulation of PAR-2 expression in a variety of inflammatory diseases may uncover common dysregulatory mechanisms underlying increased PAR-2 expression and lead to the development of novel therapeutic strategies to regulate PAR-2 expression and PAR-2-mediated inflammation.

# 2.6 References

1. Cottrell GS, Amadesi S, Schmidlin F, Bunnett N. Protease-activated receptor 2: activation, signalling and function. Biochem Soc Trans. 2003;31(Pt 6):1191-7.

2. Lohman RJ, Cotterell AJ, Suen J, Liu L, Do AT, Vesey DA, et al. Antagonism of protease-activated receptor 2 protects against experimental colitis. J Pharmacol Exp Ther. 2012;340(2):256-65.

3. Steinhoff M, Corvera CU, Thoma MS, Kong W, McAlpine BE, Caughey GH, et al. Proteinase-activated receptor-2 in human skin: tissue distribution and activation of keratinocytes by mast cell tryptase. Exp Dermatol. 1999;8(4):282-94.

4. Knight V, Tchongue J, Lourensz D, Tipping P, Sievert W. Protease-activated receptor 2 promotes experimental liver fibrosis in mice and activates human hepatic stellate cells. Hepatology. 2012;55(3):879-87.

5. Xue M, Chan YK, Shen K, Dervish S, March L, Sambrook PN, et al. Protease-activated receptor 2, rather than protease-activated receptor 1, contributes to the aggressive properties of synovial fibroblasts in rheumatoid arthritis. Arthritis Rheum. 2012;64(1):88-98.

6. Schmidlin F, Amadesi S, Vidil R, Trevisani M, Martinet N, Caughey G, et al. Expression and function of proteinase-activated receptor 2 in human bronchial smooth muscle. Am J Respir Crit Care Med. 2001;164(7):1276-81.

7. Miotto D, Hollenberg MD, Bunnett NW, Papi A, Braccioni F, Boschetto P, et al. Expression of protease activated receptor-2 (PAR-2) in central airways of smokers and non-smokers. Thorax. 2002;57(2):146-51.

8. Matsushima R, Takahashi A, Nakaya Y, Maezawa H, Miki M, Nakamura Y, et al. Human airway trypsin-like protease stimulates human bronchial fibroblast proliferation in a protease-activated receptor-2-dependent pathway. Am J Physiol Lung Cell Mol Physiol. 2006;290(2):L385-95.

9. Schmidlin F, Amadesi S, Dabbagh K, Lewis DE, Knott P, Bunnett NW, et al. Proteaseactivated receptor 2 mediates eosinophil infiltration and hyperreactivity in allergic inflammation of the airway. J Immunol. 2002;169(9):5315-21. 10. Asaduzzaman M, Nadeem A, Arizmendi N, Davidson C, Nichols HL, Abel M, et al. Functional inhibition of PAR2 alleviates allergen-induced airway hyperresponsiveness and inflammation. Clin Exp Allergy. 2015;45(12):1844-55.

11. Wygrecka M, Kwapiszewska G, Jablonska E, von Gerlach S, Henneke I, Zakrzewicz D, et al. Role of protease-activated receptor-2 in idiopathic pulmonary fibrosis. Am J Respir Crit Care Med. 2011;183(12):1703-14.

12. Kwapiszewska G, Markart P, Dahal BK, Kojonazarov B, Marsh LM, Schermuly RT, et al. PAR-2 inhibition reverses experimental pulmonary hypertension. Circ Res. 2012;110(9):1179-91.

13. Kouzaki H, O'Grady SM, Lawrence CB, Kita H. Proteases induce production of thymic stromal lymphopoietin by airway epithelial cells through protease-activated receptor-2. J Immunol. 2009;183(2):1427-34.

14. Kouzaki H, Tojima I, Kita H, Shimizu T. Transcription of interleukin-25 and extracellular release of the protein is regulated by allergen proteases in airway epithelial cells. Am J Respir Cell Mol Biol. 2013;49(5):741-50.

15. Asokananthan N, Graham PT, Fink J, Knight DA, Bakker AJ, McWilliam AS, et al. Activation of protease-activated receptor (PAR)-1, PAR-2, and PAR-4 stimulates IL-6, IL-8, and prostaglandin E2 release from human respiratory epithelial cells. J Immunol. 2002;168(7):3577-85.

16. Vliagoftis H, Befus AD, Hollenberg MD, Moqbel R. Airway epithelial cells release eosinophil survival-promoting factors (GM-CSF) after stimulation of proteinase-activated receptor 2. J Allergy Clin Immunol. 2001;107(4):679-85.

17. Vliagoftis H, Schwingshackl A, Milne CD, Duszyk M, Hollenberg MD, Wallace JL, et al. Proteinase-activated receptor-2-mediated matrix metalloproteinase-9 release from airway epithelial cells. J Allergy Clin Immunol. 2000;106(3):537-45.

18. Winter MC, Shasby SS, Ries DR, Shasby DM. PAR2 activation interrupts E-cadherin adhesion and compromises the airway epithelial barrier: protective effect of beta-agonists. Am J Physiol Lung Cell Mol Physiol. 2006;291(4):L628-35.

19. Ferrell WR, Lockhart JC, Kelso EB, Dunning L, Plevin R, Meek SE, et al. Essential role for proteinase-activated receptor-2 in arthritis. J Clin Invest. 2003;111(1):35-41.

20. Crilly A, Burns E, Nickdel MB, Lockhart JC, Perry ME, Ferrell PW, et al. PAR(2) expression in peripheral blood monocytes of patients with rheumatoid arthritis. Ann Rheum Dis. 2012;71(6):1049-54.

21. Wei L, Yu X, Shi H, Zhang B, Lian M, Li J, et al. 15-PGDH/15-KETE plays a role in hypoxia-induced pulmonary vascular remodeling through ERK1/2-dependent PAR-2 pathway. Cell Signal. 2014;26(7):1476-88.

22. Allard B, Bara I, Gilbert G, Carvalho G, Trian T, Ozier A, et al. Protease activated receptor-2 expression and function in asthmatic bronchial smooth muscle. PLoS One. 2014;9(2):e86945.

23. Aubier M, Thabut G, Hamidi F, Guillou N, Brard J, Dombret MC, et al. Airway smooth muscle enlargement is associated with protease-activated receptor 2/ligand overexpression in patients with difficult-to-control severe asthma. J Allergy Clin Immunol. 2016;138(3):729-39 e11.

24. Knight DA, Lim S, Scaffidi AK, Roche N, Chung KF, Stewart GA, et al. Proteaseactivated receptors in human airways: upregulation of PAR-2 in respiratory epithelium from patients with asthma. J Allergy Clin Immunol. 2001;108(5):797-803.

25. Xiang Y, Masuko-Hongo K, Sekine T, Nakamura H, Yudoh K, Nishioka K, et al. Expression of proteinase-activated receptors (PAR)-2 in articular chondrocytes is modulated by IL-1beta, TNF-alpha and TGF-beta. Osteoarthritis Cartilage. 2006;14(11):1163-73.

26. Hamilton JR, Frauman AG, Cocks TM. Increased expression of protease-activated receptor-2 (PAR2) and PAR4 in human coronary artery by inflammatory stimuli unveils endothelium-dependent relaxations to PAR2 and PAR4 agonists. Circ Res. 2001;89(1):92-8.

27. Nystedt S, Ramakrishnan V, Sundelin J. The proteinase-activated receptor 2 is induced by inflammatory mediators in human endothelial cells. Comparison with the thrombin receptor. J Biol Chem. 1996;271(25):14910-5.

28. Svensson KJ, Kucharzewska P, Christianson HC, Skold S, Lofstedt T, Johansson MC, et al. Hypoxia triggers a proangiogenic pathway involving cancer cell microvesicles and PAR-2-mediated heparin-binding EGF signaling in endothelial cells. Proc Natl Acad Sci U S A. 2011;108(32):13147-52.

29. Hyun E, Ramachandran R, Hollenberg MD, Vergnolle N. Mechanisms behind the antiinflammatory actions of insulin. Crit Rev Immunol. 2011;31(4):307-40.

30. Husemoen LL, Glumer C, Lau C, Pisinger C, Morch LS, Linneberg A. Association of obesity and insulin resistance with asthma and aeroallergen sensitization. Allergy. 2008;63(5):575-82.

31. Dietze J, Bocking C, Heverhagen JT, Voelker MN, Renz H. Obesity lowers the threshold of allergic sensitization and augments airway eosinophilia in a mouse model of asthma. Allergy. 2012;67(12):1519-29.

32. Calixto MC, Lintomen L, Schenka A, Saad MJ, Zanesco A, Antunes E. Obesity enhances eosinophilic inflammation in a murine model of allergic asthma. Br J Pharmacol. 2010;159(3):617-25.

33. Loffredo LF, Abdala-Valencia H, Anekalla KR, Cuervo-Pardo L, Gottardi CJ, Berdnikovs S. Beyond epithelial-to-mesenchymal transition: Common suppression of differentiation programs underlies epithelial barrier dysfunction in mild, moderate, and severe asthma. Allergy. 2017.

34. Lim J, Iyer A, Liu L, Suen JY, Lohman RJ, Seow V, et al. Diet-induced obesity, adipose inflammation, and metabolic dysfunction correlating with PAR2 expression are attenuated by PAR2 antagonism. FASEB J. 2013;27(12):4757-67.

35. Badeanlou L, Furlan-Freguia C, Yang G, Ruf W, Samad F. Tissue factor-proteaseactivated receptor 2 signaling promotes diet-induced obesity and adipose inflammation. Nat Med. 2011;17(11):1490-7.

36. Li M, Yang X, Zhang Y, Chen L, Lu H, Li X, et al. Activation of proteaseactivated receptor2 is associated with increased expression of inflammatory factors in the adipose tissues of obese mice. Mol Med Rep. 2015;12(4):6227-34.

37. Danahay H, Atherton H, Jones G, Bridges RJ, Poll CT. Interleukin-13 induces a hypersecretory ion transport phenotype in human bronchial epithelial cells. Am J Physiol Lung Cell Mol Physiol. 2002;282(2):L226-36.

38. Lee HY. Molecular mechanisms of deguelin-induced apoptosis in transformed human bronchial epithelial cells. Biochem Pharmacol. 2004;68(6):1119-24.

39. Lee CS, Yi EH, Lee JK, Won C, Lee YJ, Shin MK, et al. Simvastatin suppresses RANTES-mediated neutrophilia in polyinosinic-polycytidylic acid-induced pneumonia. Eur Respir J. 2013;41(5):1147-56.

40. Seront E, Pinto A, Bouzin C, Bertrand L, Machiels JP, Feron O. PTEN deficiency is associated with reduced sensitivity to mTOR inhibitor in human bladder cancer through the unhampered feedback loop driving PI3K/Akt activation. Br J Cancer. 2013;109(6):1586-92.

41. Nagashima T, Shigematsu N, Maruki R, Urano Y, Tanaka H, Shimaya A, et al. Discovery of novel forkhead box O1 inhibitors for treating type 2 diabetes: improvement of fasting glycemia in diabetic db/db mice. Mol Pharmacol. 2010;78(5):961-70.

42. Lytton J, Westlin M, Hanley MR. Thapsigargin inhibits the sarcoplasmic or endoplasmic reticulum Ca-ATPase family of calcium pumps. J Biol Chem. 1991;266(26):17067-71.

43. Buteau J, Spatz ML, Accili D. Transcription factor FoxO1 mediates glucagon-like peptide-1 effects on pancreatic beta-cell mass. Diabetes. 2006;55(5):1190-6.

44. Lee BC, Lee J. Cellular and molecular players in adipose tissue inflammation in the development of obesity-induced insulin resistance. Biochim Biophys Acta. 2014;1842(3):446-62.

45. Reddel RR, Yang K, Rhim JS, Brash D, Su RT, Lechner JF, et al. Immortalized human bronchial epitherial mesothelial cell lines. Google Patents; 1989.

46. Jairaman A, Yamashita M, Schleimer RP, Prakriya M. Store-Operated Ca2+ Release-Activated Ca2+ Channels Regulate PAR2-Activated Ca2+ Signaling and Cytokine Production in Airway Epithelial Cells. J Immunol. 2015;195(5):2122-33.

47. Siddle K. Signalling by insulin and IGF receptors: supporting acts and new players. J Mol Endocrinol. 2011;47(1):R1-10.

48. Tzivion G, Dobson M, Ramakrishnan G. FoxO transcription factors; Regulation by AKT and 14-3-3 proteins. Biochim Biophys Acta. 2011;1813(11):1938-45.

49. Matsuzaki H, Daitoku H, Hatta M, Tanaka K, Fukamizu A. Insulin-induced phosphorylation of FKHR (Foxo1) targets to proteasomal degradation. Proc Natl Acad Sci U S A. 2003;100(20):11285-90.

 Adams MN, Pagel CN, Mackie EJ, Hooper JD. Evaluation of antibodies directed against human protease-activated receptor-2. Naunyn Schmiedebergs Arch Pharmacol. 2012;385(9):861-73.

51. Shrestha Palikhe N, Nahirney D, Laratta C, Gandhi VD, Vethanayagam D, Bhutani M, et al. Increased Protease-Activated Receptor-2 (PAR-2) Expression on CD14++CD16+ Peripheral Blood Monocytes of Patients with Severe Asthma. PLoS One. 2015;10(12):e0144500.

52. Bohm SK, Kong W, Bromme D, Smeekens SP, Anderson DC, Connolly A, et al. Molecular cloning, expression and potential functions of the human proteinase-activated receptor-2. Biochem J. 1996;314 (Pt 3):1009-16.

53. Compton SJ, Sandhu S, Wijesuriya SJ, Hollenberg MD. Glycosylation of human proteinase-activated receptor-2 (hPAR2): role in cell surface expression and signalling. Biochem J. 2002;368(Pt 2):495-505.

54. Arizmendi NG, Abel M, Mihara K, Davidson C, Polley D, Nadeem A, et al. Mucosal allergic sensitization to cockroach allergens is dependent on proteinase activity and proteinase-activated receptor-2 activation. J Immunol. 2011;186(5):3164-72.

55. Asaduzzaman M, Davidson C, Nahirney D, Fiteih Y, Puttagunta L, Vliagoftis H. Proteinase-activated receptor-2 blockade inhibits changes seen in a chronic murine asthma model. Allergy. 2017.

56. Miakotina OL, Goss KL, Snyder JM. Insulin utilizes the PI 3-kinase pathway to inhibit SP-A gene expression in lung epithelial cells. Respir Res. 2002;3:27.

57. Molina SA, Moriarty HK, Infield DT, Imhoff BR, Vance RJ, Kim AH, et al. Insulin signaling via the PI3-kinase/Akt pathway regulates airway glucose uptake and barrier function in a CFTR-dependent manner. Am J Physiol Lung Cell Mol Physiol. 2017;312(5):L688-L702.

58. Shoji S, Ertl RF, Linder J, Koizumi S, Duckworth WC, Rennard SI. Bronchial epithelial cells respond to insulin and insulin-like growth factor-I as a chemoattractant. Am J Respir Cell Mol Biol. 1990;2(6):553-7.

59. Sanghavi JN, Rabe KF, Kim JS, Magnussen H, Leff AR, White SR. Migration of human and guinea pig airway epithelial cells in response to calcitonin gene-related peptide. Am J Respir Cell Mol Biol. 1994;11(2):181-7.

60. Romberger DJ, Pladsen P, Claassen L, Yoshida M, Beckmann JD, Rennard SI. Insulin modulation of bronchial epithelial cell fibronectin in vitro. Am J Physiol. 1995;268(2 Pt 1):L230-8.

61. Dandona P, Aljada A, Mohanty P, Ghanim H, Hamouda W, Assian E, et al. Insulin inhibits intranuclear nuclear factor kappaB and stimulates IkappaB in mononuclear cells in obese subjects: evidence for an anti-inflammatory effect? J Clin Endocrinol Metab. 2001;86(7):3257-65.

62. Seiler F, Hellberg J, Lepper PM, Kamyschnikow A, Herr C, Bischoff M, et al. FOXO transcription factors regulate innate immune mechanisms in respiratory epithelial cells. J Immunol. 2013;190(4):1603-13.

63. Hyun E, Ramachandran R, Cenac N, Houle S, Rousset P, Saxena A, et al. Insulin modulates protease-activated receptor 2 signaling: implications for the innate immune response. J Immunol. 2010;184(5):2702-9.

64. McArdle MA, Finucane OM, Connaughton RM, McMorrow AM, Roche HM. Mechanisms of obesity-induced inflammation and insulin resistance: insights into the emerging role of nutritional strategies. Front Endocrinol (Lausanne). 2013;4:52.

65. Howarth PH, Babu KS, Arshad HS, Lau L, Buckley M, McConnell W, et al. Tumour necrosis factor (TNFalpha) as a novel therapeutic target in symptomatic corticosteroid dependent asthma. Thorax. 2005;60(12):1012-8.

66. Neveu WA, Allard JL, Raymond DM, Bourassa LM, Burns SM, Bunn JY, et al. Elevation of IL-6 in the allergic asthmatic airway is independent of inflammation but associates with loss of central airway function. Respir Res. 2010;11:28.

67. Tillie-Leblond I, Pugin J, Marquette CH, Lamblin C, Saulnier F, Brichet A, et al. Balance between proinflammatory cytokines and their inhibitors in bronchial lavage from patients with status asthmaticus. Am J Respir Crit Care Med. 1999;159(2):487-94.

68. Borish L, Mascali JJ, Dishuck J, Beam WR, Martin RJ, Rosenwasser LJ. Detection of alveolar macrophage-derived IL-1 beta in asthma. Inhibition with corticosteroids. J Immunol. 1992;149(9):3078-82.

69. Tiwari S, Halagappa VK, Riazi S, Hu X, Ecelbarger CA. Reduced expression of insulin receptors in the kidneys of insulin-resistant rats. J Am Soc Nephrol. 2007;18(10):2661-71.

70. Aguirre V, Werner ED, Giraud J, Lee YH, Shoelson SE, White MF. Phosphorylation of Ser307 in insulin receptor substrate-1 blocks interactions with the insulin receptor and inhibits insulin action. J Biol Chem. 2002;277(2):1531-7.

71. Werner ED, Lee J, Hansen L, Yuan M, Shoelson SE. Insulin resistance due to phosphorylation of insulin receptor substrate-1 at serine 302. J Biol Chem. 2004;279(34):35298-305.

72. De Fea K, Roth RA. Protein kinase C modulation of insulin receptor substrate-1 tyrosine phosphorylation requires serine 612. Biochemistry. 1997;36(42):12939-47.

73. Ozes ON, Akca H, Mayo LD, Gustin JA, Maehama T, Dixon JE, et al. A phosphatidylinositol 3-kinase/Akt/mTOR pathway mediates and PTEN antagonizes tumor necrosis factor inhibition of insulin signaling through insulin receptor substrate-1. Proc Natl Acad Sci U S A. 2001;98(8):4640-5.

# **2.9 Supplementary Figures**



Supplementary Fig. 2.E1 Insulin deficiency upregulates PAR-2 expression in AECs. (A-B) Cells were cultured in media with or without insulin and PAR-2 expression was studied, at indicated time points, by qRT-PCR assay developed by our laboratory (see methods). PAR-2 mRNA expression was increased at all the studied time points (n=3-4). (C) Cells were cultured in media with or without insulin for 24h and PAR-2 expression was studied by qRT-PCR. Taqman gene expression assay was used for *PAR-2* (Hs00608346\_m1, ThermoFisher Scientific) and housekeeping gene *Cyclophilin A (PPIA)* (Hs04194521\_s1, ThermoFisher Scientific). The normalization of PAR-2 expression to PPIA, a second housekeeping gene, confirmed that insulin deficiency results in PAR-2 upregulation (n=4). Data are presented as Mean  $\pm$  SEM; \**P*<0.05, \*\**P*<0.01, \*\*\**P*<0.001 between the groups indicated by horizontal lines.


Supplementary Fig. 2.E2 PAR-2 activation induces intracellular Ca<sup>++</sup> release from endoplasmic reticulum (ER) in AECs. AECs were cultured in 96 well plates (Corning Clear Bottom Black plates) and Ca<sup>++</sup> release assay was performed (see methods). (A) PAR-2 specific activation peptide (PAR-2 AP) induced Ca<sup>++</sup> release which was absent in PAR-2 control peptide (PAR-2 CP) treated cells. (B) Treatment of cells with Thapsigargin (TG) resulted in Ca<sup>++</sup> release from intracellular stores ER. Subsequent PAR-2 activation failed to induce Ca<sup>++</sup> release indicating PAR-2 activation results in Ca<sup>++</sup> release from ER.

# Chapter 3 Effect of insulin on proteinase-activated receptor-2 expression in the lungs *in vivo*

3.1 Aim: To study PAR-2 expression in the lungs of mice with lower levels of circulating insulin

#### **3.2 Introduction**

We showed that insulin dose-dependently downregulates PAR-2 expression in AECs cultured *in vitro* (Chapter 2). We also showed that mice with increased levels of circulating insulin *in vivo* express lower levels of PAR-2 in lungs further supporting the *in vitro* observation (Chapter 2). We now hypothesized that decreased levels of circulating insulin *in vivo* will lead to increased expression of PAR-2 in lungs. Usually the animal model used to study the effect of decreased insulin levels involves destruction of  $\beta$ -cells, using streptozotocin chemical, in order to prevent synthesis of any insulin. However, the effect of streptozotocin is not specific to  $\beta$ -cells (1). Moreover, destruction of  $\beta$ -cells results in hyperglycemia (2). To avoid these confounding factors, we decided to subject mice to overnight fasting as fasting decreases the levels of insulin.

#### 3.3 Methods

C57BL6 male mice (12-14 weeks old) were used. One group of mice were subjected to 14h overnight fasting, while the other group of mice were maintained on regular diet; water was made available to both the groups. Mice were euthanized with an overdose of sodium pentobarbital and blood was collected, in regular 1.5 mL Eppendorf tubes, by cardiac puncture. Blood was left at room temp for 20-30 min to clot and then spun at 1500g for 10 minutes at 4°C to separate serum; serum was stored at -80°C. Lungs were stored in RNAlater (ThermoFisher) at 4°C until used for qRT-PCR. Experiments were approved by the University of Alberta Health Sciences Laboratory Animal Ethics committee.

ELISA, to measure serum insulin levels, and qRT-PCR, to measure PAR-2 expression in lungs, were performed as described before (Chapter 2).

#### **3.4 Results and Discussion**

Overnight fasting resulted in decreased levels of circulating insulin (Fig. 3.1A); the levels of insulin were similar to those reported in the literature for the same age group of mice (3). PAR-2 expression was not increased in lungs of these animals with lower levels of insulin (Fig. 3.1B).



Fig. 3.1 Effect of overnight fasting on insulin levels in serum and PAR-2 expression in lungs. (A) Serum insulin levels and (B) PAR-2 mRNA expression levels in mice subjected to regular diet (Fed, n=4) or overnight fasting (Fasted, n=5). Data are presented as Mean  $\pm$  SEM; ns (not significant), \**P*<0.05, 2-tailed Mann-Whitney t Test.

Our *in vitro* data with primary human AECs show that absence of insulin for 24h results in around 1.8 fold upregulation in PAR-2 mRNA expression. Kinetics of PAR-2 regulation *in vivo*, which is more complex system than pure culture of AECs *in vitro*, could be different and thus 14h of fasting that represents gradual decrease in the levels of insulin may not provide enough time to modulate PAR-2 expression. Other possibility could be that the expression of receptor was modulated on AECs but the modulation of PAR-2 expression was not limited to only AECs. Thus assessing PAR-2 expression in the entire lung may have diluted the signal. To take care of this issue, PAR-2 expression can be studied in lung epithelial cells isolated using laser capture microdissection.

Overnight fasting in mice activates catabolic state and results in loss of body mass by 15% (4). Also overnight fasting may result in improved insulin sensitivity in mice in tissue-specific manner (5). Due to this reason, the experiments performed with overnight fasting are mainly to understand glucose utilization by different tissues (4). There are very limited studies looking at the expression of insulin-regulated genes after overnight fasting. In these experiments, human subjects (6, 7) or rodents (8) are first subjected to overnight fasting. After this, they are subjected to *in vivo* (in case of human subjects) or *ex vivo* (in case of rodents) insulin treatment and gene expression was compared in the same subject before and after insulin treatment.

A very limited research has focused on the effect of fasting on lungs. These studies have focused on the effect of prolong fasting (4 days) on the expression of glutathione, which provides protection against hyperoxic damage (9, 10) and have not focused on the insulin signaling. Fasting up to 48h is shown to have no effect on lung weight (11, 12). However, fasting resulted in increased expression of genes involved in proteolysis and autophagy pathways indicating the presence of stressors in the lungs (12). Hence it will not be advisable to use animal models subjected to prolong fasting to study the effect of lower levels of circulating insulin on PAR-2 expression in the lungs.

We also noticed a difference in the basal insulin levels in mice used for two different experiments. We used C57BL6 male mice for the following experiments:

- 1. High fat diet-induced obesity: One group of mice was maintained on high fat diet, while the other group was maintained on control diet (chow diet) (referenced as Exp-1)
- Overnight fasting to decrease the levels of circulating insulin: One group of mice was subjected to overnight fasting, while the other group was maintained on control diet (chow diet) (referenced as Exp-2)

The average circulating levels of insulin in control diet-fed mice from Exp-1 had 124 pg/mL, while control diet-fed mice from Exp-2 had 1352 pg/mL. The Exp-1 was performed in collaboration with Dr. Jason Dyck hence the experiment was performed in their lab with their mice, while Exp-2 was performed in our lab.

It is known that different batches of mice harbour some differences. Also the same strain of animal obtained from different suppliers is shown to have differences in microbiome (13) and

hematologic variables (14). We compared the mice used in the above mentioned two experiments:

- Age of the mice: At the end of the experiment, the age of the mice used in Exp-1 was around 9 months, while the age of the mice used in Exp-2 was around 3.5 months. However, in C57BL6 mice, insulin levels do not fluctuate as much between 3 to 9 months of age (15).
- 2. Composition of control diet: Different control diets could have different ingredients. The control diet used in Exp-1 was PicoLab Laboratory Rodent Diet Catalog#5L0D, while the control diet used in Exp-2 was PicoLab Rodent Diet 20 Catalog#5053. The calorie composition of these two control diets was very similar. However, the control diet used for Exp-1 contained more cholesterol and cholesterol negatively regulates insulin secretion (16). This could be one of the reasons that mice used in Exp-1 had lower insulin levels.
- 3. Mouse strain: The mouse strain used in the Exp-1 was C57BL6J (from Jackson lab), while the mouse strain used in Exp-2 was C57BL6N (from KnockOut Mouse Project KOMP, UC Davis). It is shown that C57BL6J mice (17) and C57BL6NJ mice (mixed genetic background) (3) have impaired insulin secretion compared to C57BL6N mice. This could be the primary reason that mice used in Exp-1 had lower insulin levels.

The mice used in Exp-2 were obtained as heterozygous PAR-2 knockout mice. The homozygous PAR-2 mice (wild-type) were obtained by mating heterozygous PAR-2 knockout male mice with heterozygous PAR-2 knockout female mice. The wild-type colony of mice was maintained for 7-9 generations before using it for Exp-2. PAR-2 is shown to inhibit insulin secretion (18). Thus it may be possible that due to lower expression of PAR-2, in PAR-2 heterozygous knockout mice, insulin secretion and thus insulin levels were increased. However, it is not clear if this increased insulin secretion was maintained after the generation of wild-type mice. Measuring insulin levels in PAR-2 knockout, PAR-2 heterozygous and wild-type mice may provide some clarification.

#### **3.5 References**

1. Imaeda A, Kaneko T, Aoki T, Kondo Y, Nagase H. DNA damage and the effect of antioxidants in streptozotocin-treated mice. Food Chem Toxicol. 2002;40(7):979-87.

2. Lenzen S. The mechanisms of alloxan- and streptozotocin-induced diabetes. Diabetologia. 2008;51(2):216-26.

3. Attane C, Peyot ML, Lussier R, Zhang D, Joly E, Madiraju SR, et al. Differential Insulin Secretion of High-Fat Diet-Fed C57BL/6NN and C57BL/6NJ Mice: Implications of Mixed Genetic Background in Metabolic Studies. PLoS One. 2016;11(7):e0159165.

4. Ayala JE, Bracy DP, McGuinness OP, Wasserman DH. Considerations in the design of hyperinsulinemic-euglycemic clamps in the conscious mouse. Diabetes. 2006;55(2):390-7.

5. Heijboer AC, Donga E, Voshol PJ, Dang ZC, Havekes LM, Romijn JA, et al. Sixteen hours of fasting differentially affects hepatic and muscle insulin sensitivity in mice. J Lipid Res. 2005;46(3):582-8.

6. Rome S, Clement K, Rabasa-Lhoret R, Loizon E, Poitou C, Barsh GS, et al. Microarray profiling of human skeletal muscle reveals that insulin regulates approximately 800 genes during a hyperinsulinemic clamp. J Biol Chem. 2003;278(20):18063-8.

7. Ducluzeau PH, Perretti N, Laville M, Andreelli F, Vega N, Riou JP, et al. Regulation by insulin of gene expression in human skeletal muscle and adipose tissue. Evidence for specific defects in type 2 diabetes. Diabetes. 2001;50(5):1134-42.

8. Zhang Y, Chen W, Li R, Li Y, Ge Y, Chen G. Insulin-regulated Srebp-1c and Pck1 mRNA expression in primary hepatocytes from zucker fatty but not lean rats is affected by feeding conditions. PLoS One. 2011;6(6):e21342.

9. Smith LJ, Horcher P, Anderson J, Shamsuddin M. Effect of fasting on the lung glutathione redox cycle in air- and oxygen-exposed mice: beneficial effects of sugar. J Lab Clin Med. 1990;116(5):717-23.

10. Smith LJ, Anderson J, Shamsuddin M, Hsueh W. Effect of fasting on hyperoxic lung injury in mice. The role of glutathione. Am Rev Respir Dis. 1990;141(1):141-9.

11. Jenniskens FA, Jopperi-Davis KS, Walters LC, Schorr EN, Rogers LK, Welty SE, et al. Effects of fasting on tissue contents of coenzyme A and related intermediates in rats. Pediatr Res. 2002;52(3):437-42.

12. Yamamoto J, Kamata S, Miura A, Nagata T, Kainuma R, Ishii I. Differential adaptive responses to 1- or 2-day fasting in various mouse tissues revealed by quantitative PCR analysis. FEBS Open Bio. 2015;5:357-68.

13. Rodrigue L, Lavoie MC. Comparison of the proportions of oral bacterial species in BALB/c mice from different suppliers. Lab Anim. 1996;30(2):108-13.

14. Kampfmann I, Bauer N, Johannes S, Moritz A. Differences in hematologic variables in rats of the same strain but different origin. Vet Clin Pathol. 2012;41(2):228-34.

15. Teng R, Gavrilova O, Suzuki N, Chanturiya T, Schimel D, Hugendubler L, et al. Disrupted erythropoietin signalling promotes obesity and alters hypothalamus proopiomelanocortin production. Nat Commun. 2011;2:520.

16. Hao M, Head WS, Gunawardana SC, Hasty AH, Piston DW. Direct effect of cholesterol on insulin secretion: a novel mechanism for pancreatic beta-cell dysfunction. Diabetes. 2007;56(9):2328-38.

17. Fergusson G, Ethier M, Guevremont M, Chretien C, Attane C, Joly E, et al. Defective insulin secretory response to intravenous glucose in C57Bl/6J compared to C57Bl/6N mice. Mol Metab. 2014;3(9):848-54.

18. Regard JB, Kataoka H, Cano DA, Camerer E, Yin L, Zheng YW, et al. Probing cell typespecific functions of Gi in vivo identifies GPCR regulators of insulin secretion. J Clin Invest. 2007;117(12):4034-43.

### Chapter 4 Effect of insulin on proteinase-activated receptor-2 expression in primary human airway smooth muscle cells

**4.1 Aim:** To study the effect of insulin on PAR-2 expression in human airway smooth muscle cells

These experiments were done in collaboration with Dr. Andrew Halayko at the University of Manitoba. I designed the experiments with them. Dr. Halayko's lab performed the experiments and sent us samples and I evaluated PAR-2 expression.

#### 4.2 Introduction

Airway SMCs play an important role in the development of airway remodeling and airway hyperresponsiveness (AHR), which are the characteristics of asthma. One of the factors that contribute to airway remodeling in asthma is increase in airway smooth muscle thickness (1), which could be due to increase in SMC proliferation (2) that may result in narrowing of the airways. Also, airway smooth muscle in asthma has increased contractile property (1), which increases AHR meaning that increased sensitivity of the airways to agonists inducing contraction. PAR-2 is expressed on airway smooth muscle cells (SMCs) (3). PAR-2 activation on airway SMCs is shown to increase smooth muscle proliferation (4, 5) and contraction (6). In asthma, PAR-2 expression is increased in airway SMCs (5) and is associated with airway smooth muscle thickness (7). Thus, in asthma, PAR-2 airway SMCs may contribute to airway remodeling, by inducing SMC proliferation, and to AHR, by making smooth muscle hyperresponsiveness to PAR-2 agonists. Insulin is shown to induce profibrotic and procontractile phenotype in airway SMCs by upregulating collagen release and proteins involved in contraction (8, 9). Since insulin develops procontractile phenotype it could be possible that insulin upregulates PAR-2 expression in airway SMCs, which further increased the smooth muscle sensitivity to contraction by PAR-2 agonists. Also we showed that insulin negatively regulates PAR-2 expression in AECs (Chapter 2) and we wanted to assess if insulin has similar or opposite effect in ASM for the regulation of PAR-2 expression.

#### 4.3 Methods

#### Cell culture

Airway SMCs, from 3 donors, were isolated as described before (10). Briefly, airway SMCs were isolated from 2nd to 4th generation macroscopically healthy bronchial main stem surgically removed from patients undergoing lung resection surgery for adenocarcinoma. Cells were cultured in DMEM (Lonza) media containing 10% FBS (PAA Laboratories) and penicillin-streptomycin (Lonza) until they reached 80% confluency. Cells were then cultured in FBS-free DMEM media containing penicillin-streptomycin with or without 1X ITS (Insulin 10  $\mu$ g/mL, Transferrin 10  $\mu$ g/mL and Selenium 0.01  $\mu$ g/mL, Lonza) for up to 48h. PAR-2 mRNA expression was studied as described before (Chapter 2). *GAPDH* mRNA was used as an internal control. Data were calculated and represented as *PAR-2* copies /10000 *GAPDH* copies.

#### **Statistics**

Results are expressed as Mean  $\pm$  SEM. Results are analyzed using repeated measures ANOVA with Tukey's multiple comparison test. The *p*-value of less than 0.05 was considered statistically significant.

#### 4.4 Results and Discussion

The number of PCR cycles required to detect PAR-2 mRNA in airway SMCs was higher compared to the number of cycles required in AECs (36 cycles in airway SMCs vs. 29 cycles in AECS) suggesting that PAR-2 expression in airway SMCs was low compared to AECs. However, there were differences in the protocols used to study PAR-2 expression in airway SMCs and AECs, which may explain the observed difference in PAR-2 expression between the two cell types. First, Trizol was used to extract RNA from airway SMCs, while RNeasy mini kit was used for AECs. Even though there is no literature comparing these two techniques for the quality of RNA obtained from airway SMCs or AECs, it is shown that these two methods could provide different qualities of RNA in other cell types (11, 12). The quality of RNA is important as it could affect the cDNA synthesis (reverse transcription) and ultimately gene expression results obtained using qPCR (13). We will not be able to compare the quality of RNA as we do not have that data available to us. Also, for airway SMCs 150 ng RNA was used for reverse transcription using a mixture of random hexamer and oligo dT as primers, while 500 ng RNA was used for reverse transcription. The concentration of RNA used for reverse transcription plays a role in the quality of cDNA synthesized (14), which could also affect the analysis of gene expression. Thus further experiments are required where airway SMCs and AECs are processed by similar protocol to compare the PAR-2 expression between the two cell types.

Following 24h and 48h of culture in the presence of ITS, there was a trend of decreased PAR-2 expression compared to cells cultured without ITS (Fig. 4.1).



Fig. 4.1 Effect of insulin on PAR-2 expression in airway SMCs. Airway SMCs were cultured in FBS-free DMEM media containing 1X ITS for up to 48h and qRT-PCR for PAR-2 expression was performed. Data are presented as Mean  $\pm$  SEM; ns (not significant) between the groups indicated by horizontal lines.

ITS mixture contains insulin, transferrin and selenium. If insulin is responsible for the observed trend of decreased PAR-2 expression, then similar to AECs, insulin negatively regulates PAR-2 expression in airway SMCs. Thus insulin-dependent development of procontractile phenotype in airway SMCs may be because of increased expression of proteins involved in smooth muscle contraction (8, 9) and not due to PAR-2 upregulation. Transferrin is used in cell culture media as it provides iron to cells and also prevents toxic effect of excess oxygen radicals. Transferrin is also present in the media we use to culture AECs, where exclusion of transferrin does not affect PAR-2 expression (Chapter 5). Selenium acts as an anti-oxidant in cell culture system. Serum deprivation and growth factor deprivation induce oxidative stress in different cells (15-17) to upregulate PAR-2 expression (18). Hence it is possible that FBS deprivation in our experiments with SMCs may results in oxidative stress and PAR-2 upregulation and the presence of transferrin and selenium prevented PAR-2 upregulation by preventing oxidative stress. More experiments are required to determine if the observed trend of decreased PAR-2 expression in the presence of ITS is due to the presence of insulin or the other components of ITS.

#### 4.5 References

1. Seow CY, Schellenberg RR, Pare PD. Structural and functional changes in the airway smooth muscle of asthmatic subjects. Am J Respir Crit Care Med. 1998;158(5 Pt 3):S179-86.

2. Johnson PR, Roth M, Tamm M, Hughes M, Ge Q, King G, et al. Airway smooth muscle cell proliferation is increased in asthma. Am J Respir Crit Care Med. 2001;164(3):474-7.

3. Berger P, Tunon-De-Lara JM, Savineau JP, Marthan R. Selected contribution: tryptaseinduced PAR-2-mediated Ca(2+) signaling in human airway smooth muscle cells. J Appl Physiol (1985). 2001;91(2):995-1003.

4. Berger P, Perng DW, Thabrew H, Compton SJ, Cairns JA, McEuen AR, et al. Tryptase and agonists of PAR-2 induce the proliferation of human airway smooth muscle cells. J Appl Physiol (1985). 2001;91(3):1372-9.

5. Allard B, Bara I, Gilbert G, Carvalho G, Trian T, Ozier A, et al. Protease activated receptor-2 expression and function in asthmatic bronchial smooth muscle. PLoS One. 2014;9(2):e86945.

6. Schmidlin F, Amadesi S, Vidil R, Trevisani M, Martinet N, Caughey G, et al. Expression and function of proteinase-activated receptor 2 in human bronchial smooth muscle. Am J Respir Crit Care Med. 2001;164(7):1276-81.

7. Aubier M, Thabut G, Hamidi F, Guillou N, Brard J, Dombret MC, et al. Airway smooth muscle enlargement is associated with protease-activated receptor 2/ligand overexpression in patients with difficult-to-control severe asthma. J Allergy Clin Immunol. 2016;138(3):729-39 e11.

8. Schaafsma D, McNeill KD, Stelmack GL, Gosens R, Baarsma HA, Dekkers BG, et al. Insulin increases the expression of contractile phenotypic markers in airway smooth muscle. Am J Physiol Cell Physiol. 2007;293(1):C429-39.

9. Singh S, Bodas M, Bhatraju NK, Pattnaik B, Gheware A, Parameswaran PK, et al. Hyperinsulinemia adversely affects lung structure and function. Am J Physiol Lung Cell Mol Physiol. 2016;310(9):L837-45.

10. Anaparti V, Pascoe CD, Jha A, Mahood TH, Ilarraza R, Unruh H, et al. Tumor necrosis factor regulates NMDA receptor-mediated airway smooth muscle contractile function and airway responsiveness. Am J Physiol Lung Cell Mol Physiol. 2016;311(2):L467-80.

11. Deng MY, Wang H, Ward GB, Beckham TR, McKenna TS. Comparison of six RNA extraction methods for the detection of classical swine fever virus by real-time and conventional reverse transcription-PCR. J Vet Diagn Invest. 2005;17(6):574-8.

12. Tavares L, Alves PM, Ferreira RB, Santos CN. Comparison of different methods for DNA-free RNA isolation from SK-N-MC neuroblastoma. BMC Res Notes. 2011;4:3.

13. Vermeulen J, De Preter K, Lefever S, Nuytens J, De Vloed F, Derveaux S, et al. Measurable impact of RNA quality on gene expression results from quantitative PCR. Nucleic Acids Res. 2011;39(9):e63.

14. Williams WV, Rosenbaum H, Weiner DB. Effect of RNA concentration on cDNA synthesis for DNA amplification. PCR Methods Appl. 1992;2(1):86-8.

15. Pandey S, Lopez C, Jammu A. Oxidative stress and activation of proteasome protease during serum deprivation-induced apoptosis in rat hepatoma cells; inhibition of cell death by melatonin. Apoptosis. 2003;8(5):497-508.

16. Tammariello SP, Quinn MT, Estus S. NADPH oxidase contributes directly to oxidative stress and apoptosis in nerve growth factor-deprived sympathetic neurons. J Neurosci. 2000;20(1):RC53.

17. Lieberthal W, Triaca V, Koh JS, Pagano PJ, Levine JS. Role of superoxide in apoptosis induced by growth factor withdrawal. Am J Physiol. 1998;275(5 Pt 2):F691-702.

18. Aman M, Hirano M, Kanaide H, Hirano K. Upregulation of proteinase-activated receptor-2 and increased response to trypsin in endothelial cells after exposure to oxidative stress in rat aortas. J Vasc Res. 2010;47(6):494-506.

### Chapter 5: Growth media supplements regulate proteinaseactivated receptor-2 expression in human bronchial airway epithelial cells

Vivek D. Gandhi, Nami Shrestha Palikhe, Drew Nahirney, A. Dean Befus and Harissios Vliagoftis

This is a version of a manuscript that will be submitted after performing a few more experiments.

#### 5.1 Abstract

**Background:** Proteinase-Activated Receptor-2 (PAR-2) is involved in the pathogenesis of a variety of inflammatory diseases including allergic asthma. Also PAR-2 expression is increased under inflammatory conditions. In asthma, PAR-2 expression is increased on airway epithelium but mechanism of PAR-2 regulation is poorly studied. Since PAR-2 expression is increased under inflammatory conditions, we hypothesize that inflammatory microenvironment, such as decreased availability of cell growth nutrients, regulates PAR-2 expression.

**Objectives:** To study the effect of decreased cell growth nutrients on PAR-2 expression and functions in primary bronchial epithelial cells (BECs).

**Methods:** PAR-2 expression and functions were studied in submerged and air-liquid interface cultures of primary BECs from healthy and asthmatic individuals subjected to media supplement deprivation, as a model of decreased cell growth nutrients. Cells were also treated with different inhibitors to elucidate the mechanism underlying PAR-2 expression.

**Results:** Media supplement deprivation upregulated PAR-2 mRNA and protein expression and also PAR-2-dependent pro-inflammatory potential of BECs from healthy individuals. The upregulation in PAR-2 mRNA was due to increased PAR-2 gene transcription. Also this upregulation was reversible upon re-addition of media supplements. The media supplements involved in PAR-2 regulation were insulin, hydrocortisone, bovine pituitary extract and retinoic acid. Treatment of cells with different inhibitors of signaling molecules showed that media supplement deprivation-induced PAR-2 upregulation is independent of PI3K-Akt-FOXO-1 and ERK1/2 pathways. Finally, BECs from asthmatic individuals expressed increased levels of PAR-

2 compared to BECs from healthy individuals and showed similar media supplement deprivation-mediated regulation of PAR-2 expression and functions, as observed in BECs from healthy individuals.

**Conclusions:** Our results indicate that decrease in cell growth nutrients in the inflammatory microenvironment may result in increased PAR-2 expression in the airways. Since activation of upregulated PAR-2 by proteinases from inhaled aeroallergens or immune cells may augment PAR-2-mediated airway inflammation, targeting PAR-2 regulation could be an effective therapeutic strategy.

#### **5.2 Introduction**

Proteinase-Activated Receptor-2 (PAR-2), a G-coupled pro-inflammatory receptor, is expressed on cells of different tissues and activated by serine proteinases (1). PAR-2 activation is shown to play a role in the pathogenesis of inflammatory diseases in different organs (2-5). In lungs, PAR-2 plays a role in the pathogenesis of allergic asthma (6, 7), idiopathic pulmonary fibrosis (IPF) (8) and pulmonary arterial hypertension (9).

In lungs, PAR-2 is expressed on bronchial smooth muscle cells (10, 11), fibroblasts (12) and bronchial epithelial cells (BECs) (11). BECs are the first line of defense against the inhaled air constituents including allergens, pathogens and pollutants. PAR-2 activation on BECs results in impaired epithelial barrier function (13) and also in the release of mediators (14-18) which are involved in the development of the cardinal features of asthma such as allergic sensitization, allergic inflammation and airway remodeling.

In addition to PAR-2 involvement in the pathogenesis of inflammatory diseases, PAR-2 expression is also increased in these diseases (3, 8, 19-21). In asthma, PAR-2 expression is increased on airway epithelial cells (AECs) (22). By using transgenic mice overexpressing PAR-2, it was shown that increased levels of PAR-2 augments allergic airway inflammation (6). Thus, understanding regulation of PAR-2 expression may provide the knowledge required to inhibit PAR-2 upregulation and control PAR-2-mediated inflammation.

The regulation of PAR-2 expression is poorly understood. Since PAR-2 expression is increased in inflammatory diseases, it is deduced that inflammatory environment plays a role in the regulation of PAR-2 expression. Due to increased number of active inflammatory cells that release mediators and reactive oxygen species, the site of inflammation is characterized by the presence of increased levels of inflammatory mediators and oxidative stress (23). Also higher metabolic demands, consumption of oxygen and cell growth nutrients by increased number of cells may result in decreased availability of metabolic substrates (24), hypoxia (25, 26) and decreased levels of nutrients supporting cell growth at the site of inflammation. In asthma, airways are infiltrated by inflammatory cells such as eosinophils (27) and neutrophils (28). In addition of these inflammatory cells, other cells such as monocytes, macrophages, epithelial cells and smooth muscle cells, release chemokines, cytokines and reactive oxygen species (ROS) resulting in the presence of high levels of inflammatory mediators (29) and oxidative stress (30) in the airways of asthma patients. It is also shown that airway remodeling of the subepithelium, which is also observed in asthma, may limit the delivery of cell growth nutrients and oxygen (hypoxia) to the airway epithelium (31-33). There is evidence that inflammatory mediators (34-36), hypoxia (9, 37) and growth factor deprivation (38) regulate PAR-2 expression on cells other than AECs.

Asthmatic airways are chronically inflamed and may be under the influence of above mentioned stress stimuli. We hypothesised that decreased availability of nutrients supporting cell growth, in the background of inflammation, regulates PAR-2 expression on BECs. The cell culture media for BECs contains supplements that support cell growth. To study the effect of decreased availability of factors supporting cell growth on PAR-2 expression, we cultured BECs in media deprived of supplements (media supplement deprivation). Media supplement deprivation upregulated PAR-2 expression in primary cultures of normal BECs (NBECs) and also increased PAR-2-dependent pro-inflammatory potential of BECs. The media supplement deprivation-induced increase in PAR-2 expression was independent of PI3K-Akt-FOXO-1 and ERK1/2 signaling pathways. Finally, primary BECs from asthmatics (ABECs) also showed similar media supplement-mediated PAR-2 regulation.

#### 5.3 Materials and Methods

#### Cell culture

Primary normal BECs (NBECs) from five donors were purchased (normal human bronchial epithelial cells, Lonza, Walkersville, MD); all the experiments were performed on cells at passage 3-5 after receiving. Cells were cultured as described in Chapter 2 of this thesis.

Primary asthmatic BECs (ABECs) were cultured from material obtained from bronchial brushings, from five asthmatic individuals undergoing bronchoscopy for clinical or research indications at the University of Alberta Hospital. All patients gave informed consent before bronchoscopy. The study was approved by the Ethics Committee, University of Alberta. ABECs were cultured under the same conditions with NBECs.

In addition to the ABECs obtained at the University of Alberta hospital, mRNA from airway epithelial cells obtained from five healthy and four asthmatic individuals were received from our collaborator Dr. Jamila Chakir, Laval University to compare PAR-2 mRNA expression in primary healthy vs. asthmatic BECs. These airway epithelial cells were isolated from bronchial biopsy specimens as described before (39).

Air-liquid interphase (ALI) cultures of NBECs and ABECs were performed as described in this reference (40) and in Chapter 2 of this thesis.

The bronchial epithelial cell line BEAS-2B was purchased from ATCC (Manassas, VA) and cultured as described in Chapter 2 of this thesis.

The airway epithelial cell lines A549 and Calu-3 cells were purchased from ATCC and cultured using DMEM media (low glucose, Life Technologies) supplemented with 10% heat inactivated fetal bovine serum (FBS, Life Technologies) and penicillin-streptomycin.

NBECs were treated with phosphoinositide 3-kinase (PI3K) inhibitors LY294002 (20  $\mu$ M, (41, 42), Tocris, Minneapolis, MN), wortmannin (5  $\mu$ M, (43), Cayman, Ann-Arbor, MI), Forkhead box protein O1 (FOXO-1) inhibitor AS1842856 (1  $\mu$ M, (44), Calbiochem, San Diego, CA) and MEK 1/2 inhibitor U0126 (10  $\mu$ M, (45, 46), Tocris). All the inhibitors were dissolved in DMSO

(Sigma-Aldrich). Final concentration of DMSO in the treatment was 0.1% or less. DMSO at 0.1% concentration had no effect on PAR-2 expression (appendix-1).

For all the cell culture procedure, cells were cultured at  $37^{\circ}$ C in 5% CO<sub>2</sub> and 90% humidity and media was changed 3 times /week. All cell cultures maintained in our lab were tested monthly for mycoplasma using the MycoAlert kit (Lonza) and only negative cultures were used.

#### RNA extraction, reverse transcription and quantitative RT-PCR (qRT-PCR)

RNA extraction and reverse transcription were performed as described in Chapter 2 of this thesis.

qRT-PCR for *PAR-2* was performed using a probe-based assay developed in our lab (47) and as described in Chapter 2 of this thesis. *Glyceraldehyde 3-phosphate dehydrogenase (GAPDH)* was used as a house keeping gene. The results were expressed as *PAR-2* copies/1000 *GAPDH* copies. PAR-2 expression was also normalized to a second housekeeping gene, *Cyclophilin A*. For this Taqman gene expression assays were used for *PAR-2* (Hs00608346\_m1, ThermoFisher Scientific) and housekeeping gene *Cyclophilin A* (Hs04194521\_s1, ThermoFisher Scientific).

#### Intracellular calcium release assay

PAR-2-mediated increase in intracellular calcium was measured as described in Chapter 2 of this thesis. PAR-2-mediated calcium release from ER was expressed as % of thapsigargin (TG)-induced calcium release from ER.

#### Preparations of PAR-2 reporter plasmid

The protocol on the development of this construct is described in Chapter 2 of this thesis.

#### Transfection studies

For promoter studies, BEAS-2B and Calu-3 cells were seeded in 12-well plates (Falcon) and cultured until 85% confluent. On the day of transfection, BEAS-2B and Calu-3 cells were rinsed with PBS and 400  $\mu$ L of respective media without supplements (BEBM or DMEM without FBS) was added to the wells. The transfection complexes were formed by mixing 1  $\mu$ g PAR-2 reporter plasmid DNA (preparation of reporter plasmid is described in the supplementary methods for

Chapter-2) or empty pGL3-Basic vector, 5 ng Renilla plasmid DNA and 2  $\mu$ g of Lipofectamine® 2000 Transfection Reagent (Life Technologies) in 100  $\mu$ L respective media lacking supplements and incubating the mixture for 30 min at room temperature. Transfection complexes (100  $\mu$ L) were then added to the cells dropwise and cells were incubated for 5h. After 5h of incubation, the transfection complexes were removed and appropriate growth media was added for 24h.

Next day, to study the effect of media supplements on PAR-2 promoter activity, cells were cultured in respective media with or without supplements for 24h. The promoter activity was assessed using the Dual-Luciferase Reporter Assay Kit (Promega, Madison, WI).

#### Luciferase reporter assay

At the end of 24h of culturing cells in media with or without supplements, cells were rinsed with PBS and harvested using a cell scraper. Cells were then pelleted and lysed with 60  $\mu$ L of passive lysis buffer (Promega, Madison, WI). From this, 50  $\mu$ L lysate was assayed for luciferase activity using Dual –Luciferase Reporter Assay Kit (Promega) as described in Chapter 2 of this thesis.

To study the effect of growth supplement deprivation on PAR-2 promoter activity, results were expressed as fold change in promoter activity over cells cultured with growth supplements.

#### Lactate dehydrogenase (LDH) assay

Cells, plated in multi-well plate, were cultured in media with or without supplements for up to 48h. Cell supernatants were collected and assayed for LDH activity using LDH Cytotoxicity Assay kit (Cayman). One well, of the multi-well plate, was treated with 1% Triton X-100 for 30 min to lyse the cells and obtain maximum LDH activity. To measure LDH activity, 100  $\mu$ L of LDH standards or samples were mixed with 100  $\mu$ L of Reaction Solution and incubated for 30 min at room temperature; the absorbance was read at 490 nm. Since Triton X-100 will lyse all the cells, the LDH activity in Triton X-100 treated cells was considered as 100% and "% LDH activity" in cells cultured with or without media supplements was calculated relative to the LDH activity obtained in Triton X-100 treated cells. The LDH activity is proportional to the amount of LDH present in cell supernatant.

#### Apoptosis and cell cycle evaluation using flow cytometry

Apoptosis assay was performed using the Dead Cell Apoptosis Kit (Life Technologies). Cells, plated in multi-well plate, were cultured in media with or without supplements for 24h. Cells were then trypsinized and pelleted down by centrifugation at 4° C at 200 X g for 5min. The pellet was washed by resuspending it in cold PBS and again centrifuged at 4° C at 200 X g for 5min. The pellet was resuspended in 1X annexin-binding buffer at the ratio of 100  $\mu$ l buffer per 100,000 cells. Annexin V (component-A, 5  $\mu$ l/100  $\mu$ l cell suspension) and propidium iodide (component-B, 1  $\mu$ l of 100  $\mu$ g/mL per 100  $\mu$ l cell suspension) were added and cells were incubated for 15 min at room temperature in the dark. At the end of incubation, 400  $\mu$ l of annexin-binding buffer was added to the samples, mixed gently and kept on ice. Stained cells were examined on BD FACSCanto II flow cytometer and the data were analyzed using FlowJo (Tree Star, Ashland, OR).

Cell cycle analysis was performed using the following protocol. Cells, plated in multi-well plate, were cultured in media with or without supplements for 24h. Cells were then trypsinized and pelleted down by centrifugation at 4° C at 200 X g for 5min. The pellet was washed by resuspending it in cold PBS and again centrifuged at 4° C at 200 X g for 5min. Cell pellet was then resuspended in 500  $\mu$ l of cold PBS. Cold 3.5 mL of 70% ethanol was added to cells drop wise, mixed gently and kept on ice for 2h. Cells were then centrifuged at 4° C at 200 X g for 5min and resuspended in 5 mL cold PBS. Cells were kept on ice for 2-3 min before spinning them down again at 4° C at 200 X g for 5min. The pellet was resuspended in 500  $\mu$ l of staining solution comprising of 200  $\mu$ g/mL ribonuclease A (Sigma-Aldrich) and 20  $\mu$ g/mL propidium iodide (Sigma-Aldrich) in 0.1% Triton X-100-PBS and incubated for 30 min at room temperature in dark. Stained cells were examined on BD FACSCanto II flow cytometer.

#### ELISA

CXCL8 levels in cell supernatants were measured using the Human IL-8/CXCL8 DuoSet ELISA kit (R&D Systems, Minneapolis, MN) as per manufacturer's instructions.

#### Trans epithelial electrical resistance (TEER) measurements

TEER was measured on differentiated ALI cultures of ABECs and NBECs using Millicell<sup>®</sup> ERS-2 Electrical Resistance System (Sigma-Aldrich). TEER measurements were first taken before wound was created (Pre-cut). Mechanical wound was then created by scratching the insert membrane using the tip of a Pasteur pipette. Post-cut TEER was measured immediately after wounding and also at 24h and 48h after wounding. Pre-cut TEER values were considered as 100% and Post-cut TEER values were expressed as % of Pre-cut TEER value.

#### **Statistics**

Results are expressed as Mean  $\pm$  SEM. In case of comparisons between two groups, two-tailed paired *t*-test was used, except for the comparison between asthmatics and healthy individuals where two-tailed Mann Whitney test was used. In case of three or more groups, repeated measures ANOVA was used with Tukey's multiple comparison test or 2-way repeated measures ANOVA was used with Bonferroni post-test, depending on the experimental set-up. The *p*-value of less than 0.05 was considered statistically significant.

#### 5.4 Results

#### Media supplement deprivation regulates PAR-2 mRNA expression in NBECs

In submerged NBEC cultures, media supplement deprivation induced an over 2-fold upregulation of PAR-2 mRNA expression at 24h (Fig. 5.1A and 5.1B) and 48h (Fig. 5.1B) without affecting cell viability as measured by LDH assay (Fig. 5.1C and 5.1D). Media supplement deprivation-induced PAR-2 upregulation was confirmed using another housekeeping gene cyclophilin A (Fig. 5.1E). The addition of supplements back to supplement-deprived cells completely reversed PAR-2 upregulation after 6h and 24h (Fig. 5.1F), indicating the dynamic nature of the PAR-2 expression, while continuing the supplement deprivation further increased PAR-2 mRNA expression (Fig. 5.1F). ALI cultures of differentiated NBECs also showed supplement deprivation-induced upregulation in PAR-2 mRNA expression (Fig. 5.1G).



Fig. 5.1 Media supplements regulate PAR-2 expression in submerged and differentiated cultures of NBECs. PAR-2 expression was studied by qRT-PCR. (A-B) Cells were cultured in media with supplements until they reached 85% confluency (Time "0"). Cells were then cultured in media without or with supplements for (A) 6h and 24h (n=5), and (B) 24h and 48h (n=3). (C-D) LDH activity in supernatants of cells cultured in media without or with supplements for (C) 24h (n=4) and (D) 48h (n=3). LDH activity in Triton X-100 treated cells was considered as 100% and "% LDH activity" in cells cultured without or with media supplements was calculated relative to the LDH activity obtained in Triton X-100 treated cells. (E) NBECs were cultured in media without or with supplements for 24h (n=6). Taqman gene expression assays were used for *PAR-2* 

and housekeeping gene *Cyclophilin A*. (F) Cells were cultured in media with supplements until reached 85% confluency (Time "0"). Cells were then cultured in media without or with supplements for 24h (left part of the graph). Cells cultured in media without supplements for 24h were further cultured in media without or with supplements for the next 6h and 24h (n=3) (right part of the graph). (G) Differentiated ALI cultures of NBECs were cultured in media without or with supplements for up to 72h (n=3). Data are presented as Mean  $\pm$  SEM; ns (not significant), \**P*<0.05, \*\**P*<0.01, \*\*\**P*<0.001 at an indicated time-point between two groups or between the groups indicated by horizontal lines.

To study the role of individual media supplements on PAR-2 expression, we excluded one supplement at a time. Only insulin exclusion, as we have shown before (Chapter 2), showed increase in PAR-2 expression; hydrocortisone exclusion showed a trend towards increase in PAR-2 expression (p=0.08), while the exclusion of other supplements did not affect PAR-2 expression (Fig. 5.2A). However, insulin deprivation-induced PAR-2 upregulation was significantly lower than the upregulation induced by the absence of all media supplements.

Next, we cultured cells in the absence of media supplements for 24h to upregulate PAR-2 expression and then added selected supplements back individually, mainly hormones, growth factors and ligands for nuclear receptors hydrocortisone and RA. Insulin (Fig. 5.2B), BPE (bovine pituitary extract) (Fig. 5.2C), hydrocortisone (Fig. 5.2D) and RA (retinoic acid) (Fig. 5.2D) decreased the media supplement deprivation-induced upregulation in PAR-2 mRNA expression, while the addition of epinephrine (Fig. 5.2E) and EGF (epidermal growth factor) (Fig. 5.2F) did not induce significant alteration in PAR-2 mRNA expression. These results indicate that the increased upregulation in PAR-2 expression upon exclusion of all the media supplements compared to exclusion of only insulin is a result of combined absence of other media supplements regulating PAR-2 expression.



Fig. 5.2 Effect of different media supplements on PAR-2 expression in NBECs. PAR-2 expression was studied by qRT-PCR. (A) Cells were cultured in media with supplements until they reached 85% confluency (Time "0"). Cells were then cultured in media without all supplements or with all supplements or without indicated supplement for 24h (n=3). (B-F) Cells were first cultured in media without supplements for 24h. Cells were then further cultured in media without supplements or with only indicated supplement (B) insulin (n=13) (C) BPE (n=3) (D) hydrocortisone and RA (n=4) (E) epinephrine (n=8) and (F) EGF (n=3) for the next 24h. Data are presented as Mean  $\pm$  SEM; ns (not significant), \**P*<0.05, \*\**P*<0.01, \*\*\**P*<0.001 between the groups indicated by horizontal lines.

# Role of PI3K-FOXO-1 pathway in media supplement deprivation-induced PAR-2 upregulation

We have shown that insulin, through PI3K pathway, controls FOXO-1 activity to regulate PAR-2 expression. Since, insulin is one of the media supplements that modulates PAR-2 expression,

we evaluated the role of PI3K signaling in PAR-2 expression in supplement-deprived and insulin-deprived cells. PI3K inhibition further increased PAR-2 expression in growth supplement-deprived cells, while PI3K inhibition in insulin-deprived cells had no effect on PAR-2 expression (Fig. 5.3A). We have specifically shown that FOXO-1 inhibition prevents insulin deprivation-induced PAR-2 upregulation (Chapter 2), while FOXO-1 inhibition did not prevent PAR-2 upregulation in supplement-deprived cells (Fig. 5.3B). These results collectively indicate that the mechanism of PAR-2 regulation is different in supplement-deprived cells vs. insulin-deprived cells.



Fig. 5.3 Role of PI3K-FOXO-1 pathway in media supplement deprivation-induced PAR-2 upregulation. PAR-2 expression was studied by qRT-PCR. (A) Cells were cultured in media without all supplements or without only insulin in the presence or absence of PI3K inhibitor LY294002 for 24h (n=3). (B) Cells were cultured in media without supplements or with supplements in the presence or absence of FOXO-1 inhibitor AS1842856 for 24h (n=3). Data are presented as Mean  $\pm$  SEM; ns (not significant), \**P*<0.05, \*\**P*<0.01, between the groups indicated by horizontal lines.

#### Role of ERK1/2 in media supplement deprivation-induced PAR-2 upregulation

Media supplement deprivation activates ERK1/2 (48). Insulin and hydrocortisone, that decreased media supplement deprivation-induced PAR-2 upregulation in our BECs, are also shown to inhibit ERK1/2 (49, 50). We hence studied the role of ERK1/2 in media supplement deprivation-induced PAR-2 upregulation in BECs. Inhibition of ERK1/2 decreased media supplement deprivation- and insulin deprivation-induced PAR-2 upregulation (Fig. 5.4A). However,

inhibition of ERK1/2 also decreased PAR-2 expression in cells cultured with media supplements (Fig. 5.4A). The % decrease in PAR-2 expression upon ERK1/2 inhibition was similar in all 3 treatment groups namely; media supplement deprivation, insulin deprivation and cells cultured with media supplements indicating role of ERK1/2 in basal expression of PAR-2 (Fig. 5.4B).



Fig. 5.4 Role of ERK1/2 in media supplement deprivation-induced PAR-2 upregulation. Cells were cultured in media without all supplements, without only insulin or with all supplements in the presence or absence of ERK1/2 inhibitor U0126 for 24h. (A) PAR-2 expression was studied by qRT-PCR (n=3). (B) % decrease in PAR-2 expression was calculated in cells treated with ERK1/2 inhibitor in different media compared to respective untreated cells (n=3). Data are presented as Mean  $\pm$  SEM; ns (not significant), \**P*<0.05, \*\**P*<0.01, \*\*\**P*<0.001 between the groups indicated by horizontal lines.

## Media supplement deprivation-induced PAR-2 upregulation is due to increased PAR-2 gene transcription

We have shown that insulin deprivation did not alter PAR-2 mRNA stability and resulted in FOXO-1-mediated transcriptional regulation of PAR-2 (Chapter 2). Since media supplement deprivation-induced PAR-2 mRNA upregulation does not involve FOXO-1, it is possible that this upregulation is due to increased PAR-2 mRNA stability. After blocking mRNA transcription using actinomycin D, we found that the rate of PAR-2 mRNA degradation was similar whether cells were cultured without or with supplements indicating media supplement deprivation did not alter PAR-2 mRNA stability (Fig. 5.5A).

To study the effect of media supplement deprivation on PAR-2 gene transcription we used a PAR-2 reporter plasmid. For these experiments, we used immortalized airway epithelial cell lines (AECs) because primary AECs have lower transfection efficiency. We first studied if media supplement deprivation regulates PAR-2 expression in these cells. Media supplement deprivation induced PAR-2 mRNA upregulation in BEAS-2B cells (Fig. 5.5B) as observed in primary NBECs. Other cell lines, Calu-3 and A549, which were cultured with FBS as the source of supplements did not show alteration in PAR-2 expression when cultured in FBS-deprived media (Fig. 5.5C and 5.5D).

Media supplement deprivation in BEAS-2B cells transfected with a PAR-2 reporter plasmid showed increase in promoter activity indicating supplement deprivation-induced increase in PAR-2 mRNA is due to increased transcription of PAR-2 gene (Fig. 5.5E). Serum deprivation in Calu-3 cells transfected with a PAR-2 reporter plasmid did not increase PAR-2 promoter activity (Fig. 5.5F); which was expected as serum deprivation did not increase PAR-2 mRNA expression in these cells (Fig. 5.5C).



Fig. 5.5 Media supplement deprivation transcriptionally regulates PAR-2 mRNA expression in NBECs. PAR-2 expression was studied by qRT-PCR. (A) Cells were cultured in media with supplements until they reached 85% confluency. Cells were then cultured in media without or with supplements for 24h (Time "0"). Cells were further cultured in media without or with supplements, in the presence of actinomycin D, and PAR-2 mRNA expression was studied at indicated time points (n=5). PAR-2 levels were expressed as % PAR-2 expression compared to the PAR-2 expression at time "0". The numbers on the lines show the rate of PAR-2 mRNA degradation for 8h after the addition of actinomycin D. (B-D) Cells were cultured in media with supplements or FBS until reached 80% confluency (Time "0"). (B) BEAS-2B (n=3), (C) Calu-3 (n=3) and (D) A549 cells (n=3) were then cultured in media without or with respective supplements for 24h. (E-F) PAR-2 promoter activity expressed as fold change in luciferase induction in (E) BEAS-2B cells (n=9) and (F) Calu-3 cells (n=3) transfected with PAR-2 reporter plasmid and cultured in media without or with respective supplements for 24h. Data are presented as Mean  $\pm$  SEM; ns (not significant), \*\**P*<0.01, \*\*\**P*<0.01 between the groups indicated by horizontal lines.

#### Media supplement deprivation upregulates PAR-2 protein expression and PAR-2dependent pro-inflammatory potential of BECs

Media supplement deprivation increased PAR-2 protein expression in BECs after 24h, as shown by measuring fluorescence intensity of PAR-2, for 30 cells/ experiment in 6 independent experiments, using confocal microscopy (Fig. 5.6A-C). To study the functional significance of this PAR-2 upregulation, we studied PAR-2-mediated calcium signaling and CXCL8 release in cells cultured in media with or without supplements for 24h.

We showed that PAR-2 activation increases intracellular  $Ca^{++}$  by inducing  $Ca^{++}$  release from ER (Chapter 2). Also upregulated PAR-2 expression leads to increased PAR-2-mediated intracellular  $Ca^{++}$  release (51). We also found that PAR-2-activating peptide (PAR-2 AP)-mediated increase in intracellular  $Ca^{++}$  was higher in cells subjected to media supplement deprivation compared to cells cultured in media with supplements (Fig. 5.6D and 5.6E).

Further, PAR-2 AP-mediated calcium signaling is shown to induce the release of inflammatory mediators including CXCL8 (52). The baseline release of CXCL8 was higher in cells cultured with media supplements compared to cells cultured without media supplements. The treatment of cells with PAR-2 control peptide (PAR-2 CP) had no effect on baseline CXCL8 release irrespective of the presence or absence of media supplements. PAR-2 activation induced 1.2 fold higher CXCL8 release over PAR-2 CP in cells cultured with media supplements, while PAR-2 activation induced 1.8 fold higher CXCL8 release over PAR-2 CP in cells cultured with supplements (Fig. 5.6F).



Fig. 5.6 Media supplements deprivation upregulates functional PAR-2 expression. (A-B) Images of PAR-2 expression (Green) in (A) a representative individual cell and (B) a representative field of view displaying multiple cells cultured in media without (left image) or with (right image) supplements for 24h. Nuclei are stained with DAPI (Blue). (C) Corrected Total Cell Fluorescence representing PAR-2 expression in cells cultured in media without or with supplements. CTCF was calculated for 30 cells/experiment in 6 independent experiments (n=6). (D) Representative graph of PAR-2-mediated intracellular Ca<sup>++</sup> release from ER in cells cultured in media without or with supplements for 24h. Cells were activated using PAR-2 AP (50  $\mu$ M). (E) PAR-2-mediated Ca<sup>++</sup> release from ER was quantified as % of thapsigargin-induced Ca<sup>++</sup> release from ER (n=4). (F) Cells were cultured in media without or with supplements for 24h and then treated with PAR-2 AP or PAR-2 CP (50  $\mu$ M) for 12h in the same media that they were cultured in. CXCL8 levels (pg/mL) in cell culture supernatant were measured using ELISA (n=7). The graph shows PAR-2 AP-mediated CXCL8 release over PAR-2 CP. Data are presented as Mean ± SEM; \**P*<0.05.

#### Effect of media supplement deprivation on cell cycle and apoptosis

Media supplement deprivation leads to cell cycle arrest in "G01" phase and activation of apoptosis (53). The cell cycle analysis showed that media supplement deprivation increased cells in G01 phase by only 6% compared to cells with supplements (Fig. 5.7A). We also found that media supplement deprivation marginally increased apoptotic cells (Fig. 5.7B) but not dead cells (Fig. 5.7C). These results indicate that after 24h of media supplement deprivation, the growth of BECs was not significantly altered.



Fig. 5.7 Effect of media supplement deprivation on cell cycle and apoptosis in NBECs. Cells were cultured in media without or with supplements for 24h and subjected to flow cytometry. (A) % cells in different phases of cell cycle (n=3) based on the propidium iodide-stained DNA profile analysis, (B) % apoptotic cells (annexin V-positive and annexin V/propidium iodide double positive cells were considered as apoptotic cells) (n=2) (C) % dead cells (only propidium iodide positive cells were considered as dead cells). Data are presented as Mean  $\pm$  SEM for (A) and as Mean for (B) and (C); \*\**P*<0.01

## BECs from asthma patients express increased PAR-2 compared to NBECs and show similar media supplement-mediated PAR-2 regulation as NBECs

BECs obtained from asthmatic individuals, when cultured *in vitro*, retain some of their *in vivo* characteristics such as defective barrier property (54), poor differentiation (55) and poor efficiency at wound healing (56, 57). We first studied if our ABECs show the previously demonstrated characteristic of inefficient wound healing. Mechanical wound resulted in decrease in TEER in both NBECs and ABECs as expected. In NBECs, TEER was increased to over 100%

after 24h of injury indicating that the wound was completely healed. Further the TEER was maintained around 100% up to 48h (Fig. 5.8A). Contrary to this, ABECs were not able to reach 100% TEER even after 48h of injury showing defective repair property of ABECs (Fig. 5.8B).

PAR-2 expression is increased in *ex vivo* cultures of ABECs compared to NBECs (Fig. 5.8C). We next studied if ABECs obtained from four donors respond to media supplement deprivation similar to NBECs. Media supplement deprivation induced upregulation in PAR-2 expression in all four cultures of ABECs (Fig. 5.8D), as observed in NBECs. The addition of media supplements, to supplement-deprived cells, reversed PAR-2 upregulation at 6h (Fig. 5.8E) and 24h (Fig. 5.8E). The ALI cultures of ABECs also showed increased PAR-2 expression upon media supplement deprivation (Fig. 5.8F and 5.8G). Finally, similar to NBECs, PAR-2 activation in media supplement deprived ABECs released increased CXCL8 compared to cells cultured with media supplements (Fig. 5.8H).



Fig. 5.8 BECs from asthmatic individuals express increased PAR-2 mRNA and show similar media supplement-mediated PAR-2 regulation. (A-C) BECs were obtained from asthmatic and healthy individuals and cultured *in vitro*. (A-B) Differentiated ALI cultures of BECs from (A) healthy (NBEC) and (B) asthmatic (ABEC) individuals were subjected to TEER measurements before (Pre-cut) and after (Post-cut) wound generation. The Pre-cut TEER values were considered as 100% and Post-cut TEER values were expressed as % of Pre-cut TEER value. (C) NBECs and ABECs were cultured in media with supplements PAR-2 expression was studied by qRT-PCR. (D-E) PAR-2 expression was studied by qRT-PCR. Cells were cultured in media with supplements until reached 85% confluency (Time "0"). (D) ABECs were then cultured in media without or with supplements for 24h (n=7). (E) ABECs were then cultured in media without or
with supplements for 24h (left part of the graph). Cells cultured in media without supplements for 24h were further cultured in media without or with supplements for the next 6h and 24h (n=6) (right part of the graph). (F-G) PAR-2 expression was studied by qRT-PCR. Differentiated ALI cultures of ABECs were cultured in media without or with supplements for (F) 24h (n=3) or (G) 48h and 72h (n=1). (H) Cells were cultured in media without or with supplements for 24h and then treated with PAR-2 AP or PAR-2 CP (50  $\mu$ M) for 12h in the same media that they were cultured in. CXCL8 levels (pg/mL) in cell culture supernatant were measured using ELISA (n=4). The graph shows PAR-2 AP-mediated CXCL8 release over PAR-2 CP. Data are presented as Mean  $\pm$  SEM; ns (not significant), \**P*<0.05, \*\**P*<0.01, \*\*\**P*<0.001 between the groups indicated by horizontal lines.

#### **5.5 Discussion**

The expression of PAR-2, a G-coupled receptor, is increased in inflammatory conditions in many organ systems (3, 8, 19-21). PAR-2 upregulation can further augment inflammation (6). In asthma, PAR-2 expression is increased on AECs (22), but the mechanisms underlying regulation of PAR-2 expression is poorly studied. The presence of increased number of metabolically active inflammatory cells may result in decreased levels of cell growth nutrients at the site of inflammation (24) and it was shown that growth factor deprivation can regulate PAR-2 expression in cells other than AECs (38). We hypothesized that the localized deprivation of growth nutrients, in the background of inflammation, upregulates PAR-2 expression in AECs. To mimic the decreased availability of factors supporting cell growth *in vivo*, we excluded cell growth supplements from cell culture media *in vitro*. We observed that media supplement deprivation resulted in PAR-2 upregulation in undifferentiated submerged and differentiated ALI cultures of BECs obtained from healthy individuals. PAR-2 upregulation also increased PAR-2-dependent pro-inflammatory potential of the epithelial cells as observed by increased calcium signaling and CXCL8 release upon PAR-2 activation.

To understand the exclusion of which supplement resulted in increased PAR-2 expression, we excluded one supplement at a time. As demonstrated (Chapter 2), exclusion of insulin resulted in increased PAR-2 expression on NBECs, while exclusion of other media supplements individually did not upregulate PAR-2 expression. We also used another approach to identify the supplements involved in the regulation of PAR-2 expression. We first cultured cells in the media lacking supplements to upregulate PAR-2 expression and then added individual supplement. Insulin had the maximum effect in decreasing PAR-2 expression while hydrocortisone, BPE and RA were able to decrease PAR-2 expression partially. Inhaled corticosteroids are the most effective treatment to control airway inflammation in asthma hence, downregulation of PAR-2 expression by hydrocortisone could be an anti-inflammatory mechanism of corticosteroids. BPE is a non-defined source of growth factors and hormones; the observation again supports the notion that decreased levels of cell growth supplement at the site of inflammation may result in increased PAR-2 expression. RA is shown to be important for the differentiation and secretory function of BECs (58). RA is a metabolite of vitamin A; vitamin A is an anti-oxidant vitamin, which is found to be decreased in asthmatic kids (59) and is a suggested dietary supplement to

control oxidative stress (60). Thus the observation suggest that the stress stimuli present in the lung microenvironment such as decreased cell growth nutrients and increased oxidative stress can modulate PAR-2 expression.

Media supplement deprivation upregulated PAR-2 expression in primary BECs and virally transformed bronchial epithelial cell line BEAS-2B but not in cancerous epithelial cell lines Calu-3 (61) and alveolar epithelial cells, A549 (62). The observed difference could be because of the cancer phenotype of Calu-3 and A549 cell line, as cancer cell lines may behave differently than primary cells and transformed cell lines. Also, specialized cell culture media was used for BECs and BEAS-2B, where the media supplements are defined, while FBS, an undefined mixture of supplements, was used to culture Calu-3 and A549 cells. We also tried growing primary BECs in media containing FBS to study if FBS deprivation affects PAR-2 expression in primary BECs but experiments were terminated as cell morphology was altered.

Media supplement deprivation activates ERK1/2 (48). Insulin and hydrocortisone that decreased PAR-2 upregulation in our BECs are shown to inhibit ERK1/2 (49, 50). Hence it is possible that media supplement deprivation in BECs activated ERK1/2 to upregulate PAR-2 expression. Inhibition of ERK1/2 decreased media supplement deprivation- and insulin deprivation-induced PAR-2 upregulation but it also decreased, at the same extent, basal PAR-2 expression in cells cultured with media supplements. These results suggest that ERK1/2 is not involved in media supplement deprivation- and insulin deprivation.

We also studied the involvement of PI3K in media supplement deprivation-induced PAR-2 regulation as we have shown that insulin regulates PAR-2 expression by inhibiting FOXO-1 transcription factor through PI3K-Akt pathway (Chapter 2). PAR-2 expression was not affected upon PI3K inhibition in insulin deprived cells, while in media supplement deprived cells, PAR-2 expression was increased upon PI3K inhibition. We have shown that insulin deprivation-induced PAR-2 upregulation is FOXO-1 dependent (Chapter 2). In current study inhibition of FOXO-1 did not prevent the supplement deprivation-induced PAR-2 upregulation indicating the involvement of different mechanisms in PAR-2 upregulation induced by supplement deprivation vs. insulin deprivation.

It is shown that during growth factor deprivation / serum deprivation, FOXO-1 gets activated and localizes to nucleus to induce apoptosis (63) but if FOXO-1 is present in cytoplasm (inactive FOXO-1) then autophagy is induced. It is also shown that growth factor deprivation activates stress-activated protein kinase (SAPK) c-Jun N-terminal kinase (JNK), which inhibits FOXO-1-induced apoptosis by decreasing FOXO-1 expression (64) and induces autophagy (65). Autophagy is a cell survival mechanism that usually occurs before apoptosis but prolonged autophagy results in apoptosis (66). Since we did not see increase in apoptosis in our cells, it is possible that FOXO-1 was not activated in our media supplement-deprived cells, hence was not involved in media supplement deprivation-induced PAR-2 upregulation. This can be evaluated by studying (i) if growth factor deprivation results in less nuclear FOXO-1 localization compared to insulin deprivation (ii) if media supplement deprivation-induced JNK activation decreases FOXO-1 expression.

There is other evidence that support the notion that the mechanisms involved in PAR-2 upregulation, induced by media supplement deprivation vs. insulin deprivation, are different. First, supplements other than insulin also had an effect on PAR-2 expression and thus, compared to removal of only insulin, removal of all the supplements may result in activation or repression of different signaling pathways involved in PAR-2 regulation. Second, the time course for PAR-2 upregulation is different between the cells subjected to insulin deprivation vs. media supplement deprivation. In our previous study, we did not find increase in PAR-2 expression after 6h of insulin deprivation (Chapter 2), but in this study PAR-2 expression was increased after 6h of supplement deprivation. Since exclusion of all the supplements could exert more cellular stress than the exclusion of only insulin, this early upregulation in PAR-2 expression in cells subjected to supplement deprivation may be the result of early cellular responses such as activation of stressed-induced signaling pathways involving stress kinases (67).

As discussed, we observed a marginal increase in apoptotic cells under the growth supplement deprivation condition. It is possible that autophagy, which precedes apoptosis, was activated in our culture system against the cellular stress that could result upon supplement deprivation. PAR-2 upregulation at this pre-apoptotic phase could be a protective mechanism for cell survival as PAR-2 activation is shown to inhibit apoptosis in airway (68) and colonic (69) epithelial cells.

Further experiments are required to assess if autophagy is activated upon supplement deprivation and if autophagy plays any role in supplement deprivation-induced PAR-2 upregulation.

We obtained ABECs through bronchial brushes and cultured them ex vivo. The baseline expression of PAR-2 was higher in ABECs compared to NBECs. The upregulated PAR-2 expression was maintained in ex vivo cultures of ABECs, which suggests an involvement of epigenetic changes in PAR-2 upregulation. Since media supplement deprivation increased PAR-2 expression in NBECs, we then studied the effect of media supplement deprivation on PAR-2 expression in ABECs. We hypothesized that if upregulated PAR-2 expression in ABECs is due to the epigenetic changes induced by decreased cell growth nutrients, then ABECs will not further respond to media supplement deprivation for PAR-2 expression. However, ABECs showed media supplement deprivation-induced PAR-2 upregulation. Further addition of media supplements to supplement-deprived ABECs decreased PAR-2 expression and brought it down close to baseline PAR-2 expression of ABECs, which is higher than baseline PAR-2 expression of NBECs. These results support the notion that increased PAR-2 expression in ABECs is imprinted into the cells due to epigenetic changes, however the factor responsible these epigenetic changes are not known. It is possible that other stress stimuli present in the in vivo inflammatory lung microenvironment may play a role in the development of these epigenetic changes.

In conclusion, our results suggest that decrease in cell growth nutrients, which could occur locally in the background of inflammation, may be a possible mechanism involved in PAR-2 upregulation observed in inflammatory conditions. Further studies are required to understand if the pre-apoptotic pathways such as autophagy or stress-activated protein kinases are activated and involved in this limiting cell growth nutrient condition-induced PAR-2 upregulation. Also the results suggest that regulation of PAR-2 expression could be one of the anti-inflammatory mechanisms of hydrocortisone. Understanding the mechanisms of PAR-2 upregulation may provide us new therapeutic strategies to inhibit PAR-2 upregulation and thus PAR-2-mediated inflammation in different inflammatory conditions.

#### **5.6 References**

1. Cottrell GS, Amadesi S, Schmidlin F, Bunnett N. Protease-activated receptor 2: activation, signalling and function. Biochem Soc Trans. 2003;31(Pt 6):1191-7.

2. Lohman RJ, Cotterell AJ, Suen J, Liu L, Do AT, Vesey DA, et al. Antagonism of protease-activated receptor 2 protects against experimental colitis. J Pharmacol Exp Ther. 2012;340(2):256-65.

3. Steinhoff M, Corvera CU, Thoma MS, Kong W, McAlpine BE, Caughey GH, et al. Proteinase-activated receptor-2 in human skin: tissue distribution and activation of keratinocytes by mast cell tryptase. Exp Dermatol. 1999;8(4):282-94.

4. Knight V, Tchongue J, Lourensz D, Tipping P, Sievert W. Protease-activated receptor 2 promotes experimental liver fibrosis in mice and activates human hepatic stellate cells. Hepatology. 2012;55(3):879-87.

5. Xue M, Chan YK, Shen K, Dervish S, March L, Sambrook PN, et al. Protease-activated receptor 2, rather than protease-activated receptor 1, contributes to the aggressive properties of synovial fibroblasts in rheumatoid arthritis. Arthritis Rheum. 2012;64(1):88-98.

6. Schmidlin F, Amadesi S, Dabbagh K, Lewis DE, Knott P, Bunnett NW, et al. Proteaseactivated receptor 2 mediates eosinophil infiltration and hyperreactivity in allergic inflammation of the airway. J Immunol. 2002;169(9):5315-21.

7. Asaduzzaman M, Nadeem A, Arizmendi N, Davidson C, Nichols HL, Abel M, et al. Functional inhibition of PAR2 alleviates allergen-induced airway hyperresponsiveness and inflammation. Clin Exp Allergy. 2015;45(12):1844-55.

8. Wygrecka M, Kwapiszewska G, Jablonska E, von Gerlach S, Henneke I, Zakrzewicz D, et al. Role of protease-activated receptor-2 in idiopathic pulmonary fibrosis. Am J Respir Crit Care Med. 2011;183(12):1703-14.

9. Kwapiszewska G, Markart P, Dahal BK, Kojonazarov B, Marsh LM, Schermuly RT, et al. PAR-2 inhibition reverses experimental pulmonary hypertension. Circ Res. 2012;110(9):1179-91.

10. Schmidlin F, Amadesi S, Vidil R, Trevisani M, Martinet N, Caughey G, et al. Expression and function of proteinase-activated receptor 2 in human bronchial smooth muscle. Am J Respir Crit Care Med. 2001;164(7):1276-81.

11. Miotto D, Hollenberg MD, Bunnett NW, Papi A, Braccioni F, Boschetto P, et al. Expression of protease activated receptor-2 (PAR-2) in central airways of smokers and non-smokers. Thorax. 2002;57(2):146-51.

12. Matsushima R, Takahashi A, Nakaya Y, Maezawa H, Miki M, Nakamura Y, et al. Human airway trypsin-like protease stimulates human bronchial fibroblast proliferation in a protease-activated receptor-2-dependent pathway. Am J Physiol Lung Cell Mol Physiol. 2006;290(2):L385-95.

13. Winter MC, Shasby SS, Ries DR, Shasby DM. PAR2 activation interrupts E-cadherin adhesion and compromises the airway epithelial barrier: protective effect of beta-agonists. Am J Physiol Lung Cell Mol Physiol. 2006;291(4):L628-35.

14. Kouzaki H, O'Grady SM, Lawrence CB, Kita H. Proteases induce production of thymic stromal lymphopoietin by airway epithelial cells through protease-activated receptor-2. J Immunol. 2009;183(2):1427-34.

15. Kouzaki H, Tojima I, Kita H, Shimizu T. Transcription of interleukin-25 and extracellular release of the protein is regulated by allergen proteases in airway epithelial cells. Am J Respir Cell Mol Biol. 2013;49(5):741-50.

16. Asokananthan N, Graham PT, Fink J, Knight DA, Bakker AJ, McWilliam AS, et al. Activation of protease-activated receptor (PAR)-1, PAR-2, and PAR-4 stimulates IL-6, IL-8, and prostaglandin E2 release from human respiratory epithelial cells. J Immunol. 2002;168(7):3577-85.

17. Vliagoftis H, Befus AD, Hollenberg MD, Moqbel R. Airway epithelial cells release eosinophil survival-promoting factors (GM-CSF) after stimulation of proteinase-activated receptor 2. J Allergy Clin Immunol. 2001;107(4):679-85.

18. Vliagoftis H, Schwingshackl A, Milne CD, Duszyk M, Hollenberg MD, Wallace JL, et al. Proteinase-activated receptor-2-mediated matrix metalloproteinase-9 release from airway epithelial cells. J Allergy Clin Immunol. 2000;106(3):537-45.

19. Ferrell WR, Lockhart JC, Kelso EB, Dunning L, Plevin R, Meek SE, et al. Essential role for proteinase-activated receptor-2 in arthritis. J Clin Invest. 2003;111(1):35-41.

20. Crilly A, Burns E, Nickdel MB, Lockhart JC, Perry ME, Ferrell PW, et al. PAR(2) expression in peripheral blood monocytes of patients with rheumatoid arthritis. Ann Rheum Dis. 2012;71(6):1049-54.

21. Wei L, Yu X, Shi H, Zhang B, Lian M, Li J, et al. 15-PGDH/15-KETE plays a role in hypoxia-induced pulmonary vascular remodeling through ERK1/2-dependent PAR-2 pathway. Cell Signal. 2014;26(7):1476-88.

22. Knight DA, Lim S, Scaffidi AK, Roche N, Chung KF, Stewart GA, et al. Proteaseactivated receptors in human airways: upregulation of PAR-2 in respiratory epithelium from patients with asthma. J Allergy Clin Immunol. 2001;108(5):797-803.

23. Biswas SK. Does the Interdependence between Oxidative Stress and Inflammation Explain the Antioxidant Paradox? Oxid Med Cell Longev. 2016;2016:5698931.

24. Eltzschig HK, Carmeliet P. Hypoxia and inflammation. N Engl J Med. 2011;364(7):656-65.

25. Baay-Guzman GJ, Bebenek IG, Zeidler M, Hernandez-Pando R, Vega MI, Garcia-Zepeda EA, et al. HIF-1 expression is associated with CCL2 chemokine expression in airway inflammatory cells: implications in allergic airway inflammation. Respir Res. 2012;13:60.

26. Huerta-Yepez S, Baay-Guzman GJ, Bebenek IG, Hernandez-Pando R, Vega MI, Chi L, et al. Hypoxia inducible factor promotes murine allergic airway inflammation and is increased in asthma and rhinitis. Allergy. 2011;66(7):909-18.

27. Possa SS, Leick EA, Prado CM, Martins MA, Tiberio IF. Eosinophilic inflammation in allergic asthma. Front Pharmacol. 2013;4:46.

28. Monteseirin J. Neutrophils and asthma. J Investig Allergol Clin Immunol. 2009;19(5):340-54.

29. Chung KF, Barnes PJ. Cytokines in asthma. Thorax. 1999;54(9):825-57.

30. Sahiner UM, Birben E, Erzurum S, Sackesen C, Kalayci O. Oxidative stress in asthma. World Allergy Organ J. 2011;4(10):151-8.

31. Polosukhin VV, Lawson WE, Milstone AP, Egunova SM, Kulipanov AG, Tchuvakin SG, et al. Association of progressive structural changes in the bronchial epithelium with subepithelial fibrous remodeling: a potential role for hypoxia. Virchows Arch. 2007;451(4):793-803.

32. Polosukhin VV, Cates JM, Lawson WE, Milstone AP, Matafonov AG, Massion PP, et al. Hypoxia-inducible factor-1 signalling promotes goblet cell hyperplasia in airway epithelium. J Pathol. 2011;224(2):203-11. 33. Lee SY, Kwon S, Kim KH, Moon HS, Song JS, Park SH, et al. Expression of vascular endothelial growth factor and hypoxia-inducible factor in the airway of asthmatic patients. Ann Allergy Asthma Immunol. 2006;97(6):794-9.

34. Xiang Y, Masuko-Hongo K, Sekine T, Nakamura H, Yudoh K, Nishioka K, et al. Expression of proteinase-activated receptors (PAR)-2 in articular chondrocytes is modulated by IL-1beta, TNF-alpha and TGF-beta. Osteoarthritis Cartilage. 2006;14(11):1163-73.

35. Hamilton JR, Frauman AG, Cocks TM. Increased expression of protease-activated receptor-2 (PAR2) and PAR4 in human coronary artery by inflammatory stimuli unveils endothelium-dependent relaxations to PAR2 and PAR4 agonists. Circ Res. 2001;89(1):92-8.

36. Nystedt S, Ramakrishnan V, Sundelin J. The proteinase-activated receptor 2 is induced by inflammatory mediators in human endothelial cells. Comparison with the thrombin receptor. J Biol Chem. 1996;271(25):14910-5.

37. Svensson KJ, Kucharzewska P, Christianson HC, Skold S, Lofstedt T, Johansson MC, et al. Hypoxia triggers a proangiogenic pathway involving cancer cell microvesicles and PAR-2-mediated heparin-binding EGF signaling in endothelial cells. Proc Natl Acad Sci U S A. 2011;108(32):13147-52.

38. Davydova ON, Yakovlev AA, Lyzhin AA, Khaspekov LG, Gulyaeva NV. Growth Factors Deprivation Induces a Specific Increase in PAR2 Receptor mRNA Expression in Primary Cerebellar Cultures. Neurochem J+. 2010;4(4):279-83.

39. Goulet F, Boulet LP, Chakir J, Tremblay N, Dube J, Laviolette M, et al. Morphologic and functional properties of bronchial cells isolated from normal and asthmatic subjects. Am J Respir Cell Mol Biol. 1996;15(3):312-8.

40. Danahay H, Atherton H, Jones G, Bridges RJ, Poll CT. Interleukin-13 induces a hypersecretory ion transport phenotype in human bronchial epithelial cells. Am J Physiol Lung Cell Mol Physiol. 2002;282(2):L226-36.

41. Lee HY. Molecular mechanisms of deguelin-induced apoptosis in transformed human bronchial epithelial cells. Biochem Pharmacol. 2004;68(6):1119-24.

42. Lee CS, Yi EH, Lee JK, Won C, Lee YJ, Shin MK, et al. Simvastatin suppresses RANTES-mediated neutrophilia in polyinosinic-polycytidylic acid-induced pneumonia. Eur Respir J. 2013;41(5):1147-56.

43. Seront E, Pinto A, Bouzin C, Bertrand L, Machiels JP, Feron O. PTEN deficiency is associated with reduced sensitivity to mTOR inhibitor in human bladder cancer through the unhampered feedback loop driving PI3K/Akt activation. Br J Cancer. 2013;109(6):1586-92.

44. Nagashima T, Shigematsu N, Maruki R, Urano Y, Tanaka H, Shimaya A, et al. Discovery of novel forkhead box O1 inhibitors for treating type 2 diabetes: improvement of fasting glycemia in diabetic db/db mice. Mol Pharmacol. 2010;78(5):961-70.

45. Abdullah LH, Davis CW. Regulation of airway goblet cell mucin secretion by tyrosine phosphorylation signaling pathways. Am J Physiol Lung Cell Mol Physiol. 2007;293(3):L591-9.

46. Wesley UV, Bove PF, Hristova M, McCarthy S, van der Vliet A. Airway epithelial cell migration and wound repair by ATP-mediated activation of dual oxidase 1. J Biol Chem. 2007;282(5):3213-20.

47. Shrestha Palikhe N, Nahirney D, Laratta C, Gandhi VD, Vethanayagam D, Bhutani M, et al. Increased Protease-Activated Receptor-2 (PAR-2) Expression on CD14++CD16+ Peripheral Blood Monocytes of Patients with Severe Asthma. PLoS One. 2015;10(12):e0144500.

48. Sinha D, Bannergee S, Schwartz JH, Lieberthal W, Levine JS. Inhibition of ligandindependent ERK1/2 activity in kidney proximal tubular cells deprived of soluble survival factors up-regulates Akt and prevents apoptosis. J Biol Chem. 2004;279(12):10962-72.

49. van der Heide LP, Hoekman MF, Biessels GJ, Gispen WH. Insulin inhibits extracellular regulated kinase 1/2 phosphorylation in a phosphatidylinositol 3-kinase (PI3) kinase-dependent manner in Neuro2a cells. J Neurochem. 2003;86(1):86-91.

50. Kassel O, Sancono A, Kratzschmar J, Kreft B, Stassen M, Cato AC. Glucocorticoids inhibit MAP kinase via increased expression and decreased degradation of MKP-1. EMBO J. 2001;20(24):7108-16.

51. Allard B, Bara I, Gilbert G, Carvalho G, Trian T, Ozier A, et al. Protease activated receptor-2 expression and function in asthmatic bronchial smooth muscle. PLoS One. 2014;9(2):e86945.

52. Jairaman A, Maguire CH, Schleimer RP, Prakriya M. Allergens stimulate store-operated calcium entry and cytokine production in airway epithelial cells. Sci Rep. 2016;6:32311.

53. Mason EF, Rathmell JC. Cell metabolism: an essential link between cell growth and apoptosis. Biochim Biophys Acta. 2011;1813(4):645-54.

54. Xiao C, Puddicombe SM, Field S, Haywood J, Broughton-Head V, Puxeddu I, et al. Defective epithelial barrier function in asthma. J Allergy Clin Immunol. 2011;128(3):549-56 e1-12.

55. Hackett TL, Singhera GK, Shaheen F, Hayden P, Jackson GR, Hegele RG, et al. Intrinsic phenotypic differences of asthmatic epithelium and its inflammatory responses to respiratory syncytial virus and air pollution. Am J Respir Cell Mol Biol. 2011;45(5):1090-100.

56. Freishtat RJ, Watson AM, Benton AS, Iqbal SF, Pillai DK, Rose MC, et al. Asthmatic airway epithelium is intrinsically inflammatory and mitotically dyssynchronous. Am J Respir Cell Mol Biol. 2011;44(6):863-9.

57. Stevens PT, Kicic A, Sutanto EN, Knight DA, Stick SM. Dysregulated repair in asthmatic paediatric airway epithelial cells: the role of plasminogen activator inhibitor-1. Clin Exp Allergy. 2008;38(12):1901-10.

58. Biesalski HK, Nohr D. Importance of vitamin-A for lung function and development. Mol Aspects Med. 2003;24(6):431-40.

59. Arora P, Kumar V, Batra S. Vitamin A status in children with asthma. Pediatr Allergy Immunol. 2002;13(3):223-6.

60. Riccioni G, Barbara M, Bucciarelli T, di Ilio C, D'Orazio N. Antioxidant vitamin supplementation in asthma. Ann Clin Lab Sci. 2007;37(1):96-101.

61. Fogh J, Fogh JM, Orfeo T. One hundred and twenty-seven cultured human tumor cell lines producing tumors in nude mice. J Natl Cancer Inst. 1977;59(1):221-6.

62. Giard DJ, Aaronson SA, Todaro GJ, Arnstein P, Kersey JH, Dosik H, et al. In vitro cultivation of human tumors: establishment of cell lines derived from a series of solid tumors. J Natl Cancer Inst. 1973;51(5):1417-23.

63. Zhao Y, Yang J, Liao W, Liu X, Zhang H, Wang S, et al. Cytosolic FoxO1 is essential for the induction of autophagy and tumour suppressor activity. Nat Cell Biol. 2010;12(7):665-75.

64. Grabiec AM, Angiolilli C, Hartkamp LM, van Baarsen LG, Tak PP, Reedquist KA. JNKdependent downregulation of FoxO1 is required to promote the survival of fibroblast-like synoviocytes in rheumatoid arthritis. Ann Rheum Dis. 2015;74(9):1763-71.

65. Li C, Capan E, Zhao Y, Zhao J, Stolz D, Watkins SC, et al. Autophagy is induced in CD4+ T cells and important for the growth factor-withdrawal cell death. J Immunol. 2006;177(8):5163-8.

66. Marino G, Niso-Santano M, Baehrecke EH, Kroemer G. Self-consumption: the interplay of autophagy and apoptosis. Nat Rev Mol Cell Biol. 2014;15(2):81-94.

67. Tibbles LA, Woodgett JR. The stress-activated protein kinase pathways. Cell Mol Life Sci. 1999;55(10):1230-54.

68. Ahmad S, Ahmad A, Rancourt RC, Neeves KB, Loader JE, Hendry-Hofer T, et al. Tissue factor signals airway epithelial basal cell survival via coagulation and protease-activated receptor isoforms 1 and 2. Am J Respir Cell Mol Biol. 2013;48(1):94-104.

69. Iablokov V, Hirota CL, Peplowski MA, Ramachandran R, Mihara K, Hollenberg MD, et al. Proteinase-activated receptor 2 (PAR2) decreases apoptosis in colonic epithelial cells. J Biol Chem. 2014;289(49):34366-77.

# Chapter 6: Effect of cellular stress on proteinase-activated receptor-2 expression in primary human airway epithelial cells

**6.1 Aim:** To study the effect of oxidative stress, hypoxia, cyclic stretch and human rhino virus infection on PAR-2 expression in primary human airway epithelial cells (AECs)

# **6.2 Introduction**

The regulation of PAR-2 expression is poorly understood. Since PAR-2 expression is increased in inflammatory diseases (1-5), it is deduced that inflammatory environment plays a role in the regulation of PAR-2 expression. Due to increased number of active inflammatory cells that release reactive oxygen species, sites of inflammation are characterized by the presence of oxidative stress (6); also increased metabolic demands of cells and decreased availability of metabolic substrates at the site of inflammation may result in hypoxia (7).

Asthmatic airways are chronically inflamed. The presence of oxidative stress in the asthmatic airways is well documented (8, 9). Also hypoxia-induced transcription factors are activated in the airways of asthmatic individuals (10), suggesting the presence of hypoxic environment in the airways in asthma. There is evidence that oxidative stress (11) and hypoxia (12, 13) regulate PAR-2 expression on cells other than AECs.

AECs undergo normal mechanical stretch during respiration. Mechanical stretch can influence cell growth, differentiation and also cell activation (14, 15). Human rhino virus (HRV) are major cause of asthma exacerbation (16, 17) and mechanical stretching could be more pronounced during exacerbations.

We studied effect of these cellular stress stimuli on PAR-2 expression in primary human AECs.

# 6.3 Methods

Studies in this chapter were performed on AECs cultured as described in Chapter 2. After subjecting AECs to the stresses described below PAR-2 expression was studied using qRT-PCR as described in Chapter 2.

# Oxidative stress

SIN-1 hydrochloride spontaneously decomposes to yield reactive oxygen species (ROS) such as nitric oxide, superoxide anion and peroxynitrite leading to oxidative stress and is used to study the effects of ROS on cells. AECs were treated with SIN-1 hydrochloride (Calbiochem) for up to 24h.

# Hypoxia

AECs cultured in media with or without supplements (as described in chapter 5) were subjected to hypoxia (1% oxygen, 5% CO<sub>2</sub>, 94% nitrogen; Baker Ruskinn hypoxia chamber) or normoxia (Thermo Scientific Forma Steri-Cult CO<sub>2</sub> incubator) for 24h. To confirm that cells experienced hypoxia, we performed qRT-PCR for hypoxia-inducible factor-1-alpha (HIF-1A) using Taqman gene expression assay for HIF-1A (ThermoFisher, Hs00153153\_m1) and housekeeping gene GAPDH (ThermoFisher, Hs02786624\_g1).

# Cyclic stretch and human rhino virus (HRV) infection

These experiments were performed in collaboration with the laboratories of Dr. Richard Leigh and Dr. David Proud at the University of Calgary. I designed the experiments with them. The labs of Dr. Leigh and Dr. Proud performed the experiments and sent us samples to evaluate PAR-2 expression.

Primary human bronchial epithelial cells (AECs) were obtained by protease digestion of dissected airways from non-transplanted human lungs as described before (18). Cells were cultured in BEGM media (Lonza) and then subjected to the following treatments in BEBM media (Lonza) containing 1X ITS (Insulin 10  $\mu$ g/mL, Transferrin 10  $\mu$ g/mL and Selenium 0.01  $\mu$ g/mL, Lonza) for 24h:

- a. Media
- b. Human rhino virus-16 (HRV-16): Cells were infected with 10<sup>4.5</sup> Tissue Culture Infective Dose (TCID50) U/ml of HRV-16.
- c. Stretch: Cells were cultured on BioFlex<sup>®</sup> Culture Plates and subjected to 30 cycles of 18% stretch intensity per minute using FLEXCELL<sup>®</sup> FX-4000<sup>TM</sup> Tension Plus<sup>TM</sup> System and Tissue Train<sup>TM</sup> System.
- d. HRV-16 and Stretch

CXCL8 mRNA expression was studied using Taqman gene expression assay for CXCL8 (ThermoFisher, Hs00174103\_m1 and housekeeping gene GAPDH (ThermoFisher, Hs02786624\_g1).

### **Statistics**

Results are expressed as Mean  $\pm$  SEM. Results are analyzed using repeated measures ANOVA with Tukey's multiple comparison test. The *p*-value of less than 0.05 was considered statistically significant.

#### 6.4 Results and Discussion

#### **Oxidative stress**

Preliminary results show that treatment of AECs with a compound generating ROS does not upregulate PAR-2 expression (Fig. 6.1). The limitation of the experiments is that whether cells experienced oxidative stress upon treatment with SIN-1 hydrochloride was not assessed, which can be done by measuring the levels of ROS and anti-oxidant enzymes. Hence it would not be appropriate to say that preliminary results show no effect of oxidative stress on PAR-2 expression in AECs.



Fig. 6.1 Effect of oxidative stress on PAR-2 expression in AECs. AECs were treated with only growth media or growth media containing different concentration of SIN-1 and qRT-PCR for PAR-2 expression was performed at indicated time points. Data are presented as Mean of the two experiments.

#### Hypoxia

To validate that cells underwent hypoxia, upregulation in HIF-1A protein levels or HIF-1Amediated gene expression can be studied. HIF-1A mRNA expression is shown to decrease under prolonged hypoxic condition (12h to 24h) in alveolar epithelial cells (19), human lung microvascular endothelial cells (mentioned but data not shown in the referenced paper) (19) and human umbilical vein endothelial cells (20). We also observed decrease in HIF-1A expression upon 24h hypoxia in bronchial AECs (Fig. 6.2A) indicating that cells in fact underwent hypoxia. We also studied if simultaneous exposure to two stressors (hypoxia and media supplement deprivation) has any synergistic or additive effect on PAR-2 expression in AECs. Under normoxic condition, cells cultured in media without supplements showed 2.50 fold increase in PAR-2 expression compared to cells culture in media with supplements (Fig. 6.2B, black bars). Similarly, under hypoxic condition also, cells cultured in media without supplements showed 2.44 fold upregulation in PAR-2 expression compared to cells cultured in media with supplements (Fig. 6.2B, red bars), suggesting the absence of any interaction between the two stressors to modulate PAR-2 expression.

Cell subjected to hypoxia showed a tendency of decreased PAR-2 expression compared to cells subjected to normoxia. This hypoxia-induced decreased in PAR-2 expression was similar whether cells were cultured in media with supplements (1.36 fold decrease, right part of the Fig. 6.2B) or in media without supplements (1.39 fold decrease, left part of the Fig. 6.2B).



Fig. 6.2 Effect of hypoxia on PAR-2 expression in AECs. (A) AECs were cultured in media with supplements and subjected to normoxia (black bar) or hypoxia (red bar) for 24h and qRT-PCR for HIF-1A expression was done. (B) AECs were cultured in media without or with supplements and subjected to normoxia (black bar) or hypoxia (red bar) for 24h. qRT-PCR for PAR-2 expression was performed. Data are presented as Mean of the two experiments.

Using pulmonary arterial smooth muscle cells, it was shown that HIF-1A is not involved in PAR-2 expression under normoxic condition but was involved in hypoxia-induced PAR-2 upregulation (13). Even though our preliminary data indicates the activation of HIF-1A in AECs under hypoxic condition, more markers should be studied such as upregulation in HIF-1A protein levels and increase in HIF-1A-mediated gene expression to confirm HIF-1A activation in our cell culture system. PAR-2 mRNA expression was repressed under hypoxic condition

compared to normoxic condition in AECs. The mechanism responsible for hypoxia-induced gene repression is not well studied but there is evidence that Reptin, a chromatin-remodelling factor (21) and Repressor Element 1-Silencing Transcription Factor (22) may play a role in hypoxia-induced gene repression. Hence, further studies are required to understand if these mechanisms are involved in hypoxia-induced decrease in PAR-2 expression in AECs. It could also be possible that PAR-2 mRNA stability is decreased under hypoxic condition, which results in decreased expression of PAR-2 mRNA irrespective of the presence or absence of media supplements. This can be studied by comparing the rate of PAR-2 mRNA degradation in cells subjected to normoxia and hypoxia.

#### Cyclic stretch and HRV

Our collaborators have shown that mechanical stretch and HRV infection independently and synergistically upregulate CXCL8 mRNA expression in AECs (23). We first checked the induction of CXCL8 in our experiments as an experimental control to make sure that cells responded to virus infection and mechanical stretch similarly to what is observed before. Even though we observed similar trend to what our collaborators have shown, there was no significant upregulation in CXCL8 mRNA expression due to variability in the results (Fig. 6.3A). The variability could be due to differences in the susceptibility of AECs obtained from different donors to the studied stressors. We looked at the individual experiments and found that in one of the three experiments AECs did not upregulate CXCL8 mRNA expression in response to HRV infection or mechanical stretch alone or in combination.

Mechanical stretch and HRV infection, independently or in combination, did not modulate PAR-2 mRNA expression in AECs (Fig. 6.3B). There are not many reports demonstrating the effect of mechanical stress or virus infection on PAR-2 expression in primary cultures of AECs.

Mechanical stressors such as centrifugation and repeated pipetting are shown to upregulate PAR-2 expression in peripheral blood monocytes by mobilizing the intracellular stores of PAR-2 to the cell surface; authors did not study the effect of mechanical stressors on PAR-2 mRNA expression (24). In our study, it is possible that mechanical stretch upregulated PAR-2 protein expression on cell surface of AECs. We could not study this due to methodological limitations. For e.g. (i) as explained in Chapter 2, we were not able to study PAR-2 protein expression using

flow cytometry or western blot in AECs. (ii) We also could not perform confocal microscopy as the mechanical stretch treatment requires cells to be grown on a specific plate which cannot be mounted on microscope.

For virus infection, there are no reports of virus infection increasing PAR-2 expression in primary bronchial epithelial cells, the cell culture system that we used in our study. However there is evidence that influenza A upregulates PAR-2 expression in A549 cells (25), which is a cancerous alveolar epithelial cell line (26). Thus it is possible that, for PAR-2 expression, virus infections have different effect on primary cells vs. cell line and also have different effect on epithelial cells from different parts of the airways such as bronchi vs. alveoli. The other difference is that, in addition to the difference in the cell culture system used, influenza A is negative-sense single stranded RNA virus while HRV is positive-sense single stranded RNA virus. Hence, further experiments are required to understand if virus-induced PAR-2 upregulation is cell type specific or virus specific phenomenon.

(B)





Fig. 6.3 cyclic stretch and HRV infection do not modulate PAR-2 expression in AECs. NHBE cells were treated with only media or HRV or mechanical stretch or combination of HRV and mechanical stretch for 24h. qRT-PCR for (A) CXCL8 expression and (B) PAR-2 expression were performed. Data are presented as Mean  $\pm$  SEM; ns (not significant) between the groups indicated by horizontal lines.

#### 6.5 References

1. Steinhoff M, Corvera CU, Thoma MS, Kong W, McAlpine BE, Caughey GH, et al. Proteinase-activated receptor-2 in human skin: tissue distribution and activation of keratinocytes by mast cell tryptase. Exp Dermatol. 1999;8(4):282-94.

2. Ferrell WR, Lockhart JC, Kelso EB, Dunning L, Plevin R, Meek SE, et al. Essential role for proteinase-activated receptor-2 in arthritis. J Clin Invest. 2003;111(1):35-41.

3. Crilly A, Burns E, Nickdel MB, Lockhart JC, Perry ME, Ferrell PW, et al. PAR(2) expression in peripheral blood monocytes of patients with rheumatoid arthritis. Ann Rheum Dis. 2012;71(6):1049-54.

4. Wygrecka M, Kwapiszewska G, Jablonska E, von Gerlach S, Henneke I, Zakrzewicz D, et al. Role of protease-activated receptor-2 in idiopathic pulmonary fibrosis. Am J Respir Crit Care Med. 2011;183(12):1703-14.

5. Wei L, Yu X, Shi H, Zhang B, Lian M, Li J, et al. 15-PGDH/15-KETE plays a role in hypoxia-induced pulmonary vascular remodeling through ERK1/2-dependent PAR-2 pathway. Cell Signal. 2014;26(7):1476-88.

6. Biswas SK. Does the Interdependence between Oxidative Stress and Inflammation Explain the Antioxidant Paradox? Oxid Med Cell Longev. 2016;2016:5698931.

Eltzschig HK, Carmeliet P. Hypoxia and inflammation. N Engl J Med. 2011;364(7):656 65.

8. Kirkham P, Rahman I. Oxidative stress in asthma and COPD: antioxidants as a therapeutic strategy. Pharmacol Ther. 2006;111(2):476-94.

9. Dworski R. Oxidant stress in asthma. Thorax. 2000;55 Suppl 2:S51-3.

10. Lee SY, Kwon S, Kim KH, Moon HS, Song JS, Park SH, et al. Expression of vascular endothelial growth factor and hypoxia-inducible factor in the airway of asthmatic patients. Ann Allergy Asthma Immunol. 2006;97(6):794-9.

11. Aman M, Hirano M, Kanaide H, Hirano K. Upregulation of proteinase-activated receptor-2 and increased response to trypsin in endothelial cells after exposure to oxidative stress in rat aortas. J Vasc Res. 2010;47(6):494-506.

12. Svensson KJ, Kucharzewska P, Christianson HC, Skold S, Lofstedt T, Johansson MC, et al. Hypoxia triggers a proangiogenic pathway involving cancer cell microvesicles and PAR-2-

mediated heparin-binding EGF signaling in endothelial cells. Proc Natl Acad Sci U S A. 2011;108(32):13147-52.

13. Kwapiszewska G, Markart P, Dahal BK, Kojonazarov B, Marsh LM, Schermuly RT, et al. PAR-2 inhibition reverses experimental pulmonary hypertension. Circ Res. 2012;110(9):1179-91.

14. Ingber DE. Tensegrity II. How structural networks influence cellular information processing networks. J Cell Sci. 2003;116(Pt 8):1397-408.

15. Vogel V, Sheetz M. Local force and geometry sensing regulate cell functions. Nat Rev Mol Cell Biol. 2006;7(4):265-75.

16. Nicholson KG, Kent J, Ireland DC. Respiratory viruses and exacerbations of asthma in adults. BMJ. 1993;307(6910):982-6.

 Johnston SL, Pattemore PK, Sanderson G, Smith S, Lampe F, Josephs L, et al. Community study of role of viral infections in exacerbations of asthma in 9-11 year old children. BMJ. 1995;310(6989):1225-9.

18. Churchill L, Chilton FH, Resau JH, Bascom R, Hubbard WC, Proud D. Cyclooxygenase metabolism of endogenous arachidonic acid by cultured human tracheal epithelial cells. Am Rev Respir Dis. 1989;140(2):449-59.

19. Uchida T, Rossignol F, Matthay MA, Mounier R, Couette S, Clottes E, et al. Prolonged hypoxia differentially regulates hypoxia-inducible factor (HIF)-1alpha and HIF-2alpha expression in lung epithelial cells: implication of natural antisense HIF-1alpha. J Biol Chem. 2004;279(15):14871-8.

20. Bartoszewska S, Kochan K, Piotrowski A, Kamysz W, Ochocka RJ, Collawn JF, et al. The hypoxia-inducible miR-429 regulates hypoxia-inducible factor-1alpha expression in human endothelial cells through a negative feedback loop. FASEB J. 2015;29(4):1467-79.

21. Lee JS, Kim Y, Kim IS, Kim B, Choi HJ, Lee JM, et al. Negative regulation of hypoxic responses via induced Reptin methylation. Mol Cell. 2010;39(1):71-85.

22. Cavadas MA, Mesnieres M, Crifo B, Manresa MC, Selfridge AC, Keogh CE, et al. REST is a hypoxia-responsive transcriptional repressor. Sci Rep. 2016;6:31355.

23. Sergei Nikitenko SS, Jason Arnason, Chris Shelfoon, Cora Kooi, David Proud, and Richard Leigh. Cyclic stretch augments human rhinovirus induced inflammatory responses in airway epithelial cells. Allergy Asthma Clin Immunol. 2014;10(Suppl 1):A71.

24. Johansson U, Lawson C, Dabare M, Syndercombe-Court D, Newland AC, Howells GL, et al. Human peripheral blood monocytes express protease receptor-2 and respond to receptor activation by production of IL-6, IL-8, and IL-1 {beta}. J Leukoc Biol. 2005;78(4):967-75.

25. Kampfmann I, Bauer N, Johannes S, Moritz A. Differences in hematologic variables in rats of the same strain but different origin. Vet Clin Pathol. 2012;41(2):228-34.

26. Giard DJ, Aaronson SA, Todaro GJ, Arnstein P, Kersey JH, Dosik H, et al. In vitro cultivation of human tumors: establishment of cell lines derived from a series of solid tumors. J Natl Cancer Inst. 1973;51(5):1417-23.

# Chapter 7: Effect of media supplement deprivation on the expression of G protein-coupled receptors in primary human AECs

**7.1 Aim:** To study effect of media supplement deprivation on the expression of G-coupled receptors in primary human airway epithelial cells (AECs)

#### 7.2 Introduction and rationale for selecting PAR-1, ADRB2 and P2RY2 for study

We showed that the expression of PAR-2, a G-coupled protein receptor (GPCRs) that belongs to the PAR family, is increased in AECs cultured in media without supplements (media supplement deprivation) compared to cells cultured in media with supplements (Chapter 4). AECs express all four receptors of the PAR family, PAR-1 to PAR-4 (1, 2). To study if media supplement deprivation specifically regulates expression of PAR-2 within PAR family of receptors, we examined the effect of media supplement deprivation on the expression of PAR-1 in AECs. We opted to study the expression PAR-1 because, in contrast to PAR-3 and PAR-4, there are standardized reagents available for PAR-1, mainly antagonists, which will enable us to study the functional significance of media supplement deprivation-mediated regulation of PAR-1 expression, if the expression of PAR-1 is regulated.

PAR-1: PAR-1 and PAR-2 both are activated by serine proteinases. However, in contrast to PAR-2, which is activated by trypsin, PAR-1 is activated by thrombin and neutrophil elastase. PAR-1 is coupled with multiple G-proteins such as  $G_q$ ,  $G_{12/13}$ ,  $G_i$ . On AECs, PAR-1 activation increases the survival of basal cells through a tissue factor-dependent pathway (3). Further, similar to PAR-2, PAR-1 activation on AECs releases inflammatory mediators CXCL8 (4), CCL2 (5), IL-6 and PGE2 (1) but there are no studies showing involvement of PAR-1 in the pathogenesis of asthma. Finally, PAR-1 activation of fibroblasts induces proliferation (6) and myofibroblast differentiation (7), suggesting a role in the pathogenesis of fibroproliferative lung diseases (8).

In addition to PAR family, primary cultures of AECs are shown to express GPCRs belonging to adrenergic receptor family (9), purinergic receptor family (10), prostanoid receptor family (11, 12) and chemokine receptor family (13, 14). To study if media supplement deprivation affects expression of other families of GPCRs in AECs, we studied the expression of receptors that

belong to adrenergic receptor family and purinergic receptor family as these families of GPCRs are known to play a regulatory role on primary innate defense mechanisms of AECs as described below:

Beta-2 adrenergic receptor (ADRB2, adrenergic receptor family): The agonist for this G<sub>s</sub>coupled receptor is epinephrine. On AECs, activation of ADRB2 regulates innate defense functions such as ciliary beating frequency (15) and airway epithelial barrier properties (16). Beta-2 agonists (bronchodilators) are the primary medication for asthma that act through ADRB2 receptors to induce smooth muscle relaxation and bronchodilation (17). Interestingly, ADRB2 activation on airway epithelium plays a protective role against bronchoconstriction. By removing epithelial layer from tracheal preparations *in vitro* (18, 19) and by using transgenic mice over-expressing ADRB2 in airway epithelium (20), it was shown that ADRB2 activation on airway epithelial cells may regulate airway responsiveness and have bronchodilating effect by inducing smooth muscle relaxation.

P2Y purinoceptor 2 (P2RY2, purinergic receptor family): The agonists for this purinergic receptor are ATP and UTP nucleotides. P2RY2 is a  $G_q$ -coupled receptor that on airway epithelium regulates ciliary beating frequency and mucus secretion (10), thus provides protection against invading foreign particles. Interestingly, recent evidence has showed that aeroallergens acts on AECs to release ATP, which works as a "danger signal" and activates P2RY2 in an autocrine manner to induce the release of IL-33 from AECs (21); IL-33 is a mediator that plays a vital role in the pathogenesis of Th2 inflammation (22).

Finally, to evaluate if the media supplement deprivation-mediated regulation of receptor expression in AECs is limited to GPCRs, we also studied the effect of media supplement deprivation on expression of a non-GPCR, toll-like receptor 3 (TLR-3).

TLR-3: This receptor of the innate immune system belongs to the family of pattern recognition receptors (PRRs), a receptor family that gets activated by pathogenassociated molecular patterns (molecules associated with microbial pathogens). TLR-3 is a receptor for double-stranded RNA viruses. In AECs, TLR-3 activation induces the release of cytokines, chemokines and interferons as an immune response (23, 24). Rhinoviruses (RV) are a major cause of asthma exacerbation (25, 26). TLR-3 is shown to have an anti-viral response against RV by decreasing virus replication in AECs (27). However, in asthma, the anti-viral effects of TLR-3 in AECs are decreased not because of decreased expression of TLR-3 but because of impaired TLR-3 signaling (28).

# 7.3 Methods

AECs were cultured and subjected to media supplement deprivation as described before (Chapter 5). The following Taqman gene expression assays (ThermoFisher) were used to study the expression of gene of interest and *Cyclophilin A (PPIA)* (housekeeping gene) by qRT-PCR.

Gene of interest	Taqman Gene expression assay
PAR-1	Hs00169258_m1
ADRB2	Hs00240532_s1
P2RY2	Hs01856611 s1
TID 2	- Ha01551070 a1
1LK-5	ns01551079_g1
Cyclophilin A (PPIA)	Hs04194521_s1

#### 7.4 Results and Discussion



Fig. 7.1 Effect of media supplement deprivation on the expression of PAR-1, ADRB2, P2RY2 and TLR-3 receptors. AECs were cultured in media with supplements until reached 80% confluency (Time "0"). Cells were further cultured in media without supplements (red bars) or media with supplements (black bars) for 24h. The graphs show relative mRNA expression for (A) PAR-1, (B) ADRB2 (C) P2RY2 and (D) TLR-3. Data are presented as Mean  $\pm$  SEM; ns (not significant), \*\**P*<0.01, \*\*\**P*<0.001 between the groups indicated by horizontal lines.

Six experiments that were used to analyze the effect of media supplement deprivation on the expression of PAR-2 were also evaluated for the expression of other receptors. Media supplement deprivation modulated expression of all 3 G-coupled receptors. The expression of PAR-1 (Fig. 7.1A) and ADRB2 (Fig. 7.1B) was increased, while the expression of P2RY2 (Fig. 7.1C) was decreased. The expression of TLR-3, a non-GPCR was not affected (Fig. 7.1D).

TLR-3 activation is shown to release type I interferons and increase TLR-3 expression in AECs (29), probably through an autocrine action of type I interferons (30). It is shown that

inflammatory mediators such as TLR-4 agonist lipopolysaccharide (LPS), TNF and IL-1 $\beta$  do not modulate TLR-3 expression in AECs (29). On the other hand, the expression of GPCRs is regulated by inflammatory mediators. For example, TGF $\beta$  is shown to increase PAR-1 expression in AECs (31). Otherwise there is no information if inflammatory mediators alter expression of ADRB2 and P2RY2 in AECs. For cells other than AECs, TNF upregulates PAR-1 expression in fibroblast (32), while LPS, TNF and IL-1 $\beta$  upregulate P2RY2 expression in vascular smooth muscle cells (33). In summary, the literature suggests that expression of TLR-3 and GPCRs is regulated differently; this is what we observed in our experiments also. It is possible that media supplement deprivation may affect expression of only GPCRs but still expression of other GPCRs and non-GPCRs need to be studied before reaching this conclusion.

Similar to PAR-2 (2.3 fold upregulation, Fig. 5.1E in Chapter 5), the expression of PAR-1 was also increased (2.6 fold) upon media supplement deprivation. For PAR-2, the primary media supplement regulating PAR-2 expression was insulin (Chapter 2). Hence further experiments are required to study if insulin also regulates expression of PAR-1, another member of the same receptor family. There is one study showing that LPS increased PAR-2 expression in AECs without affecting PAR-1 expression (34). This suggests that the regulation of these two receptors may be stimulus specific and hence it is possible that media supplement other than insulin may regulate PAR-1 expression in AECs.

Very limited information is available regarding the mechanisms that regulate expression of the studied GPCRs in general and especially on AECs. So there are a few possibilities for what could regulate expression of GPCRs studied:

(i) The exclusion of media supplements regulate expression of receptors by activating signaling pathways which are otherwise not activated when a single supplement is excluded from the cell culture media. For example, media supplement deprivation can activate stress-activated protein kinase JNK and p38 signaling pathways (35) and ERK1/2 (36), which result in activation of a range of transcription factors including AP-1, ELK-1, ATF-2. The study of these signaling pathways will show whether these GPCRs are regulated by a single signaling pathway.

Cells release ATP as a result of respiration and ATP is an agonist for P2RY2. Cellular stressors such as injury (37), exposure to air pollutants (38) or virus infections (39) modulate the release of ATP or the expression of P2RY2. Some of the media supplements are growth factors for cells and it is possible that exclusion of media supplements may generate stress that could alter ATP release and thus modulate P2RY2 expression in an autocrine manner. This can be studied by measuring ATP release in cell supernatants in cell cultured with and without media supplements. Further, cells cultured in the absence of media supplements can be supplied with ATP and P2RY2 expression can be studied.

Out of GPCR families studied, media supplement deprivation upregulated receptors belonging to PAR and adrenergic receptor families, while downregulated receptor belonging to purinergic receptor family. It may be possible that the cellular stress induced by media supplement deprivation exerts differential effects on the expression of different families of GPCRs. The expression of other members of these receptor family are regulated similarly by media supplement deprivation. To do this we require information about what other receptors are expressed in AECs, but this information is limited in case of adrenergic and purinergic receptor families. Microarray-based detection can be performed to identify the GPCRs can be challenging as microarrays that study genome-wide RNA expression are not optimized to detect GPCRs (40). This is most probably because GPCRs are expressed at low levels; qRT-PCR based approaches are developed to overcome this limitation (41). Once known, other relevant receptors can be selected for future studies.

(ii) It is also possible that a particular media supplement exerts regulatory role on all or a particular GPCR. For example (a) we showed that exclusion of insulin from cell culture media upregulated PAR-2 expression (Chapter 2). Even though there is no evidence that insulin has a regulatory role for a wide range of GPCRs, further experiments are required to identify if insulin regulates GPCRs belonging to various families in our cell culture system. (b) One of the components of media supplement is epinephrine. Epinephrine is

shown to regulate the expression of a class of adrenergic receptors, Alpha-2A adrenergic receptors (42). Epinephrine is a ligand for ADRB2, so exclusion of epinephrine may modulate the expression of adrenergic receptors. (c) Finally, retinoic acid, another component of media supplements, is shown to upregulate the expression of P2RY2 in dose-dependent manner in keratinocytes (43). If retinoic acid has the similar effect on P2RY2 expression in AECs, then it is possible that the removal of retinoic acid may result in the downregulation of P2RY2 expression. Thus further experiments are required to study all these possibilities.

(iii)Another possibility is that there is an interaction between two or more media supplements for regulating expression of these GPCRs. Thus the observed effect on the expression of GPCRs is may be the result of exclusion of more than one media supplements but not all. For e.g. in human uterine cervical epithelial cells, epinephrine increases the degradation of purinergic receptor P2X7, the effect which is potentiated by EGF. The study to check this possibility will involve very complicated experiments where absence or presence of different media supplement combinations need to be studied to understand their role in regulation of GPCRs expression.

Further experiments are required to understand if the observed changes in mRNA expression of these receptors are the result of alteration in gene transcription or mRNA stability. It is also important to study whether the observed changes in mRNA expression are translated at protein levels. For example, in cells other than AECs, it is shown that increase in PAR-1 mRNA did not result in increase in PAR-1 protein (44, 45). If the observed changes in mRNA expression are translated into protein expression, the upregulation in PAR-1 (3) and ADRB2 (15, 16) could be protective from cell death or airway hyperresponsiveness, respectively based on their described roles in AECs. The decreased expression of P2RY2 may play an important role in controlling inflammation by decreasing the release of IL-33 but on the other hand this may limit the mucociliary clearance function of AECs as P2RY2 regulates ciliary beating frequency and mucus secretion in airway epithelium (10). Differentiated cultures of AECs can be subjected to media supplement deprivation and then treated with agonists of these receptors or allergens and AEC functions, such as barrier property, ciliary beating and mucus secretion, and inflammatory

mediator release can be studied to understand the functional significance of this modulation in receptor expression.

#### 7.5 References

1. Asokananthan N, Graham PT, Fink J, Knight DA, Bakker AJ, McWilliam AS, et al. Activation of protease-activated receptor (PAR)-1, PAR-2, and PAR-4 stimulates IL-6, IL-8, and prostaglandin E2 release from human respiratory epithelial cells. J Immunol. 2002;168(7):3577-85.

2. Knight DA, Lim S, Scaffidi AK, Roche N, Chung KF, Stewart GA, et al. Proteaseactivated receptors in human airways: upregulation of PAR-2 in respiratory epithelium from patients with asthma. J Allergy Clin Immunol. 2001;108(5):797-803.

3. Ahmad S, Ahmad A, Rancourt RC, Neeves KB, Loader JE, Hendry-Hofer T, et al. Tissue factor signals airway epithelial basal cell survival via coagulation and protease-activated receptor isoforms 1 and 2. Am J Respir Cell Mol Biol. 2013;48(1):94-104.

4. Chiu LL, Perng DW, Yu CH, Su SN, Chow LP. Mold allergen, pen C 13, induces IL-8 expression in human airway epithelial cells by activating protease-activated receptor 1 and 2. J Immunol. 2007;178(8):5237-44.

5. Wang H, Yi T, Zheng Y, He S. Induction of monocyte chemoattractant protein-1 release from A549 cells by agonists of protease-activated receptor-1 and -2. Eur J Cell Biol. 2007;86(4):233-42.

6. Blanc-Brude OP, Archer F, Leoni P, Derian C, Bolsover S, Laurent GJ, et al. Factor Xa stimulates fibroblast procollagen production, proliferation, and calcium signaling via PAR1 activation. Exp Cell Res. 2005;304(1):16-27.

7. Scotton CJ, Krupiczojc MA, Konigshoff M, Mercer PF, Lee YC, Kaminski N, et al. Increased local expression of coagulation factor X contributes to the fibrotic response in human and murine lung injury. J Clin Invest. 2009;119(9):2550-63.

8. Jose RJ, Williams AE, Chambers RC. Proteinase-activated receptors in fibroproliferative lung disease. Thorax. 2014;69(2):190-2.

9. Salathe M. Effects of beta-agonists on airway epithelial cells. J Allergy Clin Immunol. 2002;110(6 Suppl):S275-81.

10. Lazarowski ER, Boucher RC. Purinergic receptors in airway epithelia. Curr Opin Pharmacol. 2009;9(3):262-7.

11. Chiba T, Kanda A, Ueki S, Ito W, Yamaguchi K, Kamada Y, et al. Possible novel receptor for PGD2 on human bronchial epithelial cells. Int Arch Allergy Immunol. 2007;143 Suppl 1:23-7.

12. Stinson SE, Amrani Y, Brightling CE. D prostanoid receptor 2 (chemoattractant receptorhomologous molecule expressed on TH2 cells) protein expression in asthmatic patients and its effects on bronchial epithelial cells. J Allergy Clin Immunol. 2015;135(2):395-406.

 Farkas L, Hahn MC, Schmoczer M, Jentsch N, Kratzel K, Pfeifer M, et al. Expression of CXC chemokine receptors 1 and 2 in human bronchial epithelial cells. Chest. 2005;128(5):3724-34.

14. Beck LA, Tancowny B, Brummet ME, Asaki SY, Curry SL, Penno MB, et al. Functional analysis of the chemokine receptor CCR3 on airway epithelial cells. J Immunol. 2006;177(5):3344-54.

15. Devalia JL, Sapsford RJ, Rusznak C, Toumbis MJ, Davies RJ. The effects of salmeterol and salbutamol on ciliary beat frequency of cultured human bronchial epithelial cells, in vitro. Pulm Pharmacol. 1992;5(4):257-63.

16. Coraux C, Kileztky C, Polette M, Hinnrasky J, Zahm JM, Devillier P, et al. Airway epithelial integrity is protected by a long-acting beta2-adrenergic receptor agonist. Am J Respir Cell Mol Biol. 2004;30(5):605-12.

17. Barnes PJ. Beta-adrenergic receptors and their regulation. Am J Respir Crit Care Med. 1995;152(3):838-60.

18. Flavahan NA, Aarhus LL, Rimele TJ, Vanhoutte PM. Respiratory epithelium inhibits bronchial smooth muscle tone. J Appl Physiol (1985). 1985;58(3):834-8.

19. Stuart-Smith K, Vanhoutte PM. Heterogeneity in the effects of epithelium removal in the canine bronchial tree. J Appl Physiol (1985). 1987;63(6):2510-5.

20. McGraw DW, Forbes SL, Mak JC, Witte DP, Carrigan PE, Leikauf GD, et al. Transgenic overexpression of beta(2)-adrenergic receptors in airway epithelial cells decreases bronchoconstriction. Am J Physiol Lung Cell Mol Physiol. 2000;279(2):L379-89.

21. Kouzaki H, Iijima K, Kobayashi T, O'Grady SM, Kita H. The danger signal, extracellular ATP, is a sensor for an airborne allergen and triggers IL-33 release and innate Th2-type responses. J Immunol. 2011;186(7):4375-87.

22. Borish L, Steinke JW. Interleukin-33 in asthma: how big of a role does it play? Curr Allergy Asthma Rep. 2011;11(1):7-11.

23. Ritter M, Mennerich D, Weith A, Seither P. Characterization of Toll-like receptors in primary lung epithelial cells: strong impact of the TLR3 ligand poly(I:C) on the regulation of Toll-like receptors, adaptor proteins and inflammatory response. J Inflamm (Lond). 2005;2:16.

24. Sha Q, Truong-Tran AQ, Plitt JR, Beck LA, Schleimer RP. Activation of airway epithelial cells by toll-like receptor agonists. Am J Respir Cell Mol Biol. 2004;31(3):358-64.

25. Nicholson KG, Kent J, Ireland DC. Respiratory viruses and exacerbations of asthma in adults. BMJ. 1993;307(6910):982-6.

26. Johnston SL, Pattemore PK, Sanderson G, Smith S, Lampe F, Josephs L, et al. Community study of role of viral infections in exacerbations of asthma in 9-11 year old children. BMJ. 1995;310(6989):1225-9.

27. Hewson CA, Jardine A, Edwards MR, Laza-Stanca V, Johnston SL. Toll-like receptor 3 is induced by and mediates antiviral activity against rhinovirus infection of human bronchial epithelial cells. J Virol. 2005;79(19):12273-9.

28. Parsons KS, Hsu AC, Wark PA. TLR3 and MDA5 signalling, although not expression, is impaired in asthmatic epithelial cells in response to rhinovirus infection. Clin Exp Allergy. 2014;44(1):91-101.

29. Guillot L, Le Goffic R, Bloch S, Escriou N, Akira S, Chignard M, et al. Involvement of toll-like receptor 3 in the immune response of lung epithelial cells to double-stranded RNA and influenza A virus. J Biol Chem. 2005;280(7):5571-80.

30. Ciencewicki JM, Brighton LE, Jaspers I. Localization of type I interferon receptor limits interferon-induced TLR3 in epithelial cells. J Interferon Cytokine Res. 2009;29(5):289-97.

31. Smoktunowicz N, Plate M, Stern AO, D'Antongiovanni V, Robinson E, Chudasama V, et al. TGFbeta upregulates PAR-1 expression and signalling responses in A549 lung adenocarcinoma cells. Oncotarget. 2016;7(40):65471-84.

32. Artuc M, Hermes B, Algermissen B, Henz BM. Expression of prothrombin, thrombin and its receptors in human scars. Exp Dermatol. 2006;15(7):523-9.

33. Hou M, Moller S, Edvinsson L, Erlinge D. Cytokines induce upregulation of vascular P2Y(2) receptors and increased mitogenic responses to UTP and ATP. Arterioscler Thromb Vasc Biol. 2000;20(9):2064-9.

34. Ostrowska E, Sokolova E, Reiser G. PAR-2 activation and LPS synergistically enhance inflammatory signaling in airway epithelial cells by raising PAR expression level and interleukin-8 release. Am J Physiol Lung Cell Mol Physiol. 2007;293(5):L1208-18.

35. Tibbles LA, Woodgett JR. The stress-activated protein kinase pathways. Cell Mol Life Sci. 1999;55(10):1230-54.

36. Sinha D, Bannergee S, Schwartz JH, Lieberthal W, Levine JS. Inhibition of ligandindependent ERK1/2 activity in kidney proximal tubular cells deprived of soluble survival factors up-regulates Akt and prevents apoptosis. J Biol Chem. 2004;279(12):10962-72.

37. Wesley UV, Bove PF, Hristova M, McCarthy S, van der Vliet A. Airway epithelial cell migration and wound repair by ATP-mediated activation of dual oxidase 1. J Biol Chem. 2007;282(5):3213-20.

38. Ahmad S, Ahmad A, McConville G, Schneider BK, Allen CB, Manzer R, et al. Lung epithelial cells release ATP during ozone exposure: signaling for cell survival. Free Radic Biol Med. 2005;39(2):213-26.

39. Seror C, Melki MT, Subra F, Raza SQ, Bras M, Saidi H, et al. Extracellular ATP acts on P2Y2 purinergic receptors to facilitate HIV-1 infection. J Exp Med. 2011;208(9):1823-34.

40. Insel PA, Snead A, Murray F, Zhang L, Yokouchi H, Katakia T, et al. GPCR expression in tissues and cells: are the optimal receptors being used as drug targets? Br J Pharmacol. 2012;165(6):1613-6.

41. Fredriksson R, Schioth HB. The repertoire of G-protein-coupled receptors in fully sequenced genomes. Mol Pharmacol. 2005;67(5):1414-25.

42. Reutter MA, Richards EM, Sumners C. Regulation of alpha2A-adrenergic receptor expression by epinephrine in cultured astroglia from rat brain. J Neurochem. 1998;70(1):86-95.

43. Fujishita K, Koizumi S, Inoue K. Upregulation of P2Y2 receptors by retinoids in normal human epidermal keratinocytes. Purinergic Signal. 2006;2(3):491-8.

44. Zhang H, Yang H, Zhang L, Yang X, Zhang Z, Lin Q, et al. Induction of IL-4 release and upregulated expression of protease activated receptors by GM-CSF in P815 cells. Cytokine. 2009;48(3):196-202.

45. Zhang H, Lin L, Yang H, Zhang Z, Yang X, Zhang L, et al. Induction of IL-13 production and upregulation of gene expression of protease activated receptors in P815 cells by IL-6. Cytokine. 2010;50(2):138-45.

# **Chapter 8: General Discussion**

# 8.1 Discussion and Future Directions

In asthma, PAR-2 expression is increased on AECs (1); cells that play an important role in the pathogenesis of lung inflammatory diseases. Upregulated PAR-2 expression leads to increased lung inflammation (2). Thus inhibition of PAR-2 upregulation is a rational strategy to control PAR-2-mediated inflammation, but regulation of PAR-2 expression is poorly studied. Since upregulated PAR-2 expression is reported in a variety of inflammatory conditions, it is believed that the inflammatory milieu regulates PAR-2 expression.

Inflammation leads to insulin resistance, a condition where insulin signaling is decreased or blocked and the function of insulin is impaired. HFD-induced obesity in mice is a standard model to induce insulin resistance in adipose tissue and skeletal muscle tissue. PAR-2 expression increases in insulin resistant adipose tissue in HFD fed obese mice (3-5). Further, PAR-2 expression decreases upon improving insulin sensitivity in insulin resistant adipose tissue (3). Although it was not shown directly, it is possible that insulin may directly regulate PAR-2 expression in these tissues. Due to insulin resistance in adipose tissue and skeletal muscle, these tissues do not perform insulin-directed glucose uptake. Pancreatic  $\beta$ -cells sense the increased levels of circulating glucose and, in response, secrete more insulin, which results in higher levels of insulin in circulation. Even though this state is known as systemic insulin resistance, the status of insulin sensitivity in lungs is not known.

To study if insulin modulates PAR-2 expression in lungs *in vivo*, we used a similar mouse model of HFD-induced obesity. We found that the mice fed with HFD had higher levels of circulating insulin, compared to mice fed normal diet, but the baseline insulin signaling was not altered in the lungs of obese mice. Further, we showed that higher levels of circulating insulin in obese mice were associated with lower levels of lung PAR-2 expression, hence it is likely that insulin regulates PAR-2 expression in lungs and increased levels of insulin decreased PAR-2 expression in lungs of obese mice.

Obesity is associated with allergic sensitization and allergic asthma in humans (6). Using similar model of HFD-induced obesity, obese animals are shown to be prone to develop allergic
sensitization (7) and eosinophilic airway inflammation (7, 8). PAR-2 is shown to play a vital role in the development of allergic sensitization, allergic airway inflammation and airway remodeling (9-13) and thus it may be possible that PAR-2 expression would increase in lungs of obese mice and humans, which makes them more susceptible to develop allergic asthma. Opposite to this rationale, we found decreased PAR-2 expression in lungs of obese mice. Hence it may be possible that the mechanism underlying obesity-induced asthma does not involve PAR-2.

There are several mechanisms proposed for the development of asthma in obesity (14). Low grade systemic inflammation is present in obesity resulting in increased systemic levels of inflammatory cytokines IL-6, TNF and IL-1. These cytokines could affect lung functions by acting on SMCs, fibroblasts and epithelial cells in the airways (15, 16). Also during obesity, the expression of adipokines is altered and this alteration may favor the development of asthma. For example, adipokine leptin, whose level is increased in obesity, is shown to increase AHR (17). On the other hand, adipokine adiponectin prevents allergic airway inflammation, AHR and airway remodeling (18, 19) but the level of adiponectin is decreased in obesity. Also higher levels of circulating insulin, in obesity, is shown to develop AHR (20), alter lung structure (21) and decrease lung function (22) by upregulating collagen deposition and peribronchial fibrosis. Finally, insulin increases expression of contractile phenotypic markers in airway smooth muscle (23-25) indicating insulin may upregulate contractile potential of the airways. These results indicate that the alteration in the levels of cytokines, adipokines and hormones in obesity make animals susceptible to the development of asthma. Hence the decreased expression of PAR-2 in the lungs of these obese animals may indicate that PAR-2 play a very limited role in the development of asthma. Further studies are required to understand the role of PAR-2 in the development of allergic sensitization and allergic airway inflammation in obesity. This can be done by developing animal model of asthma in PAR-2 KO obese mice or by developing animal model of asthma in wild type obese mice with the allergens containing PAR-2 activating property or PAR-2 agonists.

Using primary human AECs, we show that insulin regulates PAR-2 expression through the PI3K-Akt-FOXO-1 pathway and alterations in insulin signaling result in increased PAR-2 expression in AECs. Thus, *in vivo*, alteration in insulin signaling may result in PAR-2

upregulation. One of the mechanisms that alters insulin signaling *in vivo* is insulin resistance. In adipose tissue, where PAR-2 upregulation is correlated with insulin resistance, increased levels of inflammatory mediators such as IL-6, IL-1β and TNF are shown to induce insulin resistance (26). Since asthmatic airways are chronically inflamed and the mediators that induce insulin resistance in adipose tissue are also increased in the airways in asthma (27-30), it would be an interesting hypothesis to study whether the observed upregulation in expression of epithelial PAR-2 in asthma is due to localized insulin resistance in the airway epithelium. Our hypothesis can be studied by obtaining bronchial biopsies, from healthy and asthmatic individuals, and correlating the presence of markers of insulin resistance with expression of PAR-2 in the airway epithelium. A positive correlation between presence of insulin resistance and PAR-2 expression will further support our hypothesis. Finally, to strengthen the idea that insulin resistance upregulates PAR-2 expression in asthmatic AECs, it is important to show that PAR-2 expression is increased upon inducing insulin resistance in AECs in vitro. Currently, there is no established *in vitro* cellular model of insulin resistance in AECs. Also it is not known whether AECs develop insulin resistance. A recent paper has shown decreased expression of insulin receptors in AECs obtained from asthma patients compared to healthy individuals (31). However, it is not clear whether decrease in insulin receptor expression, in these AECs, alters insulin signaling to induce insulin resistance. Other in vitro cultures of hepatoma cells and adipocytes have shown that phorbol myristate acetate (PMA), anisomycin (c-Jun N-terminal kinase-1 (JNK-1) activator), calyculin A (inhibitor of serine/threonine phosphatase) (32) and TNF (33) can induce insulin resistance by serine phosphorylation of IRS-1. On the other hand IL-1B (34), IL-6 (33) and dexamethasone (35) decrease IRS-1 expression to induce insulin resistance. Similar strategies can be used to develop *in vitro* cellular model of insulin resistance in AECs.

In addition to evaluating the effect of insulin resistance on PAR-2 expression, *in vitro* cellular models of insulin resistance in AECs will be a valuable tool to understand the unstudied area of role of insulin in innate immune functions and homeostatic functions of AECs. A couple of studies published in 1990s suggested chemotactic activity of insulin for AECs (36, 37). This function of insulin can play an important role in epithelial repair after airway injury and inefficient insulin function could result in delayed wound healing. Recent evidence indicates an important role of insulin in AEC differentiation (31) and regulation of their innate immune

functions, such as barrier function (38) and inflammatory mediator release (39). Further, it will be worth evaluating if insulin has any role in other defense mechanisms of AECs, such as ciliary beating and mucus secretion. Altered insulin signaling may limit these immune functions of AECs and result in decreased protection against inhaled particles. Finally, in AECs, insulin regulates glucose uptake and thus maintains the levels of glucose in the airway surface liquid (ASL) (38). Altered insulin signaling in the airway epithelium could result in dysregulated glucose homeostasis and increased levels of glucose in ASL. Using an animal model, it was shown that higher levels of glucose in ASL could increase the likelihood of bacterial respiratory infections (40). The discussed literature describing the effect of insulin on AECs is derived from proof of principle experiments where insulin was either added to or removed from cell culture media. However, the currently used approaches such as addition or exclusion of insulin or the use of inhibitors of insulin signaling do not mimic the actual *in vivo* condition of altered insulin signaling.

We have used *in vitro* differentiated cultures of AECs that mimic the *in vivo* airway epithelium. Thus, developing an *in vitro* model of insulin resistance in these differentiated AECs will not provide insight into the role of insulin in the functions of AECs, but will also show if AECs have any alternate strategies to cope up with altered insulin signaling. For example, insulin regulates glucose uptake in cells by regulating glucose transporters. There are two types of glucose transporters; one being dependent on insulin actions, while the other is independent of insulin actions for glucose uptake. AECs express both insulin-dependent and -independent glucose transporters (38, 41, 42). Hence it is possible that under conditions where insulin signaling is altered, AECs increase their efficiency of glucose uptake through insulin independent glucose transporters to maintain their glucose uptake.

After understanding the mechanism of insulin-mediated regulation of PAR-2 expression, we focused on the other potential mechanisms involved in the regulation of PAR-2 expression in AECs. There is evidence that characteristics of inflammatory microenvironment such as inflammatory mediators (43-45), hypoxia (46, 47) and growth factor deprivation (48) regulate PAR-2 expression on cells other than AECs. We hypothesized that these cellular stressors of inflammatory microenvironment regulates PAR-2 expression in AECs.

We have not yet studied the effect of inflammatory mediators on PAR-2 expression in AECs. The literature suggests that mainly TNF and IL-1 $\beta$  upregulate PAR-2 expression (Table- 1.2). Thus future studies are required to identify if TNF and IL-1 $\beta$  upregulates PAR-2 expression in AECs.

Our preliminary results indicate that hypoxia may decrease PAR-2 expression on AECs. Hypoxia is shown to upregulate PAR-2 expression in smooth muscle cells (47), where the upregulation was HIF-1A and platelet-derived growth factor-BB dependent. We found that hypoxia decreased HIF-1A mRNA expression in AECs indicating possible HIF-1A activation, but PAR-2 expression was decreased. Due to limited literature available on the effect of hypoxia in AECs and airway SMCs, it is not clear why hypoxia exerts opposite effects on PAR-2 expression in these two cell types. It has been shown that hypoxia decreases proliferation of airway SMCs (49), while increases proliferation of goblet cells, a type of AECs (50). Hence it may be possible that hypoxia has different effects on these cells. In our experiments we assessed HIF-1A mRNA levels, which is an indirect approach to evaluate if AECs underwent hypoxia and activated HIF-1A. HIF-1A activation should be confirmed by showing increase in HIF-1A protein expression of HIF-1A-regulated genes. For example, in AECs, HIF-1A increases mucin gene expression upon hypoxia (50). So, further analysis is required to confirm that cells underwent hypoxia in our experimental settings.

To mimic decreased availability of nutrients supporting cell growth *in vivo*, we excluded culture media supplements from the growth media in our AEC cultures *in vitro*. These supplements include growth factors and hormones among other factors. We found that exclusion of media supplements resulted in higher upregulation in PAR-2 expression compared to PAR-2 upregulation induced by the exclusion of only insulin. The media supplements, other than insulin, that had a regulatory effect on PAR-2 expression were hydrocortisone, BPE and RA.

The exclusion of hydrocortisone showed a trend of increased PAR-2 expression in AECs, while the addition of hydrocortisone to cells subjected to media supplement deprivation, decreased the media supplement deprivation-induced upregulation in PAR-2 expression. Since hydrocortisone has anti-inflammatory properties, PAR-2 downregulation could be one of its anti-inflammatory mechanisms. However, it is shown that asthmatics that were on inhaled steroids had similar upregulation in PAR-2 expression as asthmatics which were not using inhaled steroids (1). There are a few possibilities that could explain this difference in effect of corticosteroid on PAR-2 expression in vivo vs. in vitro. First, the regulation of PAR-2 expression by hydrocortisone in vitro could be an artifact of cell culture system and corticosteroids do not regulate PAR-2 expression in vivo. Second, the study that showed increased PAR-2 expression in airway epithelium of asthmatics that were on inhaled steroid included only 10 patients. Airway epithelial PAR-2 expression in 4 out of 10 asthma patients was very similar to healthy controls. Within these 10 patients, the dose of steroid ranged 400-1600 µg/day. The authors did not show any correlation between the dose of steroid and PAR-2 expression, but it may be possible that steroids decrease PAR-2 expression and the patients who were on higher dose of steroids had lower expression of airway epithelial PAR-2. Thus a study with larger sample size, where patients are divided into groups based on their dose of steroids, might explain if patients taking higher dose of steroids have lower PAR-2 expression. Third, it may be possible that the upregulation of PAR-2 expression in asthma is the result of epigenetic changes in the airway epithelium. Hence even though steroids act anti-inflammatory, PAR-2 expression stayed upregulated. Finally, corticosteroids are shown to induce insulin resistance (51, 52); this observation is true about systemic delivery of steroid that shows the development of insulin resistance in liver, skeletal muscle and adipose tissues. Moreover, direct in vitro exposure of dexamethasone is shown to induce insulin resistance in cultures of adipocytes (35). AECs are directly exposed to inhaled steroids. It may be possible that the dose of inhaled steroids and duration of exposure to inhaled steroids play a role in the development of insulin resistance in AECs. This may disrupt the insulin-mediated regulation of PAR-2 expression resulting in PAR-2 upregulation. Further studies can be done to understand the time- and dose-dependent effects of hydrocortisone on insulin sensitivity and PAR-2 expression in AECs.

The addition of BPE and RA to cells subjected to media supplement deprivation decreased the media supplement deprivation-induced upregulation in PAR-2 expression. Growth factor deprivation induces oxidative stress in different cells (53-55) to upregulate PAR-2 expression (56). Media supplements for AECs also contained growth factors hence it is possible that deprivation of media supplements in AECs may induce oxidative stress to upregulate PAR-2

expression. BPE and RA have anti-oxidant property (57, 58). Thus when we add BPE and RA to media supplement-deprived cells, BPE and RA may decrease PAR-2 expression by decreasing oxidative stress induced by media supplement deprivation.

Contrary to the hypothesis, that oxidative stress upregulates PAR-2 expression in AECs, our preliminary experiments did not show upregulation in PAR-2 expression when AECs were subjected to oxidative stress. To induce oxidative stress in AECs, we treated cells with SIN-1 hydrochloride compound, which spontaneously decomposes to yield reactive oxygen species (ROS) such as nitric oxide, superoxide anion and peroxynitrite. However the limitation of our experiments is that we did not evaluate if oxidative stress was induced, in AECs, at the used concentration of SIN-1 hydrochloride. Further experiments are required where AECs will be treated with different compounds, such as SIN-1 hydrochloride and hydrogen peroxide, to induce oxidative stress. First the induction of oxidative stress will be confirmed by measuring the levels of ROS and anti-oxidant enzymes and then PAR-2 expression will be studied.

In addition to oxidative stress, growth factor deprivation or media supplement deprivation can lead to activation of cellular stress responses such as stress-activated protein kinase (SAPK) JNK or p38 kinase (59, 60) or autophagy (59). Whether SAPKs are activated upon media supplement deprivation in AECs should be studied first. SAPK signaling leads to activation of AP-1, ELK-1, ATF-2 and other transcription factors, hence further experiments are required to understand the signaling pathways involved in media supplement deprivation-induced PAR-2 upregulation. Further, these SAPKs are known to activate autophagy (59, 61) as a survival mechanism. Autophagy occurs before apoptosis but prolonged autophagy results in apoptosis (62); activated SAPK maintain balance between autophagy and apoptosis (63). Since we did not see significant upregulation in the number of apoptotic cells upon media supplement deprivation, it is possible that autophagy was activated to upregulate PAR-2 expression. Further experiments are required to assess if autophagy is activated upon supplement deprivation and if autophagy plays any role in supplement deprivation-induced PAR-2 upregulation in AECs during autophagy (pre-apoptotic stage) could be a protective mechanism because PAR-2 activation is shown to inhibit apoptosis in colonic (64) and airway (65) epithelial cells.

We showed that exclusion of insulin, one of the media supplements, activates FOXO-1 transcription factor to upregulate PAR-2 expression. We then showed that exclusion of all the media supplements resulted in higher upregulation in PAR-2 expression compared to PAR-2 upregulation induced by the exclusion of only insulin. Based on these observations, we hypothesize that PAR-2 upregulation upon exclusion of all the media supplements will partially be dependent on FOXO-1. However, FOXO-1 inhibition further increased media supplement deprivation-induced PAR-2 upregulation indicating the involvement of different mechanisms in media supplement deprivation- vs. insulin deprivation-induced PAR-2 upregulation. The question is why does FOXO-1 involve in insulin deprivation-induced PAR-2 upregulation and not involved in media supplement deprivation-induced PAR-2 upregulation? Media supplements contained multiple growth factors. It is shown that under growth factor deprived condition, FOXO-1 activation (nuclear localization of FOXO-1) induces apoptosis (66, 67), while cytoplasmic localization of FOXO-1 (inactive FOXO-1 due to acetylation) during growth factor deprivation prevents apoptosis (66). It is also shown that growth factor deprivation activates JNK, which inhibits FOXO-1-induced apoptosis by decreasing FOXO-1 expression (68). In our hands, media supplement deprivation led to marginal increase in apoptotic cells. Thus, this evidence suggests that FOXO-1 was not activated during media supplement deprivation in our cells and hence played no role in PAR-2 upregulation. This possibility can be studied by (i) evaluating whether growth factor deprivation results in less nuclear localization of FOXO-1, compared to insulin deprivation and (ii) if FOXO-1 expression is decreased during media supplement deprivation, in AECs.

Studies have shown that asthmatic AECs cultured *ex vivo* maintain the intrinsic phenotypic differences (69-73). We have an observation that PAR-2 expression is increased in AECs from asthmatic individuals compared to healthy individuals and this upregulated PAR-2 expression is maintained when cells were cultured *ex vivo* for multiple passages. This property of asthmatic AECs will be useful to understand differences in the functions of epithelial PAR-2 between healthy and asthmatics individuals in order to understand the consequences of upregulated PAR-2 expression in asthma. Cell culture supernatants of PAR-2-activated healthy and asthmatics AECs can be analyzed for an array of cytokine and chemokines to identify the inflammatory mediators that are released in increased amount from asthmatic AECs cells. Further to confirm

that the increased release of mediators from asthmatic AECs is because of the higher expression of PAR-2, AECs from healthy individuals can be transfected with PAR-2 overexpression vector to upregulate PAR-2 expression to a level similar to that of asthmatic AECs. PAR-2 activation in these cells expressing higher levels of PAR-2 mimics increased PAR-2 functions of asthmatic cells (74). Thus higher release of inflammatory mediators by PAR-2 in these transfected cells will confirm that increased release of mediators from asthmatic AECs is because of the upregulated expression of PAR-2. This analysis of PAR-2-mediated release of inflammatory mediators from healthy vs. asthmatic AECs may guide us for the differential release of mediators between the two different phenotypes of cells. Finally, comparison of other outcomes of PAR-2 activation, such as mucus secretion (75) and epithelial barrier disruption (76, 77), between healthy and asthmatic AECs may show significant differences. Over all these comparative studies of PAR-2 functions between healthy and asthmatic AECs will help us understand the role of epithelial PAR-2 functions between so for a sthmatic AECs will help us understand the role of epithelial PAR-2 functions between so for asthmatic AECs will help us understand the role of epithelial PAR-2 functions between healthy and asthmatic AECs will help us understand the role of epithelial PAR-2 in the pathogenesis of asthma.

It is also reported that the phenotypic differences in asthmatic AECs are associated with the severity of asthma (78). In our study, two of the asthmatic individuals showed PAR-2 expression similar to healthy controls. Due to limited available information about the asthmatic individuals, we have not correlated PAR-2 expression with asthma severity and other characteristics of patients that the epithelial cells came from to identify if any clinical characteristic is associated with PAR-2 expression. We have shown that in patients having severe asthma compared to mild/moderate asthma (i) PAR-2 expression on peripheral pro-inflammatory monocytes is higher and (ii) higher number of pro-inflammatory monocytes expresses PAR-2 (79). In the same manuscript, we also showed that the patients that had higher number of PAR-2-expressing pro-inflammatory monocytes also experienced asthma exacerbation over the last year. Further studies are required to understand if PAR-2 expression in AECs varies based on the disease severity, as upregulated PAR-2 could exacerbate proteinase-mediated airway inflammation.

We are not sure at what point PAR-2 upregulation in AECs in asthma becomes irreversible as upregulated PAR-2 expression is maintained in asthmatic AECs cultured *ex vivo*. It could be possible that *in vivo* inflammatory microenvironment induces epigenetic changes that render asthmatic airway epithelium with higher PAR-2 expression. Epigenetic changes in the gene for

PAR-2, F2RL1, can be compared between asthmatic and healthy individuals to understand the epigenetic changes present in asthma. To identify if the observed epigenetic changes are involved in upregulated PAR-2 expression in asthma, in vitro epigenome editing assays can be performed where site-directed epigenetic changes can be introduced on DNA to determine the function of the epigenetic modifications. Using in vitro epigenome editing assays, we can introduce the same epigenetic changes that we observed in asthma in F2RL1 gene and study if these epigenetic changes increase PAR-2 expression. Even after this point, the questions still remain that what induces these epigenetic changes and at what time these epigenetic changes are incorporated into the PAR-2 gene? The recent literature suggests that chronic inflammation may induce epigenetic changes (80). Asthmatic airways are chronically inflamed. Chronic localized inflammation may lead to insulin resistance, decrease in nutrients supporting cell growth, oxidative stress and hypoxia in the airways. Insulin resistance (81), oxidative stress (82) and hypoxia (83) may induce epigenetic changes. We showed that insulin and cell growth nutrients regulate PAR-2 expression in AECs. Hence, it may be possible that localized insulin resistance in the airway epithelium and decrease in cell growth nutrients in the airways, in the background of inflammation, upregulate PAR-2 expression in AECs in asthma (Fig. 8.1 Conceptual model).

Our findings will encourage studies to evaluate if similar mechanisms are involved in the regulation of PAR-2 expression in other cells of lungs or cells of other organs such as colon and skin because in inflammatory diseases of these organs, localized inflammation is present and PAR-2 expression is increased. These studies may identify common dysregulatory mechanisms involved in increased PAR-2 expression in different inflammatory conditions and may lead to develop novel therapeutic strategies to inhibit PAR-2 upregulation and control PAR-2-mediated inflammation.

#### 8.2 Conceptual Model

In AECs, binding of insulin to insulin receptor activates the PI3K-Akt signaling cascade, which inactivates FOXO-1transcription factor. Also the presence of cell growth nutrients prevents activation of any SAPK signaling pathways. In this condition, PAR-2 expression is maintained at lower levels on AECs and PAR-2 activation results in controlled inflammation. Chronic airway inflammation leads to localized insulin resistance in the airway epithelium, as characterized by decrease in insulin receptor expression or serine phosphorylation of IRS-1, which blocks insulin signaling. The blockade in insulin signaling results in FOXO-1 activation and FOXO-1-mediated PAR-2 expression. Chronic airway inflammation also leads to decrease in the levels of cell growth nutrients, resulting in activation of the SAPK signaling cascade leading to upregulated PAR-2 expression through unknown transcription factors. The activation of upregulated PAR-2 on AECs augments airway inflammation.



Fig. 8.1 Conceptual Model.

#### 8.3 References

1. Knight DA, Lim S, Scaffidi AK, Roche N, Chung KF, Stewart GA, et al. Proteaseactivated receptors in human airways: upregulation of PAR-2 in respiratory epithelium from patients with asthma. J Allergy Clin Immunol. 2001;108(5):797-803.

2. Schmidlin F, Amadesi S, Dabbagh K, Lewis DE, Knott P, Bunnett NW, et al. Proteaseactivated receptor 2 mediates eosinophil infiltration and hyperreactivity in allergic inflammation of the airway. J Immunol. 2002;169(9):5315-21.

3. Lim J, Iyer A, Liu L, Suen JY, Lohman RJ, Seow V, et al. Diet-induced obesity, adipose inflammation, and metabolic dysfunction correlating with PAR2 expression are attenuated by PAR2 antagonism. FASEB J. 2013;27(12):4757-67.

4. Badeanlou L, Furlan-Freguia C, Yang G, Ruf W, Samad F. Tissue factor-proteaseactivated receptor 2 signaling promotes diet-induced obesity and adipose inflammation. Nat Med. 2011;17(11):1490-7.

5. Li M, Yang X, Zhang Y, Chen L, Lu H, Li X, et al. Activation of proteaseactivated receptor2 is associated with increased expression of inflammatory factors in the adipose tissues of obese mice. Mol Med Rep. 2015;12(4):6227-34.

6. Husemoen LL, Glumer C, Lau C, Pisinger C, Morch LS, Linneberg A. Association of obesity and insulin resistance with asthma and aeroallergen sensitization. Allergy. 2008;63(5):575-82.

7. Dietze J, Bocking C, Heverhagen JT, Voelker MN, Renz H. Obesity lowers the threshold of allergic sensitization and augments airway eosinophilia in a mouse model of asthma. Allergy. 2012;67(12):1519-29.

8. Calixto MC, Lintomen L, Schenka A, Saad MJ, Zanesco A, Antunes E. Obesity enhances eosinophilic inflammation in a murine model of allergic asthma. Br J Pharmacol. 2010;159(3):617-25.

9. Asaduzzaman M, Nadeem A, Arizmendi N, Davidson C, Nichols HL, Abel M, et al. Functional inhibition of PAR2 alleviates allergen-induced airway hyperresponsiveness and inflammation. Clin Exp Allergy. 2015;45(12):1844-55. 10. Davidson CE, Asaduzzaman M, Arizmendi NG, Polley D, Wu Y, Gordon JR, et al. Proteinase-activated receptor-2 activation participates in allergic sensitization to house dust mite allergens in a murine model. Clin Exp Allergy. 2013;43(11):1274-85.

11. Ebeling C, Lam T, Gordon JR, Hollenberg MD, Vliagoftis H. Proteinase-activated receptor-2 promotes allergic sensitization to an inhaled antigen through a TNF-mediated pathway. J Immunol. 2007;179(5):2910-7.

12. Arizmendi NG, Abel M, Mihara K, Davidson C, Polley D, Nadeem A, et al. Mucosal allergic sensitization to cockroach allergens is dependent on proteinase activity and proteinase-activated receptor-2 activation. J Immunol. 2011;186(5):3164-72.

13. Asaduzzaman M, Davidson C, Nahirney D, Fiteih Y, Puttagunta L, Vliagoftis H. Proteinase-activated receptor-2 blockade inhibits changes seen in a chronic murine asthma model. Allergy. 2018;73(2):416-20.

14. Baffi CW, Winnica DE, Holguin F. Asthma and obesity: mechanisms and clinical implications. Asthma Res Pract. 2015;1:1.

15. Rincon M, Irvin CG. Role of IL-6 in asthma and other inflammatory pulmonary diseases. Int J Biol Sci. 2012;8(9):1281-90.

16. Peters MC, Fahy JV. Metabolic consequences of obesity as an "outside in" mechanism of disease severity in asthma. Eur Respir J. 2016;48(2):291-3.

17. Shore SA, Schwartzman IN, Mellema MS, Flynt L, Imrich A, Johnston RA. Effect of leptin on allergic airway responses in mice. J Allergy Clin Immunol. 2005;115(1):103-9.

18. Shore SA, Terry RD, Flynt L, Xu A, Hug C. Adiponectin attenuates allergen-induced airway inflammation and hyperresponsiveness in mice. J Allergy Clin Immunol. 2006;118(2):389-95.

19. Medoff BD, Okamoto Y, Leyton P, Weng M, Sandall BP, Raher MJ, et al. Adiponectin deficiency increases allergic airway inflammation and pulmonary vascular remodeling. Am J Respir Cell Mol Biol. 2009;41(4):397-406.

20. Leiria LOS, Arantes-Costa FM, Calixto MC, Alexandre EC, Moura RF, Folli F, et al. Increased airway reactivity and hyperinsulinemia in obese mice are linked by ERK signaling in brain stem cholinergic neurons. Cell Rep. 2015;11(6):934-43.

21. Singh S, Bodas M, Bhatraju NK, Pattnaik B, Gheware A, Parameswaran PK, et al. Hyperinsulinemia adversely affects lung structure and function. Am J Physiol Lung Cell Mol Physiol. 2016;310(9):L837-45.

22. Forno E, Han YY, Muzumdar RH, Celedon JC. Insulin resistance, metabolic syndrome, and lung function in US adolescents with and without asthma. J Allergy Clin Immunol. 2015;136(2):304-11 e8.

23. Schaafsma D, Gosens R, Ris JM, Zaagsma J, Meurs H, Nelemans SA. Insulin induces airway smooth muscle contraction. Br J Pharmacol. 2007;150(2):136-42.

24. Schaafsma D, McNeill KD, Stelmack GL, Gosens R, Baarsma HA, Dekkers BG, et al. Insulin increases the expression of contractile phenotypic markers in airway smooth muscle. Am J Physiol Cell Physiol. 2007;293(1):C429-39.

25. Dekkers BG, Schaafsma D, Tran T, Zaagsma J, Meurs H. Insulin-induced laminin expression promotes a hypercontractile airway smooth muscle phenotype. Am J Respir Cell Mol Biol. 2009;41(4):494-504.

26. McArdle MA, Finucane OM, Connaughton RM, McMorrow AM, Roche HM. Mechanisms of obesity-induced inflammation and insulin resistance: insights into the emerging role of nutritional strategies. Front Endocrinol (Lausanne). 2013;4:52.

27. Howarth PH, Babu KS, Arshad HS, Lau L, Buckley M, McConnell W, et al. Tumour necrosis factor (TNFalpha) as a novel therapeutic target in symptomatic corticosteroid dependent asthma. Thorax. 2005;60(12):1012-8.

28. Neveu WA, Allard JL, Raymond DM, Bourassa LM, Burns SM, Bunn JY, et al. Elevation of IL-6 in the allergic asthmatic airway is independent of inflammation but associates with loss of central airway function. Respir Res. 2010;11:28.

29. Tillie-Leblond I, Pugin J, Marquette CH, Lamblin C, Saulnier F, Brichet A, et al. Balance between proinflammatory cytokines and their inhibitors in bronchial lavage from patients with status asthmaticus. Am J Respir Crit Care Med. 1999;159(2):487-94.

30. Borish L, Mascali JJ, Dishuck J, Beam WR, Martin RJ, Rosenwasser LJ. Detection of alveolar macrophage-derived IL-1 beta in asthma. Inhibition with corticosteroids. J Immunol. 1992;149(9):3078-82.

31. Loffredo LF, Abdala-Valencia H, Anekalla KR, Cuervo-Pardo L, Gottardi CJ, Berdnikovs S. Beyond epithelial-to-mesenchymal transition: Common suppression of differentiation programs underlies epithelial barrier dysfunction in mild, moderate, and severe asthma. Allergy. 2017;72(12):1988-2004.

32. Werner ED, Lee J, Hansen L, Yuan M, Shoelson SE. Insulin resistance due to phosphorylation of insulin receptor substrate-1 at serine 302. J Biol Chem. 2004;279(34):35298-305.

33. Rotter V, Nagaev I, Smith U. Interleukin-6 (IL-6) induces insulin resistance in 3T3-L1 adipocytes and is, like IL-8 and tumor necrosis factor-alpha, overexpressed in human fat cells from insulin-resistant subjects. J Biol Chem. 2003;278(46):45777-84.

34. Jager J, Gremeaux T, Cormont M, Le Marchand-Brustel Y, Tanti JF. Interleukin-1betainduced insulin resistance in adipocytes through down-regulation of insulin receptor substrate-1 expression. Endocrinology. 2007;148(1):241-51.

35. Buren J, Liu HX, Jensen J, Eriksson JW. Dexamethasone impairs insulin signalling and glucose transport by depletion of insulin receptor substrate-1, phosphatidylinositol 3-kinase and protein kinase B in primary cultured rat adipocytes. Eur J Endocrinol. 2002;146(3):419-29.

36. Shoji S, Ertl RF, Linder J, Koizumi S, Duckworth WC, Rennard SI. Bronchial epithelial cells respond to insulin and insulin-like growth factor-I as a chemoattractant. Am J Respir Cell Mol Biol. 1990;2(6):553-7.

37. Sanghavi JN, Rabe KF, Kim JS, Magnussen H, Leff AR, White SR. Migration of human and guinea pig airway epithelial cells in response to calcitonin gene-related peptide. Am J Respir Cell Mol Biol. 1994;11(2):181-7.

38. Molina SA, Moriarty HK, Infield DT, Imhoff BR, Vance RJ, Kim AH, et al. Insulin signaling via the PI3-kinase/Akt pathway regulates airway glucose uptake and barrier function in a CFTR-dependent manner. Am J Physiol Lung Cell Mol Physiol. 2017;312(5):L688-L702.

39. Seiler F, Hellberg J, Lepper PM, Kamyschnikow A, Herr C, Bischoff M, et al. FOXO transcription factors regulate innate immune mechanisms in respiratory epithelial cells. J Immunol. 2013;190(4):1603-13.

40. Gill SK, Hui K, Farne H, Garnett JP, Baines DL, Moore LS, et al. Increased airway glucose increases airway bacterial load in hyperglycaemia. Sci Rep. 2016;6:27636.

41. Garnett JP, Nguyen TT, Moffatt JD, Pelham ER, Kalsi KK, Baker EH, et al. Proinflammatory mediators disrupt glucose homeostasis in airway surface liquid. J Immunol. 2012;189(1):373-80.

42. Kalsi KK, Baker EH, Medina RA, Rice S, Wood DM, Ratoff JC, et al. Apical and basolateral localisation of GLUT2 transporters in human lung epithelial cells. Pflugers Arch. 2008;456(5):991-1003.

43. Xiang Y, Masuko-Hongo K, Sekine T, Nakamura H, Yudoh K, Nishioka K, et al. Expression of proteinase-activated receptors (PAR)-2 in articular chondrocytes is modulated by IL-1beta, TNF-alpha and TGF-beta. Osteoarthritis Cartilage. 2006;14(11):1163-73.

44. Hamilton JR, Frauman AG, Cocks TM. Increased expression of protease-activated receptor-2 (PAR2) and PAR4 in human coronary artery by inflammatory stimuli unveils endothelium-dependent relaxations to PAR2 and PAR4 agonists. Circ Res. 2001;89(1):92-8.

45. Nystedt S, Ramakrishnan V, Sundelin J. The proteinase-activated receptor 2 is induced by inflammatory mediators in human endothelial cells. Comparison with the thrombin receptor. J Biol Chem. 1996;271(25):14910-5.

46. Svensson KJ, Kucharzewska P, Christianson HC, Skold S, Lofstedt T, Johansson MC, et al. Hypoxia triggers a proangiogenic pathway involving cancer cell microvesicles and PAR-2-mediated heparin-binding EGF signaling in endothelial cells. Proc Natl Acad Sci U S A. 2011;108(32):13147-52.

47. Kwapiszewska G, Markart P, Dahal BK, Kojonazarov B, Marsh LM, Schermuly RT, et al. PAR-2 inhibition reverses experimental pulmonary hypertension. Circ Res. 2012;110(9):1179-91.

48. Davydova ON, Yakovlev AA, Lyzhin AA, Khaspekov LG, Gulyaeva NV. Growth Factors Deprivation Induces a Specific Increase in PAR2 Receptor mRNA Expression in Primary Cerebellar Cultures. Neurochem J+. 2010;4(4):279-83.

49. Keglowich L, Baraket M, Tamm M, Borger P. Hypoxia exerts dualistic effects on inflammatory and proliferative responses of healthy and asthmatic primary human bronchial smooth muscle cells. PLoS One. 2014;9(2):e89875.

50. Polosukhin VV, Cates JM, Lawson WE, Milstone AP, Matafonov AG, Massion PP, et al. Hypoxia-inducible factor-1 signalling promotes goblet cell hyperplasia in airway epithelium. J Pathol. 2011;224(2):203-11.

51. Saad MJ, Folli F, Kahn JA, Kahn CR. Modulation of insulin receptor, insulin receptor substrate-1, and phosphatidylinositol 3-kinase in liver and muscle of dexamethasone-treated rats. J Clin Invest. 1993;92(4):2065-72.

52. Buren J, Lai YC, Lundgren M, Eriksson JW, Jensen J. Insulin action and signalling in fat and muscle from dexamethasone-treated rats. Arch Biochem Biophys. 2008;474(1):91-101.

53. Pandey S, Lopez C, Jammu A. Oxidative stress and activation of proteasome protease during serum deprivation-induced apoptosis in rat hepatoma cells; inhibition of cell death by melatonin. Apoptosis. 2003;8(5):497-508.

54. Tammariello SP, Quinn MT, Estus S. NADPH oxidase contributes directly to oxidative stress and apoptosis in nerve growth factor-deprived sympathetic neurons. J Neurosci. 2000;20(1):RC53.

55. Lieberthal W, Triaca V, Koh JS, Pagano PJ, Levine JS. Role of superoxide in apoptosis induced by growth factor withdrawal. Am J Physiol. 1998;275(5 Pt 2):F691-702.

56. Aman M, Hirano M, Kanaide H, Hirano K. Upregulation of proteinase-activated receptor-2 and increased response to trypsin in endothelial cells after exposure to oxidative stress in rat aortas. J Vasc Res. 2010;47(6):494-506.

57. Kent KD, Bomser JA. Bovine pituitary extract provides remarkable protection against oxidative stress in human prostate epithelial cells. In Vitro Cell Dev Biol Anim. 2003;39(8-9):388-94.

58. Siddikuzzaman, Grace VM. Antioxidant potential of all-trans retinoic acid (ATRA) and enhanced activity of liposome encapsulated ATRA against inflammation and tumor-directed angiogenesis. Immunopharmacol Immunotoxicol. 2013;35(1):164-73.

59. Li C, Capan E, Zhao Y, Zhao J, Stolz D, Watkins SC, et al. Autophagy is induced in CD4+ T cells and important for the growth factor-withdrawal cell death. J Immunol. 2006;177(8):5163-8.

60. Tibbles LA, Woodgett JR. The stress-activated protein kinase pathways. Cell Mol Life Sci. 1999;55(10):1230-54.

61. Sridharan S, Jain K, Basu A. Regulation of autophagy by kinases. Cancers (Basel). 2011;3(2):2630-54.

62. Marino G, Niso-Santano M, Baehrecke EH, Kroemer G. Self-consumption: the interplay of autophagy and apoptosis. Nat Rev Mol Cell Biol. 2014;15(2):81-94.

63. Sui X, Kong N, Ye L, Han W, Zhou J, Zhang Q, et al. p38 and JNK MAPK pathways control the balance of apoptosis and autophagy in response to chemotherapeutic agents. Cancer Lett. 2014;344(2):174-9.

64. Iablokov V, Hirota CL, Peplowski MA, Ramachandran R, Mihara K, Hollenberg MD, et al. Proteinase-activated receptor 2 (PAR2) decreases apoptosis in colonic epithelial cells. J Biol Chem. 2014;289(49):34366-77.

65. Ahmad S, Ahmad A, Rancourt RC, Neeves KB, Loader JE, Hendry-Hofer T, et al. Tissue factor signals airway epithelial basal cell survival via coagulation and protease-activated receptor isoforms 1 and 2. Am J Respir Cell Mol Biol. 2013;48(1):94-104.

66. Zhao Y, Yang J, Liao W, Liu X, Zhang H, Wang S, et al. Cytosolic FoxO1 is essential for the induction of autophagy and tumour suppressor activity. Nat Cell Biol. 2010;12(7):665-75.

67. Zhang X, Tang N, Hadden TJ, Rishi AK. Akt, FoxO and regulation of apoptosis. Biochim Biophys Acta. 2011;1813(11):1978-86.

68. Grabiec AM, Angiolilli C, Hartkamp LM, van Baarsen LG, Tak PP, Reedquist KA. JNKdependent downregulation of FoxO1 is required to promote the survival of fibroblast-like synoviocytes in rheumatoid arthritis. Ann Rheum Dis. 2015;74(9):1763-71.

Kiao C, Puddicombe SM, Field S, Haywood J, Broughton-Head V, Puxeddu I, et al.
Defective epithelial barrier function in asthma. J Allergy Clin Immunol. 2011;128(3):549-56 e112.

70. Kicic A, Sutanto EN, Stevens PT, Knight DA, Stick SM. Intrinsic biochemical and functional differences in bronchial epithelial cells of children with asthma. Am J Respir Crit Care Med. 2006;174(10):1110-8.

71. Hackett TL, Singhera GK, Shaheen F, Hayden P, Jackson GR, Hegele RG, et al. Intrinsic phenotypic differences of asthmatic epithelium and its inflammatory responses to respiratory syncytial virus and air pollution. Am J Respir Cell Mol Biol. 2011;45(5):1090-100.

72. Freishtat RJ, Watson AM, Benton AS, Iqbal SF, Pillai DK, Rose MC, et al. Asthmatic airway epithelium is intrinsically inflammatory and mitotically dyssynchronous. Am J Respir Cell Mol Biol. 2011;44(6):863-9.

73. Stevens PT, Kicic A, Sutanto EN, Knight DA, Stick SM. Dysregulated repair in asthmatic paediatric airway epithelial cells: the role of plasminogen activator inhibitor-1. Clin Exp Allergy. 2008;38(12):1901-10.

74. Allard B, Bara I, Gilbert G, Carvalho G, Trian T, Ozier A, et al. Protease activated receptor-2 expression and function in asthmatic bronchial smooth muscle. PLoS One. 2014;9(2):e86945.

75. Lin KW, Park J, Crews AL, Li Y, Adler KB. Protease-activated receptor-2 (PAR-2) is a weak enhancer of mucin secretion by human bronchial epithelial cells in vitro. Int J Biochem Cell Biol. 2008;40(6-7):1379-88.

76. Zheng J, Liu W, Fan Y, Ye X, Xia W, Wang H, et al. Suppression of connexin 26 is related to protease-activated receptor 2-mediated pathway in patients with allergic rhinitis. Am J Rhinol Allergy. 2012;26(1):e5-9.

77. Heijink IH, van Oosterhout A, Kapus A. Epidermal growth factor receptor signalling contributes to house dust mite-induced epithelial barrier dysfunction. Eur Respir J. 2010;36(5):1016-26.

78. Gras D, Bourdin A, Vachier I, de Senneville L, Bonnans C, Chanez P. An ex vivo model of severe asthma using reconstituted human bronchial epithelium. J Allergy Clin Immunol. 2012;129(5):1259-66 e1.

79. Shrestha Palikhe N, Nahirney D, Laratta C, Gandhi VD, Vethanayagam D, Bhutani M, et al. Increased Protease-Activated Receptor-2 (PAR-2) Expression on CD14++CD16+ Peripheral Blood Monocytes of Patients with Severe Asthma. PLoS One. 2015;10(12):e0144500.

80. Abu-Remaileh M, Bender S, Raddatz G, Ansari I, Cohen D, Gutekunst J, et al. Chronic inflammation induces a novel epigenetic program that is conserved in intestinal adenomas and in colorectal cancer. Cancer Res. 2015;75(10):2120-30.

81. Balakrishnan A, Guruprasad KP, Satyamoorthy K, Joshi MB. Interleukin-6 determines protein stabilization of DNA methyltransferases and alters DNA promoter methylation of genes associated with insulin signaling and angiogenesis. Lab Invest. 2018.

82. Mahalingaiah PK, Ponnusamy L, Singh KP. Oxidative stress-induced epigenetic changes associated with malignant transformation of human kidney epithelial cells. Oncotarget. 2017;8(7):11127-43.

83. Watson CJ, Collier P, Tea I, Neary R, Watson JA, Robinson C, et al. Hypoxia-induced epigenetic modifications are associated with cardiac tissue fibrosis and the development of a myofibroblast-like phenotype. Hum Mol Genet. 2014;23(8):2176-88.

# Appendix-1

To understand the mechanism of PAR-2 regulation, AECs were treated with inhibitors of ERK1/2, PI3K and FOXO-1 (data was shown in chapter 2 and 3 of the thesis). All the inhibitors were dissolved in DMSO (Sigma-Aldrich) so that the final concentration of DMSO added to cells was equal to or less than 0.1%. It is important to evaluate if DMSO, at the concentration used, has any effect on PAR-2 expression (vehicle control treatment).

## Methods

To study if this concentration of DMSO has any effect on PAR-2 expression, we cultured AECs in media without supplements or with all supplements or with all supplements except insulin in the absence or presence of 0.1% DMSO. PAR-2 expression was studied using qRT-PCR as described before.



Fig. AECs were cultured in media without supplements (Red bars) or media with all supplements (Black bars) or media with all supplements except insulin (Blue bars) for 24h in the absence or presence of 0.1% DMSO and qRT-PCR for PAR-2 expression was performed. Data are presented as Mean of the two experiments.

### Results

Preliminary results suggest that DMSO at 0.1% concentration had no effect on PAR-2 expression in AECs.

# **Appendix-2**

#### Transfection optimization

BEAS-2B and NHBE cells were transfected with pmaxGFP (Lonza) using 3 different transfection reagents from Mirus; TransIT-X2, TransIT-2020 and TransIT-LI1.

The common transfection protocol for all 3 transfection reagents: BEAS-2B and NHBE cells were cultured in multi-well plate until 85% confluent. Cells were then rinsed with PBS and 1 mL of BEGM media without antibiotics was added. Transfection complexes were formed by mixing pmaxGFP (1  $\mu$ g) and transfection reagent in 100  $\mu$ l BEBM media for 30 min at room temperature and added drop wise to the culture dishes. Cells were trypsinized after 24h and assessed for transfection efficiency using flow cytometry.

First, BEAS-2B cells were transfected with the transfection complexes prepared using 1  $\mu$ g pmaxGFP and 3ul of the transfection reagents.

Table 1. Transfection efficiency of 3 different transfection reagents based on % pmaxGFP positive BEAS-2B cells

Transfection reagent	pmaxGFP positive cells (%)	pmaxGFP positive cells (%)	Average
TransIT-2020	13.3	15.2	14.25
TransIT-LT1	9.88	10.3	10.09
TransIT-X2	30.4	28.2	29.30

Results showed that TransIT-X2 was the most effective at transfecting BEAS-2B cells followed by TransIT-2020 and TransIT-LT1.

Next, BEAS-2B cells were transfected with transfection complexes prepared using 1  $\mu$ g pmaxGFP and different amount of TransIT-X2 transfection reagent (2ul - 6ul).

Volume of TransIT-X2	pmaxGFP positive cells (%)	pmaxGFP positive cells (%)	Average
2ul	30.3	29.4	29.85
3ul	18.3	16.2	17.25
4ul	11.1	12.1	11.60
5ul	8.11	7.29	7.70
6ul	8.09	5.78	6.94

Table 2. Transfection efficiency of different volumes of TransIT-X2 based on % pmax GFP positive BEAS-2B cells

Results showed that transfection was most efficient with the transfection complexes prepared using 2ul of TransIT-X2.

We then transfected NHBE cells with 1  $\mu$ g pmaxGFP using 2ul - 4ul TransIT-X2 transfection reagent.

Table 3. Transfection efficiency of different volumes of TransIT-X2 based on % pmax GFP positive NHBE cells

Volume of TransIT-X2	pmaxGFP positive cells (%)	pmaxGFP positive cells (%)	Average
2ul	39.8	41.5	40.65
3ul	20.7	21.6	21.15
4ul	15.2	11.3	13.25

Similar to BEAS-2B cells, TransIT-X2 reagent was most effective at transfecting NHBE cells when used at 2ul transfection reagent to 1  $\mu$ g pmaxGFP ratio. This transfection protocol was further used for FOXO-1-GFP experiments.

### Appendix-3

# Sequence of 670 bp promoter region (putative FOXO-1 binding site highlighted)

# FOXO-1 core binding site: T(G/A)TTTG (1-3)

### References

1. Durham SK, Suwanichkul A, Scheimann AO, Yee D, Jackson JG, Barr FG, et al. FKHR binds the insulin response element in the insulin-like growth factor binding protein-1 promoter. Endocrinology. 1999;140(7):3140-6.

2. Buteau J, Spatz ML, Accili D. Transcription factor FoxO1 mediates glucagon-like peptide-1 effects on pancreatic beta-cell mass. Diabetes. 2006;55(5):1190-6.

3. Vander Kooi BT, Streeper RS, Svitek CA, Oeser JK, Powell DR, O'Brien RM. The three insulin response sequences in the glucose-6-phosphatase catalytic subunit gene promoter are functionally distinct. J Biol Chem. 2003;278(14):11782-93.